Arch. Rational Mech. Anal. 218 (2015) 417–485 Digital Object Identifier (DOI) 10.1007/s00205-015-0863-0



Randomization and the Gross–Pitaevskii Hierarchy

Vedran Sohinger & Gigliola Staffilani

Communicated by L. SAINT-RAYMOND

Abstract

We study the Gross–Pitaevskii hierarchy on the spatial domain \mathbb{T}^3 . By using an appropriate randomization of the Fourier coefficients in the collision operator, we prove an averaged form of the main estimate which is used in order to contract the Duhamel terms that occur in the study of the hierarchy. In the averaged estimate, we do not need to integrate in the time variable. An averaged spacetime estimate for this range of regularity exponents then follows as a direct corollary. The range of regularity exponents that we obtain is $\alpha > \frac{3}{4}$. It was shown in our previous joint work with GRESSMAN (J Funct Anal 266(7):4705–4764, 2014) that the range $\alpha > 1$ is sharp in the corresponding deterministic spacetime estimate. This is in contrast to the non-periodic setting, which was studied by KLAINERMAN and MACHEDON (Commun Math Phys 279(1):169-185, 2008), where the spacetime estimate is known to hold whenever $\alpha > 1$. The goal of our paper is to extend the range of α in this class of estimates in a probabilistic sense. We use the new estimate and the ideas from its proof in order to study randomized forms of the Gross-Pitaevskii hierarchy. More precisely, we consider hierarchies similar to the Gross-Pitaevskii hierarchy, but in which the collision operator has been randomized. For these hierarchies, we show convergence to zero in low regularity Sobolev spaces of Duhamel expansions of fixed deterministic density matrices. We believe that the study of the randomized collision operators could be the first step in the understanding of a nonlinear form of randomization.

1. Introduction

1.1. Setup of the Problem

In this paper, we study the effect of randomization in the context of the *Gross– Pitaevskii hierarchy* on the three-dimensional torus \mathbb{T}^3 . Let us recall that the Gross– Pitaevskii hierarchy is an infinite system of linear PDEs that arises naturally in the derivation of the *nonlinear Schrödinger equation*. More precisely, the Gross–Pitaevskii hierarchy on the spatial domain Λ is given by:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\gamma^{(k)} = \sum_{j=1}^k B_{j,k+1}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_0^{(k)}. \end{cases}$$
(1)

Here, $(\gamma_0^{(k)})_k$ is a fixed sequence of *density matrices*, that is of functions $\gamma_0^{(k)}$: $\Lambda^k \times \Lambda^k \to \mathbb{C}$, and $(\gamma^{(k)})_k = (\gamma^{(k)}(t))_k$ is a sequence of time-dependent density matrices. The set Λ is a spatial domain, which we take to be either \mathbb{R}^d or \mathbb{T}^d . We say that the density matrix $\gamma_0^{(k)}$ has order k. The operators $\Delta_{\vec{x}_k}$ and $\Delta_{\vec{x}'_k}$ are defined to be the Laplacian in the first and second set of k spatial variables respectively. In other words:

$$\Delta_{\vec{x}_k} := \sum_{j=1}^k \Delta_{x_j}, \quad \Delta_{\vec{x}'_k} := \sum_{j=1}^k \Delta_{x'_j}$$

The map $B_{j,k+1}$ denotes the *collision operator*, which is precisely defined in Section 2. Let us note that, in this paper, we will not assume any additional symmetry properties of the solutions to (1).

The Gross–Pitaevskii hierarchy occurs in the study of Bose–Einstein condensation, which is a state of bosonic particles at temperatures which are close to absolute zero. The particles at this low temperature have a tendency to occupy a one-particle state, which corresponds to the solution of a nonlinear Schrödinger equation. In this context, the nonlinear Schrödinger equation is called the *Gross–Pitaevskii equation* after the work of GRoss [96] and PITAEVSKII [139]. The aforementioned physical phenomenon was predicted by Bose [20] and EINSTEIN [72] in 1924–1925. Their theoretical prediction was experimentally verified by the teams of CORNELL AND WIEMAN [7] and KETTERLE [65], who were jointly awarded the 2001 Nobel Prize in Physics for this discovery.

One strategy in deriving the Gross–Pitaevskii hierarchy and nonlinear Schrödinger type equations was developed by SPOHN [149]. In this approach, the hierarchy (1) comes from a limit of the related BBGKY hierarchy, which in turn is obtained from the *N*-body Schrödinger equation of a properly scaled *N*-body Hamiltonian. Making this convergence rigorous takes quite a bit of effort. In [149], the result obtained a rigorous derivation of the nonlinear Hartree equation $iu_t + \Delta u = (V * |u|^2)u$ for bounded convolution potentials V = V(x) on \mathbb{R}^d . The fact that the convolution potential is bounded allows one to obtain a direct derivation. Later, this strategy was extended to deal with more singular convolution potentials. More precisely, on \mathbb{R}^3 , the work of Spohn was subsequently extended to the case of Coulomb potentials $V(x) = \pm \frac{1}{|x|}$ by BARDOS, GOLSE and MAUSER [11] and ERDOS and YAU [81]. A key step in this extension was devoted to the study of the uniqueness of solutions to the hierarchy (1). Let us note that an alternative approach to related problems, based on Fock space techniques was simultaneously developed by HEPP [100] and GINIBRE and VELO [88,89].

In a sequence of monumental works, ERDOS, SCHLEIN and YAU [77–80] gave a rigorous derivation of the defocusing cubic nonlinear Schrödinger equation on

 \mathbb{R}^3 . In the aforementioned works, a significant step was to check the uniqueness of solutions to (1). The authors prove this claim by the use of Feynman graph counting techniques. In a subsequent work, KLAINERMAN and MACHEDON [108] gave an alternative proof of this claim, under slightly stronger assumptions. In particular, the authors gave a combinatorial reformulation of the Feynman graph expansion in the form of a *boardgame argument*. This technique allowed them to give a proof of uniqueness without the use of Feynman diagrams, under the assumption of an a priori bound on the solution. In [107], KIRKPATRICK, SCHLEIN, and the second author applied these ideas to the two-dimensional periodic setting. In particular, they obtained a rigorous derivation of the defocusing cubic nonlinear Schrödinger equation on \mathbb{T}^2 from the dynamics of many-body quantum systems. In our previous joint work with GRESSMAN, [92], we proved a conditional uniqueness result for the Gross–Pitaevskii hierarchy on \mathbb{T}^3 , which is the 3D analogue of the uniqueness result used in [107] for regularity strictly greater than 1. In a recent paper, CHEN, HAINZL, PAVLOVIĆ, and SEIRINGER [39] used the Quantum de Finetti theorem and gave an alternative proof of the uniqueness result on \mathbb{R}^3 from [77]. An extension of the above results to the context of general rectangular tori was given in the first author's joint work with HERR [102].

Let us note that the periodic problem was first considered in the work of ERDOS, SCHLEIN, and YAU [76] and ELGART, ERDOS, SCHLEIN, and YAU [73]. In particular, the authors study the Gross–Pitaevskii hierarchy on \mathbb{T}^3 and obtain all the steps of Spohn's strategy except for uniqueness. In the subsequent work [148], the first author proved the uniqueness step needed in the analysis of [73,76] and thus obtained a rigorous derivation of the defocusing cubic nonlinear Schrödinger equation on \mathbb{T}^3 from the dynamics of many-body quantum systems. A more detailed discussion about related works is given in Section 1.2.

In the above works, a key step in the proof of uniqueness was the proof of a spacetime estimate of the type:

$$\| S^{(k,\alpha)} B_{j,k+1} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2([0,T] \times \Lambda^k \times \Lambda^k)}$$

$$\lesssim \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}$$
(2)

for a fixed regularity exponent α and for a fixed time $T \in (0, +\infty]$. Here, $\mathcal{U}^{(k)}(t)$ denotes the analogue of the free Schrödinger evolution for the operator $i \partial_t + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})$ acting on density matrices of order k and $S^{(k,\alpha)}$ denotes the operator of taking α fractional derivatives of density matrices of order k. A precise definition of both operators is given in Section 2. We remark that in (2), the implied constant depends on T.

The estimate (2) is used in the step in which one wants to contract the Duhamel terms that occur in the study of (1). The use of this spacetime bound is what allows one to reduce what is originally a very long Duhamel expansion. The range of exponents of α for which one can prove (2) typically determines in which regularity class one can prove uniqueness of solutions to (1), at least by using the general method given in [108].

In [92], it was proved that (2) holds on \mathbb{T}^3 for $\alpha > 1$. As a result, we could prove a conditional uniqueness result for (1) on \mathbb{T}^3 in a class of density matrices possessing

 $\alpha > 1$ fractional derivatives and satisfying an a priori bound. We refer the reader to [92] for a precise definition. It was also shown that the factorized solutions to (1), which come from the NLS, belong to this class. The exact definition of factorized solutions is given in Section 2.3.

By using a specific counterexample, we showed that (2) does not hold on \mathbb{T}^3 when $\alpha = 1$. We noted that, in the special case of factorized density matrices, the estimate does hold when $\alpha = 1$. The fact that on \mathbb{T}^3 (2) does not hold in the endpoint case $\alpha = 1$ is in sharp contrast to the \mathbb{R}^3 case [108]. Our goal in this paper is to explore how we may extend the range of α on \mathbb{T}^3 in a *probabilistic sense*.

We note that the estimate (2) was also studied in its own right in the recent work of BECKNER [13]. In these works, the author gives several higher-dimensional generalizations in the non-periodic setting. The motivation for studying the spacetime estimate in this context is to develop a method for understanding restriction to a non-linear sub-variety [12].

In this paper, we fix the spatial domain to be $\Lambda = \mathbb{T}^3$. Our first goal is to prove an estimate of the type (2) for a larger range of α , in an *averaged sense*, by using a randomization procedure, which is precisely defined in Definition 1.5. In particular, we randomize the Fourier coefficients by multiplying them by a sequence of independent identically distributed standard Bernoulli random variables (meaning that their expected value is equal to 0 and their standard deviation is equal to 1). In the nonlinear dispersive equation literature, this idea was first applied in the work of BOURGAIN [22–25] on almost-sure well-posedness theory for the nonlinear Schrödinger equation in low regularities. These works build on a wide range of techniques on randomization in nonlinear dispersive equations, which were first developed in the work of LEBOWITZ, ROSE and SPEER [114], and ZHIDKOV [165]. A related approach for the local problem was recently developed by BURQ and TZVETKOV [33]. We note that the general idea in all of these works is to add randomness into the problem in order to extend the range of regularity exponents for which one can study the PDE. These and other relevant sources are explained in more detail in Section 1.2.

By using an appropriate randomization procedure, we prove the following result:

Theorem 1. Let $\alpha > \frac{3}{4}$ be given. There exists a constant C_0 depending only on α such that for all $k \in \mathbb{N}$ and $1 \le j \le k$, the following bound holds:

$$\left\| S^{(k,\alpha)} \left[B_{j,k+1} \right]^{\omega} \gamma_0^{(k+1)} \right\|_{L^2(\Omega \times \mathbb{T}^{3k} \times \mathbb{T}^{3k})}$$

 $\leq C_0 \left\| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \right\|_{L^2(\mathbb{T}^{3(k+1)} \times \mathbb{T}^{3(k+1)})}.$ (3)

The operator $[B_{j,k+1}]^{\omega}$ is a *randomized collision operator*, for a fixed ω belonging to the probability space Ω . It is obtained from the collision operator $B_{j,k+1}$ by appropriately randomizing the Fourier coefficients by means of standard Bernoulli random variables. A precise definition is given in (23)–(25) below.

Using (3), and the unitarity of $\mathcal{U}^{(k)}(t)$, it is possible to prove that, for all T > 0:

$$\| S^{(k,\alpha)} [B_{j,k+1}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2(\Omega \times [0,T] \times \mathbb{T}^{3k} \times \mathbb{T}^{3k})} \leq C_0 \sqrt{T} \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\mathbb{T}^{3(k+1)} \times \mathbb{T}^{3(k+1)})},$$
(4)

whenever $\alpha > \frac{3}{4}$. The constant C_0 is the same as in (3). In particular, we see that by (4), the range of regularity exponents is extended to $\alpha > \frac{3}{4}$, if one is willing to take the L^2 norm in the probability space Ω . This is in contrast to the deterministic setting in which it is known that one has to restrict to $\alpha > 1$ [92]. Even though the bound (4) is a direct extension of the known bound in the deterministic setting, in our further analysis, we have to use the stronger bound given by (3). The bound in (3) is the content of Theorem 3.1 below. The bound in (4) is the content of Corollary 3.2.

By using Markov's inequality and Corollary (4), we can deduce the following *large deviation bound*:

For $\alpha > \frac{3}{4}$, T > 0, $k \in \mathbb{N}$, $1 \le j \le k$, and $\lambda > 0$:

$$\mathbb{P}\Big(\| S^{(k,\alpha)} [B_{j,k+1}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2([0,T] \times \Lambda^k \times \Lambda^k)} \ge \lambda \Big) \\ \le \frac{C_0^2 T \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2}{\lambda^2}.$$
(5)

Here, C_0 is again the constant from (3). The large deviation bound (5) is the content of Corollary 3.3. This type of large deviation bound was shown to be useful in the study of nonlinear dispersive equations. In the context of the GP hierarchy, one has to apply similar estimates many times when estimating the Duhamel terms. In this context, it is easier to apply an averaged estimate as in (3) and (4) than such a large deviation bound.

In the context of the randomized collision operator $[B_{j,k+1}]^{\omega}$, one is led to the study of the *randomized Gross–Pitaevskii hierarchy*, which is given by:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_0^{(k)}. \end{cases}$$
(6)

As is noted in Section 4.1, the randomized Gross–Pitaevskii hierarchy (6) *admits* factorized solutions. For a fixed $\omega \in \Omega$, the sequence of density matrices given by $(\gamma^{(k)})_k := (|\phi^{\omega}\rangle \langle \phi^{\omega}|^{\otimes k})_k$ solves (6) whenever $\phi^{\omega} = T^{\omega}\phi$ is the randomization of ϕ that solves the nonlinear Schrödinger equation:

$$i\partial_t \phi + \Delta \phi = |\phi|^2 \phi.$$

The precise definition of the operator T^{ω} is given in (40) below. We note that the existence of factorized solutions is an important property of the (deterministic) Gross–Pitaevskii hierarchy (1).

In light of the existence of the factorized solutions of (6), we can view the randomized collision operator as corresponding to a *randomization in the nonlinearity*. More precisely, in (43) in Section 4.1 below, we note that the function ϕ^{ω} , from which we build the factorized solution, solves the equation:

$$i\partial_t\phi^\omega + \Delta\phi^\omega = T^\omega \Big(|T^\omega\phi^\omega|^2 \cdot T^\omega\phi^\omega \Big).$$

This is a *nonlinear Schrödinger equation with a random nonlinearity*. The random nonlinearity is encoded in the random collision operator via the above construction of factorized solutions to (6).

An interesting open problem would be to obtain a (non-deterministic) criterion for uniqueness of solutions to (6) in a class which contains the natural energy space of time-dependent density matrices. This is the class of $(\gamma^{(k)}(t))$, for which there exists a constant C > 0 such that for all $k \in \mathbb{N}$, the following energy bound holds:

$$\|S^{(k,1)}\gamma^{(k)}(t)\|_{L^2(\Lambda^k \times \Lambda^k)} \le C^k$$

uniformly in time. Here, $S^{(k,1)}$ is the differentiation operator which is precisely defined in (20). As a first step in this direction, one would be interested in studying the Duhamel expansions corresponding to the homogeneous problem associated to (6).

More precisely, in order to prove any type of uniqueness result for (6), one would have to argue as in the deterministic setting and study the problem:

$$\begin{cases} i \partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega} (\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = 0. \end{cases}$$

and one would have to find a criterion for $(\gamma^{(k)})_k$ such that the Duhamel terms:

$$\sigma_{n;\,\omega}^{(k)}(t_k) := (-i)^n \int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} \mathcal{U}^{(k)}(t_k - t_{k+1}) \left[B^{(k+1)} \right]^\omega \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2})$$

$$[B^{(k+2)}]^\omega \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k})$$

$$[B^{(n+k)}]^\omega \gamma^{(n+k)}(t_{n+k}) \, \mathrm{d}t_{n+k} \cdots \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1}$$
(7)

converge to zero in an appropriate norm as $n \to \infty$. Here, we assume that $k \in \mathbb{N}$ and $t_k \in \mathbb{R}$ are fixed and we use the shorthand notation: $[B^{(\ell+1)}]^{\omega} := \sum_{j=1}^{\ell} [B_{j,\ell+1}]^{\omega}$. In the definition of $\sigma_{n;\omega}^{(k)}$, k denotes the order of the density matrix, n denotes the length of the Duhamel expansion and ω is a fixed parameter in the probability space Ω . There are several difficulties in applying the technique used in the deterministic setting. Arguing by analogy with the deterministic setting, one would want to apply the spacetime estimate (4) in order to contract the Duhamel term. The main issue is that, in (4), the density matrix $\gamma_0^{(k+1)}$ is not allowed to depend on ω . In the expansion (7), there is dependence on ω in each collision operator $[B^{(\ell+1)}]^{\omega}$ for $\ell = k, \ldots, n-1$. Moreover, the density matrix $\gamma^{(n+k)}$ depends, in principle, on ω in a complicated way. In any case, it is not possible to apply the spacetime bound

in order to contract the terms as was the case in the deterministic setting. This point is explained in Section 6.1.

One way of dealing with the difficulty of the ω -dependence of all of the collision operators $[B^{(\ell+1)}]^{\omega}$ is to modify the problem and to look at the *independently randomized Gross–Pitaevskii hierarchy*, that is, given a sequence $(\omega_k)_{k\geq 2}$ of elements in the probability space Ω , we look at the hierarchy:

$$\begin{cases} i \partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega_{k+1}} (\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_0^{(k)}. \end{cases}$$
(8)

In other words, the randomizations in the collision operators at different levels are now independent of each other. We symbolize this by denoting the randomization parameters ω_{k+1} differently. The hierarchy (8) is a generalization of (6) in the sense that it reduces to the latter hierarchy when all of the ω_{k+1} are mutually equal. However, it is no longer true in general that (8) admits factorized solutions, see Section 5.1.

When studying the homogeneous problem associated to (8), one is led to the study of the following Duhamel terms:

$$\sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t_k) := (-i)^n \int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} \mathcal{U}^{(k)}(t_k - t_{k+1}) \left[B^{(k+1)} \right]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \\ \left[B^{(k+2)} \right]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \\ \left[B^{(n+k)} \right]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) \, \mathrm{d}t_{n+k} \cdots \, \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1}.$$
(9)

We note that the superscript *k* denotes the order of the density matrix, the subscript *n* denotes the length of the Duhamel expansion, and $\omega_{k+1}, \omega_{k+2}, \ldots, \omega_{n+k}$ are independently chosen elements of the probability space Ω . In this form, it is still the case that $\gamma^{(n+k)}$ has a complicated dependence on $(\omega_{k+1}, \omega_{k+2}, \ldots, \omega_{n+k})$.

We deal with the last difficulty by fixing a *deterministic* sequence of timedependent density matrices $(\gamma^{(k)}(t))_k$ satisfying the a priori bound

$$\|S^{(k,\alpha)}\gamma^{(k)}(t)\|_{L^2(\mathbb{T}^{3k}\times\mathbb{T}^{3k})} \le C_1^k \tag{10}$$

for some constant $C_1 > 0$, which is independent of k and t. Here, we do not assume that $(\gamma^{(k)})_k$ solves (8). The a priori condition is natural in the sense that it is satisfied globally in time for factorized solutions to (1) and (6) when $\alpha = 1$, and it is satisfied locally in time for factorized solutions when $\alpha > 1$. The latter observation can be deduced from [21]. Moreover, sequences of density matrices satisfying such a priori bounds were shown to arise naturally in the study of the Cauchy problem for the GP hierarchy in the recent work of CHEN and PAVLOVIĆ [41–45].

We now define $\sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}$ analogously as in (9), keeping in mind that the sequence $(\gamma^{(k)}(t))_k$ is a fixed sequence of time-dependent density matrices

which do not necessarily solve (8). We now fix $n \in \mathbb{N}$ and we look at $\tilde{\gamma}^{(k)}$ for k = 1, ..., n given by:

$$\begin{split} \tilde{\gamma}^{(1)} &:= \sigma_{n; \, \omega_{2}, \omega_{3}, \omega_{4}, \omega_{5}, \dots, \, \omega_{n+1}}^{(1)} \\ \tilde{\gamma}^{(2)} &:= \sigma_{n-1; \, \omega_{3}, \omega_{4}, \omega_{5}, \dots, \, \omega_{n+1}}^{(2)} \\ \tilde{\gamma}^{(3)} &:= \sigma_{n-2; \, \omega_{4}, \omega_{5}, \dots, \, \omega_{n+1}}^{(3)} \\ &\vdots \\ \tilde{\gamma}^{(n)} &:= \sigma_{1; \, \omega_{n+1}}^{(n)} . \end{split}$$

The $\tilde{\gamma}^{(k)}$ then solve, by construction:

$$\begin{cases} i\partial_t \tilde{\gamma}^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \tilde{\gamma}^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega_{k+1}} (\tilde{\gamma}^{(k+1)}) \\ \tilde{\gamma}^{(k)} \big|_{t=0} = 0, \end{cases}$$

for all $k \in \{1, 2, ..., n - 1\}$. In other words, we obtain an *arbitrarily long subset* of solutions to the full hierarchy (8) with zero initial data.

It is now possible to apply the estimate (3) in order to study the Duhamel terms $\sigma_{n; \omega_{k+1}, \omega_{k+2}, \dots, \omega_{n+k}}^{(k)}$. We can prove the following:

Theorem 2. Suppose that $\alpha > \frac{3}{4}$ and $k \in \mathbb{N}$. There exists T > 0 depending only on the constant C_1 in (10) and on α such that:

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\Omega_{k+1}\times\Omega_{k+2}\times\cdots\times\Omega_{n+k};\,L^2(\mathbb{T}^{3k}\times\mathbb{T}^{3k})\right)} \to 0$$
(11)

as $n \to \infty$. Moreover,

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\prod_{m\geq 2}\Omega_m;L^2(\mathbb{T}^{3k}\times\mathbb{T}^{3k})\right)} \to 0$$
(12)

as $n \to \infty$.

This result is proved as Theorem 5.2 in Section 5.2. As is noted in Remark 3.7, it is possible to prove Theorem 2 when $\Lambda = \mathbb{T}^d$, for $d \ge 1$ provided that $\alpha > \frac{d}{4}$. Throughout our paper, we mostly study the case $\Lambda = \mathbb{T}^3$, since in [92] we were able to describe the full range of regularity exponents which are admissible in the estimate (4). We note that the norm in (12) is well-defined due to a result first proved by work of KAKUTANI [106], which builds on the previous work by KOLMOGOROV [111]. The significance of taking this norm, compared to the one in (11), is that, in this way, we measure convergence in a space which is independent of *n*.

Let us emphasize that the time interval *T* in the above theorem is independent of *n*. As we will see in the proof, it is the case that $T \sim \frac{1}{C_1}$. In other words, we obtain uniform convergence to zero of the Duhamel terms as $n \to \infty$ in a fixed norm on a fixed time interval. As was remarked earlier, the spacetime estimate (4) is not directly applicable in the study of the randomized Gross–Pitaevskii hierarchy (6). In this context, one would like to prove an estimate of the form:

$$\| S^{(n,\alpha)} \mathcal{U}^{(n)}(t_1 - t_2) [B^{\pm}_{j_1,k_1}]^{\omega} \mathcal{U}^{(n+1)}(t_2 - t_3) [B^{\pm}_{j_2,k_2}]^{\omega} \cdots \cdots \mathcal{U}^{(n+\ell-1)}(t_{\ell} - t_{\ell+1}) [B^{\pm}_{j_{\ell},k_{\ell}}]^{\omega} \gamma_0^{(n+\ell)} \|_{L^2(\Omega \times \mathbb{T}^{3n} \times \mathbb{T}^{3n})} \leq C \| S^{(n+\ell,\alpha)} \gamma_0^{(n+\ell)} \|_{L^2(\mathbb{T}^{3(n+\ell)} \times \mathbb{T}^{3(n+\ell)})}$$
(13)

for all sequences of density matrices $(\gamma_0^{(k)})_k$, and for $n, \ell \in \mathbb{N}, j_1, \dots, j_\ell, k_1, \dots, k_\ell \in \mathbb{N}$, with $1 \le j_1 < k_1 \le n + \ell, \dots, 1 \le j_\ell < k_\ell \le n + \ell$ and $t_1, t_2, \dots, t_{\ell+1} \in \mathbb{R}$.

The fact that (13) holds when n = 1 and $\alpha > \frac{3}{4}$ follows from (4). However, it is shown in Section 6.1 that, for general density matrices, we cannot use the randomization to obtain this type of bound. In this subsection, we note that the pairing of the frequencies as a result of the randomization does not allow us to close the estimate in the case n = 2 as it did in the case when n = 1. The argument for $n \ge 3$ similarly does not apply.

The good news is that the estimate (13) holds if we restrict the density matrices to lie in an appropriate *non-resonant class*. This is similar to the idea of *Wick-ordering* [24,61] and related ideas applied in [128] in the context of the nonlinear Schrödinger equation. In fact, we will work in the non-resonant class \mathcal{N} , which is precisely defined in Section 6.4. We note that this class does not contain the factorized solutions to (6). As we will see, it is an important matter to examine the behavior of the constant *C* obtained in (13) in terms of *n* and ℓ . The exact estimate is given in Theorem 6.4 of Section 6.4.

We again consider a time-dependent deterministic sequence of density matrices $(\gamma^{(k)}(t))$. Unlike as in the previous case, we assume that, for each fixed *t*, the sequence $(\gamma^{(k)}(t))_k$ belongs to the non-resonant class \mathcal{N} . As a part of the definition, we assume that the sequence of density matrices satisfies an a priori bound as in (10). Having chosen such a sequence $(\gamma^{(k)})$, we define $\sigma_{n;\omega}^{(k)}(t)$ as in (9). In this way, we again obtain an arbitrarily long subset of solutions to the hierarchy (6).

We are now in the position to apply the randomized spacetime estimate to the study of $\sigma_{n;\omega}^{(k)}$, that is of Duhamel expansions of order *n* of elements of the non-resonant class \mathcal{N} . In particular, we can prove:

Theorem 3. Suppose that $\alpha \ge 0$ and $k \in \mathbb{N}$. There exists T > 0, depending only on the constant C_1 in the definition of the class \mathcal{N} , α , and on k such that:

$$\sup_{t \in [0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega}^{(k)}(t) \right\|_{L^2\left(\Omega \times \mathbb{T}^{3k} \times \mathbb{T}^{3k}\right)} \to 0 \tag{14}$$

as $n \to \infty$.

Remark 1.1. Let us observe that now, the range of regularity exponents has been extended to $\alpha \ge 0$. This is a significant step in the direction of the understanding of the low-regularity problem for Gross–Pitaevskii type hierarchies. Theorem 3 is given below as Theorem 6.4.

Remark 1.2. We note that, due to the presence of only one random parameter $\omega \in \Omega$, the norm in (14) of Theorem 3 is simpler than the one in Theorem 2.

Remark 1.3. The analysis in Theorem 3 extends without any changes to the general case of $\Lambda = \mathbb{T}^d$, for $d \ge 1$.

Remark 1.4. We note that the results of Theorems 2 and 3 estimate the Duhamel expansions similar to those obtained in the uniqueness analysis in the deterministic setting, as in [41,42,92,107,108]. The results of Theorems 2 and 3 show that these Duhamel expansions converge to zero in a class of density-matrices in a low-regularity space with a random component.

In the subsequent work of the first author [147], it was shown that the full randomized hierarchies (6) and (8) have local-in-time solutions for almost every value of the random parameter. This is an existence result which was motivated by truncation techniques used by CHEN and PAVLOVIĆ [44] in the deterministic context. In [147], such solutions of (6) are obtained for regularities is $\alpha > \frac{3}{4}$, and solutions of (8) are obtained for regularities $\alpha \ge 0$. As is the case in the assumptions of Theorem 3, the initial data for (8) in [147] needs to belong to an appropriate non-resonant class.

1.2. Previously Known Results

In addition to the references mentioned above, there is a vast literature on the connection between the derivation of NLS-type equations and hierarchies of the type (1) and related problems. The derivation of a hierarchy similar to (1) coming from the limit of N-body Schrödinger dynamics was obtained in [1,2,11,15-17,45,48,51–57,74,84–87,124,162] in various different contexts. The first rate of convergence result was obtained by RODNIANSKI and SCHLEIN [145] and subsequent rate of convergence results have been obtained in [6,17,38,49,50,75,85,93-95,110,115,121,124,137,138]. The Cauchy problem associated to a hierarchy as in (1) has been studied in its own right in [41-44,46,47,59]. A new unconditional uniqueness result for the cubic Gross–Pitaevskii hierarchy on \mathbb{R}^3 , based on the Quantum de Finetti theorem has recently been obtained by CHEN, HAINZL, PAVLOVIĆ, and SEIRENGER in [39]. These techniques have been adapted in order to show scattering results in the context of the Gross-Pitaevskii hierarchy in [40]. Low regularity extensions of this method have been obtained in [103], as well as in the case of three-body interactions [104]. The periodic analogue of the methods from [39] was an important step in [147]. Recently, the Quantum de Finetti was also used in the study of the CHERN–SIMONS–SCHRÖDINGER hierarchy in [58].

Important results related to the Gross–Pitaevskii hierarchy have been studied on the level of the *N*-body Schrödinger equation in the series of works [117, 119, 120]with an expository account given in [118]. We refer the reader to the introduction of [92] for a more precise discussion of the problem and of the references mentioned above. We would also like to recall the connection with certain optical lattice models that have been studied in [4, 5]. For a detailed introduction to the general problem, we also refer the interested reader to the lecture notes by SCHLEIN [146]. An approach based on randomization has been useful in the study of lowregularity solutions to nonlinear dispersive equations. In particular, randomization techniques have been applied in the context where the low regularity techniques of the *high-low method* as in the work of BOURGAIN [26] or the *I-method*, as in the work of COLLIANDER, KEEL, STAFFILANI, TAKAOKA, and TAO [60] are known not to work. As was mentioned in the introduction, this probabilistic approach was first used by BOURGAIN [22–25] and it has its origins in previous work of LEBOWITZ, ROSE, and SPEER [114], and ZHIDKOV [165]. The main idea is that global existence can be studied by means of the existence of a Gibbs measure and its invariance under the flow. The properly constructed Gibbs measure is supported away from the set of initial data for which the above deterministic methods do not apply.

The Gibbs measure approach is only known to be applicable in the context in which there exists a Hamiltonian structure. In a more general context, the idea of a direct randomization of the Fourier coefficients without the use of an invariant Gibbs measure has been shown to be useful. One randomizes the function $f = \sum_n c_n e^{inx}$ by multiplying each Fourier coefficient c_n with $h_n(\omega)$, where $h_n(\omega)$ are independent, identically distributed standard random variables whose expected value is zero. The precise definition is given in Definition 1.5 below.

In the context of nonlinear dispersive PDE, this idea was pioneered in the work of BURQ and TZVETKOV [33]. A key fact in this work is the fact that, due to the randomization, one can obtain improved L^p regularity in the initial data almost surely. This is a phenomenon which was first noted in work of RADEMACHER [142] and PALEY and ZYGMUND [134–136]. Related results were also proved in the work of MARCINKIEWICZ and ZYGMUND [123] and Khintchine, see [161]. We also note the work [8], in which the authors study the special case of Gaussian random series in which some of these observations can be deduced by alternative means. It is noted in [33] that the randomization does not improve regularity on the Sobolev scale almost surely. Hence, all of the gain has to be obtained due to the almost sure gain in integrability. Let us remark that the idea of a gain in integrability due to randomization is related to the more general phenomenon of *hypercontractivity* [82,90,97,98,129].

In addition to the mentioned works, there is a wide range of results on the application of randomization to nonlinear dispersive PDEs. We refer the interested reader to the works [18,19,28–36,61,68–70,122,125,126,130–133,144,150–158, 163], as well as to the expository works [27,166] and to the references therein. Furthermore, we note that the idea of randomization of the Fourier coefficients without the use of an invariant measure has also been applied in the context of the Navier-Stokes equations. The relevant works are [66,67,127,164]. By applying this method, the authors were able to obtain existence results almost surely in supercritical regimes.

Probabilistic methods have previously been applied in the context of the *N*-body Schrödinger equation, but in a slightly different context. In the work [14], it is proved that the fluctuations around the limiting dynamics given by the Hartree equation satisfy a central limit theorem. This approach builds on previous central limit theorems in a quantum setting [63,64,91,99,101,105,113].

An alternative probabilistic approach was taken in [3]. Here, the authors consider a many-body system of mutually repellent bosons and they derive an explicit variational expression for the corresponding limiting free energy. In doing so, they use the cycle structure of random paths which appear in the Feynman–Kac formula. The main novelty of this paper is to recast the whole interacting system as an expectation with respect to a marked Poisson process. Methods involving point processes were previously applied in the study of Bose gases without interaction in [83,143].

In [37], the authors analyze the fluctuations of the Bose–Einstein condensate for a system of non-interacting bosons with a potential. The system is assumed to modeled according to the canonical ensemble. In this context, the authors give a rigorous proof of Bose–Einstein condensation with positive probability assuming that the temperature is sufficiently low. The probability measure on the space of configurations is the *canonical Gibbs measure*.

In a recent preprint LEWIN, NAM, and ROUGERIE [116] show that the nonlinear Gibbs measure associated to the nonlinear Schrödinger equation, can be obtained from the corresponding many-body grand canonical quantum Gibbs states. This result is shown in the one-dimensional case for the standard NLS, and in higher dimensions for specific forms of the interaction. We note that related results on lattices were also proved in [109].

Probabilistic techniques have been applied to the study of *N*-body Schrödinger problems in the experimental literature as well. In particular, we note the work [112]. Here, the experiment is based on applying a randomized Monte Carlo method of thermodynamic measurements of a unitary Fermi gas across the superfluid phase transition. In this way, it is possible to validate the theory of strongly interacting matter given by BARDEEN, COOPER and SCHRIEFFER [9, 10, 62]. It is also noted that similar probabilistic methods can be applied in the study of other physical systems, such as two-dimensional Bose and Fermi gases as well as fermions in optical lattices. The precise formulation of the randomization method, that is the *Bold diagrammatic Monte Carlo (BDMC)*, was first developed in [140, 141, 159]. This approach was also used in [160] in the context of formal summation of Feynman graphs.

1.3. Ideas and Techniques used in the Proofs

Let us first recall the definition of the randomization of a function. Here $\Lambda = \mathbb{T}^d$ is the spatial domain.

Definition 1.5. Given $f \in L^2(\Lambda)$, with Fourier series $\sum_n c_n e^{inx}$ and a sequence of independent, identically distributed random variables with expected value zero $(h_n(\omega))$, we define the function f^{ω} by:

$$f^{\omega} := \sum_{n} h_n(\omega) c_n e^{inx}, \tag{15}$$

 f^{ω} is called the randomization of f.

As was mentioned above, one of the big improvements obtained by randomization is the *gain of integrability*, almost surely. More precisely, given a sequence of independent identically distributed random Bernoulli or Gaussian random variables $(h_n(\omega))$, centered at zero, one obtains that for all 1 :

$$\left\|\sum_{n} h_n(\omega) c_n\right\|_{L^p(\Omega)} \le C(p) \left(\sum_{n} |c_n|^2\right)^{\frac{1}{2}}.$$
(16)

We observe that this is a better bound than the bound $C(p)(\sum |c_n|)$, which immediately follows from the triangle inequality. For a proof of (16), we refer the reader to the proof of Lemma 4.2 in [33].

We can then use Fubini's theorem, (16) with $c_n := \widehat{f}(n) \cdot e^{inx}$ and Plancherel's Theorem in order to deduce that:

$$\|f^{\omega}\|_{L^{p}(\Omega \times \Lambda)} \leq C(p)\|f\|_{L^{2}}.$$

In this way, the randomization gives us a gain in integrability, almost surely.

It is instructive to recall the main idea of the proof of (16). One typically takes p to be an even integer, since the other cases can be deduced by interpolation. Taking p-th powers of the left-hand side in (16), we obtain a sum of terms of the type:

$$\int_{\Omega} h_{n_1}(\omega) \cdots h_{n_k}(\omega) \cdot \overline{h}_{m_1}(\omega) \cdots \overline{h}_{m_\ell}(\omega) \cdot c_{n_1} \cdots c_{n_k} \cdot \overline{c}_{m_1} \cdots \overline{c}_{m_\ell} \, \mathrm{d} p(\omega)$$

for some $k, \ell \in \mathbb{N}$ with $k+\ell = p$. Since the random variables are mutually independent and since they have expected value equal to zero, the above expression equals to zero if an n_i or and m_j occurs exactly once in the set $\{n_1, \ldots, n_k, m_1, \ldots, m_\ell\}$. Hence, each frequency must *pair up* with at least one of the other frequencies or else the contribution equals zero. This observation reduces the original sum one is considering in (16). The claim then follows by using Hölder's inequality. We note that this argument works in all dimensions.

In applications of randomization techniques to the study of nonlinear PDE, which were mentioned above, one usually randomizes in the initial data. In particular, if one considers an NLS-type equation:

$$\begin{cases} i \partial_t u + \Delta u = N(u) \\ u \Big|_{t=0} = u_0 \end{cases}$$

for some power-type nonlinearity N(u), one typically takes the initial data to be *random*. In other words, one considers $u_0 = \phi^{\omega}$, in the sense discussed in Definition 1.5 above. Here $\phi = \phi(x)$ is a fixed deterministic function. As was noted earlier, the function ϕ^{ω} exhibits better integrability properties than the function ϕ , on average. The key step is then to write:

$$u = e^{it\Delta}\phi^{\omega} + v$$

and to note that the function v satisfies the following Cauchy problem, with homogeneous initial data:

$$\begin{cases} i\partial_t v + \Delta v = N(e^{it\Delta}\phi^{\omega} + v) \\ v\big|_{t=0} = 0. \end{cases}$$
(17)

In solving for v in (17), one uses a fixed point argument. One controls the terms coming from $e^{it\Delta}\phi^{\omega}$ in a higher L^p space. More precisely, one uses the gain of integrability almost surely combined with a large deviation bound to note that the corresponding L^p norm of the linear part is bounded for a large set of $\omega \in \Omega$. A quantitative estimate on the measure of the good set of ω is given by a large deviation bound, which is typically deduced by Markov's inequality. One can apply all of these facts and use Hölder's inequality to note that the nonlinear term satisfies the right estimate from which one can apply a fixed point argument, for ω belonging to a good subset of Ω . An important step in this approach is to use the fact that, in the expression $N(e^{it\Delta}\phi^{\omega} + v)$, the only first-order terms in v are multiplied with a factor of $e^{it\Delta}\phi^{\omega}$, which is bounded in a higher L^p norm.

The above approach does not apply to the study of the Gross–Pitaevskii hierarchy since the problem is now linear. Namely, let us consider the problem with random initial data:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\gamma^{(k)} = \sum_{j=1}^k B_{j,k+1}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_{0,\omega}^{(k)}. \end{cases}$$

We now write:

$$\gamma^{(k)} = \mathcal{U}^{(k)}(t) \, \gamma^{(k)}_{0,\omega} + \tilde{\gamma}^{(k)}.$$

The $\tilde{\gamma}^{(k)}$ now solve the following problem with homogeneous initial data:

$$\begin{cases} i \partial_t \tilde{\gamma}^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \tilde{\gamma}^{(k)} = \sum_{j=1}^k B_{j,k+1} \left(\mathcal{U}^{(k+1)}(t) \gamma_{0,\omega}^{(k+1)} + \tilde{\gamma}^{(k+1)} \right) \\ \tilde{\gamma}^{(k)} \big|_{t=0} = 0. \end{cases}$$
(18)

We note that the operator $B_{j,k+1}$ is linear so the right-hand side in (18) equals:

$$\sum_{j=1}^{k} B_{j,k+1} \left(\mathcal{U}^{(k+1)}(t) \gamma_{0,\omega}^{(k+1)} \right) + \sum_{j=1}^{k} B_{j,k+1} \left(\tilde{\gamma}^{(k+1)} \right)$$

In other words, the free evolution of the random part given by $\mathcal{U}^{(k+1)}(t) \gamma_{0,\omega}^{(k+1)}$ and the remainder $\tilde{\gamma}^{(k+1)}$ on the right-hand side are *completely decoupled*. As a result, there is no small factor multiplying the term which one would want to estimate, that is $\sum_{j=1}^{k} B_{j,k+1}(\tilde{\gamma}^{(k+1)})$.

Due to the phenomenon discussed above, we *randomize in the collision operator*, instead of in the initial data. This approach can be viewed as a first step in the direction of a *nonlinear form of randomization*, since the collision operator on the level of the Gross–Pitaevskii hierarchy corresponds to the nonlinearity on the level of the nonlinear Schrödinger equation. A remark on what is possible to say when one randomizes in the initial data in the context of the Gross–Pitaevskii hierarchy is given in Section 3.1.

An additional motivation for randomizing the collision operator is that doing so helps us obtain a gain in integrability and thus prove a randomized version of the spacetime estimate (2) for a larger class of α . More precisely [c.f. (21) and (22) below], on the Fourier transform side, the collision operator involves a sum in the frequencies as follows:

$$\begin{pmatrix} B_{1,k+1}\gamma_0^{(k+1)} \end{pmatrix} \widehat{(\xi_k; \xi'_k)}$$

$$= \sum_{\xi_{k+1},\xi'_{k+1} \in \mathbb{Z}^3} \widehat{\gamma_0}^{(k+1)}(\xi_1 - \xi_{k+1} + \xi'_{k+1}, \xi_2, \dots, \xi_k, \xi_{k+1}; \xi'_1, \dots, \xi'_k, \xi'_{k+1})$$

$$- \sum_{\xi_{k+1},\xi'_{k+1} \in \mathbb{Z}^3} \widehat{\gamma_0}^{(k+1)}(\xi_1, \dots, \xi_k, \xi_{k+1}; \xi'_1 - \xi'_{k+1} + \xi_{k+1}, \xi'_2, \dots, \xi'_k, \xi'_{k+1}).$$

Our goal is to multiply the summands in the above formula with appropriate random coefficients in such a way that we can apply the techniques used in the proof of (16) in the context of the spacetime estimate (2). The randomization given in (23) and (24) below is the one which leads to the randomized spacetime bound given by (3).

We recall that in [92], the proof of the spacetime estimate relied on the dispersive effect of the free evolution $\mathcal{U}^{(k+1)}(t)$. In the randomized setting, we only use the fact that this operator is unitary on L^2 -based spaces. It is the pairing of the frequencies which ultimately gives us the gain. This is a common phenomenon in the study of nonlinear dispersive equations by means of randomization. As a result of the proof, we will be able to prove an estimate on a quantity which does not involve an integral in time, as is seen in (3).

It is also possible that the range of α in (4) could be extended by using dispersive properties of $\mathcal{U}^{(k+1)}(t)$ by using similar ideas as in [24]. We will not pursue this approach in our paper. We remark that such an improvement could only be applicable in estimates in which the norms involve integration in time. As we will see, norms which do not involve integrals in time are more applicable to the study of randomized Gross–Pitaevskii hierarchies.

As was noted above, the construction of the randomized collision operator leads to the study of the *randomized Gross–Pitaevskii hierarchy* (6) and of the *independently randomized Gross–Pitaevskii hierarchy* (8). The randomized Gross– Pitaevskii hierarchy (6) shares a lot with properties of the regular Gross–Pitaevskii hierarchy, such as the existence of factorized solutions and of the applicability of the boardgame argument. As in the deterministic setting, we need to assume that our density matrices are invariant under permutation of the spatial variables \vec{x}_k and \vec{x}'_k in order to apply the boardgame argument. Due to the dependent randomization, we will not be able to apply the randomized spacetime estimate (3) directly in this context. More precisely, we can only apply the averaged spacetime estimate to one Duhamel iteration, and as soon as we do at least two iterations of the Duhamel principle, we can no longer directly use Theorem 3.1. The reason is that we would have to estimate a quantity of the type:

$$\left\|S^{(k,\alpha)}[B_{j,k+1}]^{\omega}\gamma_{0,\omega}^{(k+1)}\right\|_{L^2(\Omega\times\Lambda^k\times\Lambda^k)}$$

where $\gamma_{0,\omega}^{(k+1)}$ depends on ω . The proof of Theorem 3.1 relies heavily on the fact that $\gamma_0^{(k+1)}$ is independent of ω , which, in turn, allows us to deduce the pairing of the frequencies.

One way of getting around this difficulty is to work with the independently randomized GP hierarchy (8). However, in this case we have to be careful and recall that it is not possible to apply the boardgame argument in this context. The problem in applying the boardgame argument lies in the fact that, in the integrals given by the Duhamel expansion, it is possible to interchange the t_k and x_k variables, but it is not possible to interchange the ω_k variables. Nonetheless, in the context of the independently randomized Gross–Pitaevskii hierarchy, we can directly apply the spacetime estimate (3) in order to show (11).

An important observation in the proof of (11) is that, in (3), we do not need to put the free evolution $\mathcal{U}^{(k+1)}(t)$ and a time integral inside of the norm in order to obtain the estimate. As a result, we can just use the unitarity of $\mathcal{U}^{k+1}(t)$. An important fact which we use is the gain of $\frac{1}{n!}$ in the integral identity:

$$\int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} \mathrm{d}t_{n+k} \cdots \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1} = \frac{t_k^n}{n!}.$$
 (19)

This gain was previously used in the study of the Gross–Pitaevskii hierarchy on \mathbb{R} in [42]. As a result, we can control the factorial number of Duhamel terms which we obtain in the expansion. Consequently, we can prove Theorem 5.2 with a quantity T > 0, which is independent of n.

A possible approach in the context of the hierarchy (6) is to argue directly and prove a good spacetime estimate for higher-order Duhamel expansions without directly using (3). This reduces to a purely combinatorial problem of possible pairings of the frequencies. However, as is shown in Section 6.1, it is not possible to prove a good spatial estimate in the class of general density matrices by applying the combinatorial method used to prove (3). In Section 6.4, an appropriate spatial estimate is shown if one imposes an additional condition of *non-resonance*, similar to [24,61,128]. In fact, by working in the non-resonant classes, we are able to prove an estimate in a regime which allows us to go all the way down to the regularity of L^2 , that is to $\alpha = 0$. By studying the randomized Gross–Pitaevskii hierarchy in this, we show (14).

1.4. Organization of the Paper

In Section 2, we recall and define the relevant notation. In particular, in Section 2.1, we recall the definition of the Fourier transform and differentiation of density matrices, whereas in Section 2.2, we recall the definition of the collision operator and we give a precise definition of the randomized collision operator. The concept of a factorized solution is reviewed in Section 2.3.

In Section 3, we prove the main randomized spacetime estimate in the strong form given by (3). This is the content of Theorem 3.1. In Section 3.1, we explore a

different form of randomization, where the initial data is now random and we explain the difficulties of this approach in the context of the Gross–Pitaevskii hierarchy.

Section 4 is devoted to the study of the randomized Gross–Pitaevskii hierarchy. More precisely, in Section 4.1, we explore the connection between the randomized Gross–Pitaevskii hierarchy and the nonlinear Schrödinger equation. In Remark 4.1, we comment on the boardgame argument in the context of the randomized Gross– Pitaevskii hierarchy.

The independently randomized Gross–Pitaevskii hierarchy is precisely defined in Section 5. In Section 5.1, we explain some of the aspects in which this hierarchy is different from the (dependently) randomized Gross–Pitaevskii hierarchy as well as from the deterministic Gross–Pitaevskii hierarchy. In the study of the independently randomized Gross–Pitaevskii hierarchy, it is possible to apply the randomized spatial estimate given by Theorem 3.1. This is done in Section 5.2. As a result, we obtain convergence to zero of a sequence of Duhamel terms in a low-regularity space containing a random component in Theorem 5.2.

We revisit the (dependently) randomized Gross–Pitaevskii hierarchy in Section 6. In particular, in Section 6.1, we give an example showing that the methods used to prove Theorem 3.1 do not apply to the study of higher-order Duhamel expansions. In Section 6.2, we explicitly write out a Duhamel expansion of order ℓ . We also give a specific example of the expansion when $\ell = 3$ for illustration purposes. In Section 6.3, we define a class \mathcal{N} of *non-resonant* density matrices and, and we henceforth study Duhamel expansions corresponding to the dependently randomized Gross–Pitaevskii hierarchy of elements in this non-resonant class. More precisely, we prove a randomized spatial estimate for Duhamel expansions of arbitrary order starting from a density matrix in \mathcal{N} in Section 6.4. As a result, it is possible to deduce a convergence to zero of these Duhamel expansions in a low-regularity space containing a random component. This result is given in Theorem 6.4 of Section 6.5.

2. Notation

Let us first introduce some notation. Given two positive quantities A and B, we write $A \leq B$ if there exists some constant C > 0 such that $A \leq CB$. If the quantity C depends on q, we write C as C(q). We write $A \leq C(q)B$ also as $A = \mathcal{O}_q(B)$ or as $A \leq_q B$. If $A \leq_q B$ and $B \leq_q A$, we write $A \sim_q B$. In our paper, we sometimes abbreviate the Gross–Pitaevskii hierarchy as the *GP hierarchy*.

2.1. Fourier Transform and Differentiation of Density Matrices

Throughout our paper, we fix the spatial domain Λ to be the three-dimensional torus \mathbb{T}^3 , unless it is otherwise specified. Given $f \in L^2(\Lambda)$ and $\xi \in \mathbb{Z}^3$, we define the Fourier transform of f evaluated at ξ by:

$$\widehat{f}(\xi) := \int_{\Lambda} f(x) e^{-i \langle x, \xi \rangle} \,\mathrm{d}x.$$

Here, the quantity $\langle \cdot, \cdot \rangle$ denotes the inner product on \mathbb{R}^3 .

When considering density matrices $\gamma_0^{(k)} : \Lambda^k \times \Lambda^k \to \mathbb{C}$, we use the same convention as in [92] to define:

$$\left(\gamma_0^{(k)}\right) \cap \left(\vec{\xi}_k; \vec{\xi}_k'\right) := \int_{\Lambda^k \times \Lambda^k} \gamma_0^{(k)} \left(\vec{x}_k; \vec{x}_k'\right) e^{-i \cdot \sum_{j=1}^k \langle x_j, \xi_j \rangle + i \cdot \sum_{j=1}^k \langle x_j', \xi_j' \rangle} \, \mathrm{d}\vec{x}_k \, \mathrm{d}\vec{x}_k'$$

for $\vec{\xi}_k = (\xi_1, \dots, \xi_k)$, $\vec{\xi}'_k = (\xi'_1, \dots, \xi'_k) \in (\mathbb{Z}^3)^k$. In this way, the definition of the Fourier transform of density matrices is consistent with the definition of factorized solutions for the GP hierarchy (1). In what follows, we will usually write $(\gamma_0^{(k)})^{\widehat{}}$ as $\widehat{\gamma_0}^{(k)}$ for simplicity of notation. When a density matrix involves more operations, such as differentiation, we will use the first notation of the Fourier transform as in (20) below.

The *free evolution* corresponding to $i\partial_t + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})$, acting on density matrices of order k is defined as:

$$\mathcal{U}^{(k)}(t)\gamma_0^{(k)} := e^{it\sum_{j=1}^k \Delta_{x_j}}\gamma_0^{(k)}e^{-it\sum_{j=1}^k \Delta_{x'_j}}.$$

By construction, it is then the case that:

$$\left(i\partial_t + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\right)\mathcal{U}^{(k)}(t)\gamma_0^{(k)} = 0.$$

Given a regularity parameter α , we define the operator of *differentiation of order* α on density matrices of order k by using the Fourier transform:

$$(S^{(k,\alpha)}\gamma_0^{(k)})^{\widehat{}}(\xi_1,\ldots,\xi_k;\xi_1',\ldots,\xi_k') := \langle \xi_1 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \widehat{\gamma_0}^{(k)}(\xi_1,\ldots,\xi_k;\xi_1',\ldots,\xi_k').$$
(20)

Here, we define the Japanese bracket to be

$$\langle x \rangle := \sqrt{1 + |x|^2}.$$

In other words, we can write:

$$S^{(k,\alpha)}\gamma_0^{(k)} = \prod_{j=1}^k (1 - \Delta_{x_j})^{\frac{\alpha}{2}} (1 - \Delta_{x'_j})^{\frac{\alpha}{2}} \gamma_0^{(k)}.$$

2.2. The Collision Operator and the Randomized Collision Operator

Let us recall the definition of the *collision operator* $B_{j,k+1}$ for $k \in \mathbb{N}$ and for $j \in \{1, ..., k\}$. The operator $B_{j,k+1}$ acts on density matrices of order k + 1 and it is defined by:

$$B_{j,k+1} = B_{j,k+1}^+ - B_{j,k+1}^-,$$

where:

$$B_{j,k+1}^+(\gamma_0^{(k+1)})(\vec{x}_k;\vec{x}_k') := \int_{\Lambda} \delta(x_j - x_{k+1})\gamma_0^{(k+1)}(\vec{x}_k, x_{k+1};\vec{x}_k', x_{k+1}) \, \mathrm{d}x_{k+1}$$

and

$$B_{j,k+1}^{-}(\gamma_{0}^{(k+1)})(\vec{x}_{k};\vec{x}_{k}') := \int_{\Lambda} \delta\left(x_{j}'-x_{k+1}\right) \gamma_{0}^{(k+1)}\left(\vec{x}_{k},x_{k+1};\vec{x}_{k}',x_{k+1}\right) \, \mathrm{d}x_{k+1}.$$

Hence, for example, when j = 1:

$$B_{1,k+1}(\gamma_0^{(k+1)})(\vec{x}_k;\vec{x}'_k) = \gamma_0^{(k+1)}(\vec{x}_k,x_1;\vec{x}'_k,x_1) - \gamma_0^{(k+1)}(\vec{x}_k,x'_1;\vec{x}'_k,x'_1).$$

The calculation for general $j \in \{1, ..., k\}$ is similar. In particular, we note that $B_{j,k+1}$ is a linear map which takes density matrices of order k + 1 to density matrices of order k.

One can then compute that the Fourier transform of $B_{1,k+1}^+(\gamma_0^{(k+1)})$ is given by:

$$\begin{pmatrix} B_{1,k+1}^{+}\gamma_{0}^{(k+1)} \end{pmatrix} \widehat{} \left(\vec{\xi}_{k}; \vec{\xi}'_{k} \right)$$

$$= \sum_{\xi_{k+1}, \, \xi'_{k+1} \in \mathbb{Z}^{3}} \widehat{\gamma_{0}}^{(k+1)} \left(\xi_{1} - \xi_{k+1} + \xi'_{k+1}, \xi_{2}, \dots, \xi_{k}, \xi_{k+1}; \xi'_{1}, \dots, \xi'_{k}, \xi'_{k+1} \right).$$

$$(21)$$

Furthermore, the Fourier transform of $B_{1,k+1}^{-}\gamma_0^{(k+1)}$ is given by:

$$\begin{pmatrix} B_{1,k+1}^{-}\gamma_{0}^{(k+1)} \end{pmatrix} \widehat{(\vec{\xi}_{k};\vec{\xi}'_{k})} \\ = \sum_{\xi_{k+1},\,\xi'_{k+1}\in\mathbb{Z}^{3}} \widehat{\gamma_{0}}^{(k+1)}\left(\xi_{1},\ldots,\xi_{k},\xi_{k+1};\xi_{1}'-\xi_{k+1}'+\xi_{k+1},\xi_{2}',\ldots,\xi_{k}',\xi_{k+1}'\right).$$

$$(22)$$

We note that in the calculation for $B_{1,k+1}^-$ the ξ_{k+1} and ξ'_{k+1} come with a different sign, which is a consequence of our definition of the Fourier transform of a density matrix. The calculation for general $B_{j,k+1}^{\pm}$ is similar.

We now *randomize* the collision operator $B_{j,k+1}$ in order to obtain the *randomized collision operator* $[B_{j,k+1}]^{\omega}$. In order to define the latter operator, we use the Fourier transform. Let us take $(h_{\xi})_{\xi \in \mathbb{Z}^3}$ to be a sequence of independent, identically distributed Bernoulli random variables with expectation zero and standard deviation 1. We will use this notation from now on, unless we specify otherwise. Throughout our paper, the probability space associated with this sequence of random variables is denoted by (Ω, Σ, p) , where Σ is the corresponding sigma-algebra and where p is the probability measure. We will usually denote the probability space just by Ω .

With the above notation, we define, for fixed $\omega \in \Omega$ and for a fixed density matrix $\gamma_0^{(k+1)}$ of order k + 1:

$$\begin{pmatrix} \begin{bmatrix} B_{1,k+1}^{+} \end{bmatrix}^{\omega} \gamma_{0}^{(k+1)} \end{pmatrix} \widehat{} \begin{pmatrix} \vec{\xi}_{k}; \vec{\xi}_{k}' \end{pmatrix} \coloneqq \sum_{\xi_{k+1}, \xi_{k+1}' \in \mathbb{Z}^{3}} h_{\xi_{1}}(\omega) \cdot h_{\xi_{1}-\xi_{k+1}+\xi_{k+1}'}(\omega) \cdot h_{\xi_{k+1}}(\omega) \cdot h_{\xi_{k+1}'}(\omega) \cdot \widehat{\gamma_{0}}^{(k+1)} \begin{pmatrix} \xi_{1} - \xi_{k+1} + \xi_{k+1}', \xi_{2}, \dots, \xi_{k}, \xi_{k+1}; \xi_{1}', \dots, \xi_{k}', \xi_{k+1}' \end{pmatrix}.$$
(23)

In other words, we are just randomizing a subset of frequencies in (21). Analogously, we use (22) and we define $[B_{1,k+1}^-]^{\omega}$ by:

$$\begin{pmatrix} \begin{bmatrix} B_{1,k+1}^{-} \end{bmatrix}^{\omega} \gamma_{0}^{(k+1)} \end{pmatrix} \uparrow \begin{pmatrix} \vec{\xi}_{k}; \vec{\xi}_{k}' \end{pmatrix} \coloneqq \sum_{\xi_{k+1}, \xi_{k+1}' \in \mathbb{Z}^{3}} h_{\xi_{1}}(\omega) \cdot h_{\xi_{1}-\xi_{k+1}'+\xi_{k+1}}(\omega) \cdot h_{\xi_{k+1}}(\omega) \cdot h_{\xi_{k+1}'}(\omega) \cdot \widehat{\gamma_{0}}^{(k+1)} \begin{pmatrix} \xi_{1}, \dots, \xi_{k}, \xi_{k+1}; \xi_{1}' - \xi_{k+1}' + \xi_{k+1}, \xi_{2}', \dots, \xi_{k}', \xi_{k+1}' \end{pmatrix}.$$
(24)

The quantity $[B_{j,k+1}^{\pm}]^{\omega}$ is defined similarly, when $1 \leq j \leq k$. We can now define the *randomized collision operator* as:

$$[B_{j,k+1}]^{\omega} := \left[B_{j,k+1}^{+}\right]^{\omega} - \left[B_{j,k+1}^{-}\right]^{\omega}.$$
(25)

By construction, it follows that $[B_{j,k+1}]^{\omega} = B_{j,k+1}$, if $\omega \in \Omega$ is chosen such that $h_{\xi}(\omega) = 1$ for all $\xi \in \mathbb{Z}^3$ or if $h_{\xi}(\omega) = -1$ for all $\xi \in \mathbb{Z}^3$.

We define the *full randomized collision operator* by:

$$[B^{(k)}]^{\omega} := \sum_{j=1}^{k} [B_{j,k+1}]^{\omega}.$$
(26)

As a convention, we will extend the definition of the above collision operators which act on density matrices of order k+1 (that is $B_{j,k+1}$ and $[B_{j,k+1}]^{\omega}$) to density matrices $\sigma^{(\ell)}$ of order $\ell > k + 1$. This is done by acting only in the variables \vec{x}_{k+1} and \vec{x}'_{k+1} . We will use this convention in the discussion in Section 6.2.

We will sometimes write a density matrix $\gamma_0^{(k)}$ in terms of the standard Fourier basis of $L^2(\Lambda^k \times \Lambda^k)$ as follows:

$$\gamma_0^{(k)} = \sum_{\ell_r, j_r \in \mathbb{Z}^3} a_{\ell_1, \dots, \ell_k; j_1, \dots, j_k}^{(k+1)} \cdot b_{\ell_1, \dots, \ell_k; j_1, \dots, j_k}^{(k+1)}.$$
(27)

Here, $a_{\ell_1,\ldots,\ell_k;\ j_1,\ldots,j_k}^{(k)} = \widehat{\gamma_0}^{(k)}(\ell_1,\ldots,\ell_k;\ j_1,\ldots,j_k)$ denotes the Fourier coefficient and $b_{\ell_1,\ldots,\ell_k;\ j_1,\ldots,j_k}^{(k)}(\vec{x}_{k+1};\ \vec{x}'_{k+1}) = e^{i(x_1\cdot\ell_1+\cdots+x_k\cdot\ell_k)-i(x'_1\cdot j_1+\cdots+x'_k\cdot j_k)}$ denotes the canonical Fourier basis element.

2.3. Factorized Solutions

Let us now precisely explain the notion of a *factorized solution* to the Gross–Pitaevskii hierarchy (1). Suppose that the function $\phi = \phi(x, t)$ is a solution to the defocusing cubic nonlinear Schrödinger equation on Λ :

$$\begin{cases} i\partial_t \phi + \Delta \phi = |\phi|^2 \phi \\ \phi \big|_{t=0} = \phi_0. \end{cases}$$

The *Dirac bracket* $|\cdot\rangle\langle\cdot|$ is defined as $|f\rangle\langle g|(x, x') := f(x) \cdot g(x')$, for functions $f, g: \Lambda \to \mathbb{C}$. Then, using the definition of the collision operator above, it can be

shown that the sequence $(\gamma^{(k)})_k = (\gamma^{(k)}(t))_k$ of time-dependent density matrices given by:

$$\gamma^{(k)}\left(t,\vec{x}_{k};\vec{x}_{k}'\right) := \prod_{j=1}^{k} \phi(t,x_{j}) \overline{\phi\left(t,x_{j}'\right)} = |\phi\rangle\langle\phi|^{\otimes k}\left(t,\vec{x}_{k};\vec{x}_{k}'\right)$$

solves (1) with the initial data $\gamma_0^{(k)} = |\phi_0\rangle \langle \phi_0|^{\otimes k}$. We define these to be the *factor-ized solutions* of the GP hierarchy.

3. The Randomized Spacetime Estimate

In this section, we prove the main estimate for the randomized collision operator defined in (25). As we will see, the randomization will let us extend the range of regularity exponent. Let us recall, that in [92], it was proved that for the deterministic collision operator, for all $\alpha > 1$, there exists C > 0 such that for all $k \in \mathbb{N}$ and for all $1 \le j \le k$:

$$\| S^{(k,\alpha)} B_{j,k+1} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2([0,2\pi] \times \Lambda^k \times \Lambda^k)}$$

 $\leq C \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}.$

The *C* depends on α , but is independent of *j* and *k*. The condition $\alpha > 1$ is sharp for general density matrices. If we took factorized density matrices, we saw that we could take $\alpha = 1$.

We now see how the range of α can be extended if we replace $B_{j,k+1}$ by $[B_{j,k+1}]^{\omega}$. In doing so, we also have to add an Ω component to the norm on the left-hand side. The main result of this section is the following stronger result:

Theorem 3.1. Let $\alpha > \frac{3}{4}$ be given. For all $k \in \mathbb{N}$ and for all $1 \leq j \leq k$, the following bound holds:

$$\|S^{(k,\alpha)}[B_{j,k+1}]^{\omega}\gamma_{0}^{(k+1)}\|_{L^{2}(\Omega\times\Lambda^{k}\times\Lambda^{k})} \leq C_{0}\|S^{(k+1,\alpha)}\gamma_{0}^{(k+1)}\|_{L^{2}(\Lambda^{k+1}\times\Lambda^{k+1})}.$$

The constant $C_0 > 0$ depends on α , but it is independent of j, k. In particular, for $[B^{(k+1)}]^{\omega}$ as defined in (26), it follows that:

$$\|S^{(k,\alpha)}[B^{(k+1)}]^{\omega}\gamma_{0}^{(k+1)}\|_{L^{2}(\Omega\times\Lambda^{k}\times\Lambda^{k})} \leq C_{0}k \|S^{(k+1,\alpha)}\gamma_{0}^{(k+1)}\|_{L^{2}(\Lambda^{k+1}\times\Lambda^{k+1})}.$$

From Theorem 3.1, Fubini's Theorem and from the unitarity of $\mathcal{U}^{(k+1)}(t)$, we can deduce:

Corollary 3.2. Let $\alpha > \frac{3}{4}$ be given. For all T > 0, $k \in \mathbb{N}$ and $1 \le j \le k$, the following bound holds:

$$\| S^{(k,\alpha)} [B_{j,k+1}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2(\Omega \times [0,T] \times \Lambda^k \times \Lambda^k)}$$

 $\leq C_0 \sqrt{T} \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}.$

Here, $C_0 > 0$ *is the constant from Theorem* 3.1*. Hence, we can deduce that:*

$$\| S^{(k,\alpha)} [B^{(k+1)}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2(\Omega \times [0,T] \times \Lambda^k \times \Lambda^k)} \leq C_0 k \sqrt{T} \| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}.$$

Corollary 3.2 gives a direct extension of the spacetime bound from [92] in the randomized setting. In this setting, we see that we can prove a bound when the regularity exponent satisfies $\alpha > \frac{3}{4}$. This is an improvement over the result in the deterministic setting. Let us remark that in the proof of the deterministic spacetime bound in our previous joint work with Gressman, we encountered a sum of *diagonal terms* which was bounded precisely when $\alpha > \frac{3}{4}$. It was in the off-diagonal terms that the regularity was lost. We refer the interested reader to [92] for more details.

By using Markov's inequality and Corollary 3.2, we can deduce:

Corollary 3.3. For $\alpha > \frac{3}{4}$, T > 0, $k \in \mathbb{N}$, $1 \le j \le k$, and $\lambda > 0$ the following bound holds:

$$\mathbb{P}\Big(\|S^{(k,\alpha)}[B_{j,k+1}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2([0,T] \times \Lambda^k \times \Lambda^k)} \ge \lambda \Big) \\ \le \frac{C_0^2 T \|S^{(k+1,\alpha)} \gamma_0^{(k+1)}\|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2}{\lambda^2} .$$

Here, $C_0 > 0$ *is the constant from Theorem* 3.1. *Moreover,*

$$\mathbb{P}\Big(\|S^{(k,\alpha)}[B^{(k+1)}]^{\omega} \mathcal{U}^{(k+1)}(t) \gamma_0^{(k+1)} \|_{L^2([0,T] \times \Lambda^k \times \Lambda^k)} \ge \lambda \Big) \\ \le \frac{C_0^2 k^2 T \|S^{(k+1,\alpha)} \gamma_0^{(k+1)}\|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2}{\lambda^2} \,.$$

Remark 3.4. Let us note that we are just considering the analogue of (16) when p = 2 in the context of the Gross–Pitaevskii hierarchy. This already allows us to improve the range of α from the deterministic case. As we will see, the calculations when p = 2 are already quite involved. This is due to the fact that there are many possible pairings in the proof of Theorem 3.1. It is possible that one can extend this calculation to the case when p is an arbitrary even number, which would give us an exponential upper bound on the right-hand side of the estimate in Corollary 3.3, see [33]. The calculations for higher p would become extremely difficult. We will not pursue this issue in our paper.

We note that Corollary 3.3 gives a large deviation bound for the quantity that one is interested in looking at in the deterministic setting. As was noted in the introduction, this is one of the key steps in the use of randomization techniques in the analysis of nonlinear dispersive PDE. Since we would typically like to apply the spacetime estimate many times, we will see that the averaged estimate given in Corollary 3.2 will be easier to apply in the discussion that follows.

The proof of Theorem 3.1 is based on the pairing of the frequencies, similarly as in the proof of (16), which was explained in the introduction. There are many more cases to consider in our setting. In some of the cases, we use shorthand matrix notation which is defined in Definition 3.5 below. We now prove Theorem 3.1. **Proof.** It suffices to consider the special case when j = 1. The general case follows by analogous arguments. Moreover, we will prove the bound in the theorem for the operator $[B_{1,k+1}^+]^{\omega}$. The bound for $[B_{1,k+1}^-]^{\omega}$ is proved in the same way, and we thus obtain the claim for $[B_{1,k+1}]^{\omega} = [B_{1,k+1}^+]^{\omega} - [B_{1,k+1}^-]^{\omega}$. We compute:

$$\begin{pmatrix} S^{(k,\alpha)}[B_{1,k+1}^+]^{\omega}\gamma_0^{(k+1)} \end{pmatrix} \widehat{} \left(\vec{\xi}_k; \vec{\xi}'_k \right)$$

$$= \prod_{j=1}^k \langle \xi_j \rangle^{\alpha} \cdot \langle \xi'_j \rangle^{\alpha} \cdot \left(\sum_{\eta_1, \eta'_1 \in \mathbb{Z}^3} h_{\xi_1}(\omega) \cdot h_{\xi_1 - \eta_1 + \eta'_1}(\omega) \cdot h_{\eta_1}(\omega) \cdot h_{\eta'_1}(\omega) \right)$$

$$\cdot \widehat{\gamma_0}^{(k+1)} \left(\xi_1 - \eta_1 + \eta'_1, \xi_2, \dots, \xi_k, \eta_1; \xi'_1, \dots, \xi'_k, \eta'_1 \right)$$

which by (27) equals

$$\prod_{j=1}^{k} \langle \xi_{j} \rangle^{\alpha} \cdot \langle \xi_{j}' \rangle^{\alpha} \cdot \bigg(\sum_{\eta_{1}, \eta_{1}' \in \mathbb{Z}^{3}} h_{\xi_{1}}(\omega) \cdot h_{\xi_{1}-\eta_{1}+\eta_{1}'}(\omega) \cdot h_{\eta_{1}}(\omega) \cdot h_{\eta_{1}'}(\omega) \\ \cdot a_{\xi_{1}-\eta_{1}+\eta_{1}', \xi_{2}, \dots, \xi_{k}, \eta_{1}; \xi_{1}', \dots, \xi_{k}', \eta_{1}'} \bigg).$$

Consequently:

$$\begin{split} \big| (S^{(k,\alpha)}[B^+_{1,k+1}]^{\omega} \gamma_0^{(k+1)}) \widehat{(\vec{\xi}_k;\vec{\xi}'_k)} \big|^2 \\ &= \prod_{j=1}^k \langle \xi_j \rangle^{2\alpha} \cdot \langle \xi'_j \rangle^{2\alpha} \cdot \bigg(\sum_{\eta_1,\eta'_1,\eta_2,\eta'_2 \in \mathbb{Z}^3} h_{\xi_1}(\omega) \cdot h_{\xi_1-\eta_1+\eta'_1}(\omega) \cdot h_{\eta_1}(\omega) \cdot h_{\eta'_1}(\omega) \\ &\cdot \overline{h}_{\xi_1}(\omega) \cdot \overline{h}_{\xi_1-\eta_2+\eta'_2}(\omega) \cdot \overline{h}_{\eta_2}(\omega) \cdot \overline{h}_{\eta'_2}(\omega) \cdot a^{(k+1)}_{\xi_1-\eta_1+\eta'_1,\xi_2,\dots,\xi_k,\eta_1;\xi'_1,\dots,\xi'_k,\eta'_1} \\ &\cdot \overline{a}^{(k+1)}_{\xi_1-\eta_2+\eta'_2,\xi_2,\dots,\xi_k,\eta_2;\xi'_1,\dots,\xi'_k,\eta'_2} \bigg). \end{split}$$

Since we are using Bernoulli random variables, we note that $\overline{h} = h$ and $h^2 = 1$. Hence, the previous expression equals:

$$\begin{split} &\prod_{j=1}^{k} \langle \xi_{j} \rangle^{2\alpha} \cdot \langle \xi_{j}' \rangle^{2\alpha} \cdot \Big(\sum_{\eta_{1}, \eta_{1}', \eta_{2}, \eta_{2}' \in \mathbb{Z}^{3}} h_{\xi_{1} - \eta_{1} + \eta_{1}'}(\omega) \cdot h_{\eta_{1}}(\omega) \cdot h_{\eta_{1}'}(\omega) \cdot h_{\xi_{1} - \eta_{2} + \eta_{2}'}(\omega) \\ & \cdot h_{\eta_{2}}(\omega) \cdot h_{\eta_{2}'}(\omega) \cdot a_{\xi_{1} - \eta_{1} + \eta_{1}', \xi_{2}, \dots, \xi_{k}, \eta_{1}; \xi_{1}', \dots, \xi_{k}', \eta_{1}'} \cdot \overline{a}_{\xi_{1} - \eta_{2} + \eta_{2}', \xi_{2}, \dots, \xi_{k}, \eta_{2}; \xi_{1}', \dots, \xi_{k}', \eta_{2}'} \Big) \\ &= \langle \xi_{1} \rangle^{2\alpha} \cdot \Big(\sum_{\eta_{1}, \eta_{1}', \eta_{2}, \eta_{2}' \in \mathbb{Z}^{3}} h_{\xi_{1} - \eta_{1} + \eta_{1}'}(\omega) \cdot h_{\eta_{1}}(\omega) \cdot h_{\eta_{1}'}(\omega) \cdot h_{\xi_{1} - \eta_{2} + \eta_{2}'}(\omega) \cdot h_{\eta_{2}}(\omega) \\ & \cdot h_{\eta_{2}'}(\omega) \cdot \Big[\langle \xi_{2} \rangle^{\alpha} \cdot \dots \cdot \langle \xi_{k} \rangle^{\alpha} \cdot \langle \xi_{1}' \rangle^{\alpha} \cdot \dots \cdot \langle \xi_{k}' \rangle^{\alpha} \cdot a_{\xi_{1} - \eta_{1} + \eta_{1}', \xi_{2}, \dots, \xi_{k}, \eta_{1}; \xi_{1}', \dots, \xi_{k}', \eta_{1}'} \Big] \\ & \cdot \Big[\langle \xi_{2} \rangle^{\alpha} \cdot \dots \cdot \langle \xi_{k} \rangle^{\alpha} \cdot \langle \xi_{1}' \rangle^{\alpha} \cdot \dots \cdot \langle \xi_{k}' \rangle^{\alpha} \cdot \overline{a}_{\xi_{1} - \eta_{2} + \eta_{2}', \xi_{2}, \dots, \xi_{k}, \eta_{2}; \xi_{1}', \dots, \xi_{k}', \eta_{2}'} \Big] \Big). \end{split}$$

By Plancherel's Theorem in the x-variables, it follows that:

$$\begin{split} I &:= \left\| S^{(k,\alpha)} [B^+_{1,k+1}]^{\omega} \gamma_0^{(k+1)} \right\|_{L^2(\Omega \times \Lambda^k \times \Lambda^k)}^2 = \\ &= \sum_{\xi_1, \dots, \xi_k, \xi'_1, \dots, \xi'_k \in \mathbb{Z}^3} \langle \xi_1 \rangle^{2\alpha} \cdot \left(\sum_{\eta_1, \eta'_1, \eta_2, \eta'_2 \in \mathbb{Z}^3} \int_{\Omega} h_{\xi_1 - \eta_1 + \eta'_1}(\omega) \cdot h_{\eta_1}(\omega) \cdot h_{\eta'_1}(\omega) \right) \\ &\cdot h_{\xi_1 - \eta_2 + \eta'_2}(\omega) \cdot h_{\eta_2}(\omega) \cdot h_{\eta'_2}(\omega) \\ &\cdot [\langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi'_1 \rangle \cdots \langle \xi'_k \rangle^{\alpha} \cdot a^{(k+1)}_{\xi_1 - \eta_1 + \eta'_1, \xi_2, \dots, \xi_k, \eta_1; \xi'_1, \dots, \xi'_k, \eta'_1}] \\ &\cdot [\langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi'_1 \rangle \cdots \langle \xi'_k \rangle^{\alpha} \cdot \overline{a}^{(k+1)}_{\xi_1 - \eta_2 + \eta'_2, \xi_2, \dots, \xi_k, \eta_2; \xi'_1, \dots, \xi'_k, \eta'_2}] \, dp(\omega) \bigg). \end{split}$$

This is the expression that we want to estimate. By the independence of the (h_{ξ}) , and by the fact that they all have mean zero, it follows that each element in the set $\{\xi_1 - \eta_1 + \eta'_1, \eta_1, \eta'_1, \xi_1 - \eta_2 + \eta'_2, \eta_2, \eta'_2\}$ occurs at least twice in the list $(\xi_1 - \eta_1 + \eta'_1, \eta_1, \eta'_1, \xi_1 - \eta_2 + \eta'_2, \eta_2, \eta'_2)$.

Let's call this property (*).

Since we are working with Bernoulli random variables, we note that, by the triangle inequality:

$$I \leq \sum_{\xi_{1},...,\xi_{k},\xi_{1}',...,\xi_{k}',\eta_{1},\eta_{1}',\eta_{2},\eta_{2}'\in\mathbb{Z}^{3}} \left[\langle \xi_{1} \rangle^{\alpha} \cdot \langle \xi_{2} \rangle^{\alpha} \cdots \langle \xi_{k} \rangle^{\alpha} \cdot \langle \xi_{1}' \rangle^{\alpha} \cdots \langle \xi_{k}' \rangle^{\alpha} \cdot |a_{\xi_{1}-\eta_{1}+\eta_{1}',\xi_{2},...,\xi_{k},\eta_{1};\xi_{1}',...,\xi_{k}',\eta_{1}'}| \right] \\ \left[\langle \xi_{1} \rangle^{\alpha} \cdot \langle \xi_{2} \rangle^{\alpha} \cdots \langle \xi_{k} \rangle^{\alpha} \cdot \langle \xi_{1}' \rangle^{\alpha} \cdots \langle \xi_{k}' \rangle^{\alpha} \cdot |\overline{a}_{\xi_{1}-\eta_{2}+\eta_{2}',\xi_{2},...,\xi_{k},\eta_{2};\xi_{1}',...,\xi_{k}',\eta_{2}'}| \right].$$
(28)

Here,

$$\sum_{\xi_1,...,\xi_k,\xi_1',...,\xi_k',\eta_1,\eta_1',\eta_2,\eta_2'\in\mathbb{Z}^3 }^{(*)}$$

denotes the sum over all $\xi_1, \ldots, \xi_k, \xi'_1, \ldots, \xi'_k, \eta_1, \eta'_1, \eta_2, \eta'_2 \in \mathbb{Z}^3$ satisfying the condition (*) above. In particular, we see that we need to bound the sum in the frequencies under the given constraint.

We now need to consider all of the possible cases separately. This requires a combinatorial analysis of all the possible pairings. In order to simplify the notation, we develop a matrix notation for the obtained sums. The notation is explained on an example in Definition 3.5 and is used in order to analyze the subsequent cases. Big Case I $\xi_1 - \eta_1 + \eta'_1 = \xi_1 - \eta_2 + \eta'_2$. In this big case, we obtain that:

 $\eta_1' + \eta_2 = \eta_1 + \eta_2'. \tag{29}$

We now need to consider several possibilities:

Case IA { η_1 , η'_1 , η_2 , η'_2 } has cardinality 2; each element in the list (η_1 , η'_1 , η_2 , η'_2) occurs exactly twice.

Let us observe that:

$$\langle \xi_1
angle^{lpha} \lesssim \langle \xi_1 - \eta_1 + \eta_1'
angle^{lpha} \cdot \langle \eta_1
angle^{lpha} \cdot \langle \eta_1'
angle^{lpha}$$

and

$$\langle \xi_1 \rangle^{lpha} \lesssim \langle \xi_1 - \eta_2 + \eta_2' \rangle^{lpha} \cdot \langle \eta_2 \rangle^{lpha} \cdot \langle \eta_2' \rangle^{lpha}.$$

We henceforth refer to such inequalities as the fractional Leibniz Rule.

It follows from (28) that:

$$I \leq \sum_{\substack{\xi_1, \dots, \xi_k, \xi'_1, \dots, \xi'_k, \eta_1, \eta'_1, \eta_2, \eta'_2 \in \mathbb{Z}^3 \\ \cdot \left[\langle \xi_1 - \eta_1 + \eta'_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \xi'_1 \rangle^{\alpha} \cdots \langle \xi'_k \rangle^{\alpha} \cdot \langle \eta'_1 \rangle^{\alpha} \\ \cdot |a_{\xi_1 - \eta_1 + \eta'_1, \xi_2, \dots, \xi_k, \eta_1; \xi'_1, \dots, \xi'_k, \eta'_1}| \right] \\ \cdot \left[\langle \xi_1 - \eta_2 + \eta'_2 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \xi'_1 \rangle^{\alpha} \cdots \langle \xi'_k \rangle^{\alpha} \cdot \langle \eta'_2 \rangle^{\alpha} \\ \cdot |\overline{a}_{\xi_1 - \eta_2 + \eta'_2, \xi_2, \dots, \xi_k, \eta_2; \xi'_1, \dots, \xi'_k, \eta'_2}| \right].$$

Let us denote the quantity on the right-hand side by J.

It follows that the corresponding contribution to J is:

$$\lesssim \sum_{\substack{\xi_1,\ldots,\xi_k,\xi_1',\ldots,\xi_k',\eta_1,\eta_1',\eta_2,\eta_2' \in \mathbb{Z}^3\\Case\ IA}} \left[\langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_1' \rangle^{\alpha} \cdot \left| a_{\xi_1 - \eta_1 + \eta_1',\xi_2,\ldots,\xi_k,\eta_1;\xi_1',\ldots,\xi_k',\eta_1' \right| \right]} \left[\langle \xi_1 - \eta_2 + \eta_2' \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_2' \rangle^{\alpha} \cdot \left| \overline{a}_{\xi_1 - \eta_2 + \eta_2',\xi_2,\ldots,\xi_k,\eta_2;\xi_1',\ldots,\xi_k',\eta_2'} \right| \right].$$

Here, by:

$$\sum_{\substack{\xi_1,...,\xi_k,\xi_1',...,\xi_k',\eta_1,\eta_1',\eta_2,\eta_2' \in \mathbb{Z}^3 \\ Case \ (IA)}}$$

we denote the sum over all frequencies which occur in Case (IA).

We need to consider two possibilities: IA1 $\eta_1 = \eta_2$, $\eta'_1 = \eta'_2$. We use the Cauchy–Schwarz inequality in η_1 , η'_1 , as well as in ξ_2 , ..., ξ_k , ξ'_1 , ..., ξ'_k and we use Plancherel's Theorem in order to deduce that the corresponding contribution to *J* is:

$$\lesssim \|S^{(k+1,\alpha)}\gamma_0^{(k+1)}\|_{L^2(\Lambda^{k+1}\times\Lambda^{k+1})}^2$$

IA2 $\eta_1 = \eta'_2, \ \eta'_1 = \eta_2.$

We note that $\xi_1 - \eta_1 + \eta'_1 = \xi_1 - \eta_2 + \eta'_2$ implies that $\eta_1 = \eta'_1 = \eta_2 = \eta'_2$. Hence, $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$ has cardinality 1. This contradicts the assumption that this set has cardinality 2 and hence this case does not occur.

IA3 $\eta_1 = \eta'_1$, $\eta_2 = \eta'_2$. In this case, we need to estimate:

$$\begin{split} \tilde{J} &:= \sum_{\substack{\xi_1, \dots, \xi_k, \xi_1', \dots, \xi_k', \eta_1, \eta_2 \in \mathbb{Z}^3}} \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \\ & \left| a_{\xi_1, \xi_2, \dots, \xi_k, \eta_1; \xi_1', \dots, \xi_k', \eta_1}^{(k+1)} \right| \right] \\ & \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \left| \overline{a}_{\xi_1, \xi_2, \dots, \xi_k, \eta_2; \xi_1', \dots, \xi_k', \eta_2}^{(k+1)} \right| \right] \end{split}$$

Now, we need to argue differently, since the fractional Leibniz Rule

$$\langle \xi_1 \rangle^{lpha} = \langle \xi_1 - \eta_1 + \eta_1' \rangle^{lpha} \lesssim \langle \xi_1 \rangle^{lpha} \cdot \langle \eta_1 \rangle^{lpha} \cdot \langle \eta_1' \rangle^{lpha}$$

does not give us a good upper bound. Therefore we leave the factor $\langle \xi_1 \rangle^{\alpha}$ by itself.

We can then rewrite \tilde{J} as:

$$\sum_{\xi_1,\ldots,\xi_k,\xi_1',\ldots,\xi_k',\eta_1,\eta_2\in\mathbb{Z}^3} \langle \eta_1 \rangle^{-2\alpha} \cdot \langle \eta_2 \rangle^{-2\alpha} \cdot \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \left| a_{\xi_1,\xi_2,\ldots,\xi_k,\eta_1;\xi_1',\ldots,\xi_k',\eta_1}^{(k+1)} \right| \right] \\ \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \left| \overline{a}_{\xi_1,\xi_2,\ldots,\xi_k,\eta_2;\xi_1',\ldots,\xi_k',\eta_2}^{(k+1)} \right| \right]$$

Let us note that, since $\alpha > \frac{3}{4}$, the sequence $(\langle q \rangle^{-2\alpha})_{q \in \mathbb{Z}^3}$ belongs to $\ell^2(\mathbb{Z}^3)$. Hence, we can use the Cauchy–Schwarz inequality in $\xi_1, \ldots, \xi_k, \xi'_1, \ldots, \xi'_k, \eta_1, \eta_2$ and Plancherel's Theorem in order to deduce that:

$$\tilde{J} \lesssim \|S^{(k+1,\alpha)}\gamma_0^{(k+1)}\|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2.$$
(30)

Case IB { η_1 , η'_1 , η_2 , η'_2 } has cardinality 2; one element in the list (η_1 , η'_1 , η_2 , η'_2) occurs three times.

We recall from (29) that $\eta'_1 + \eta_2 = \eta_1 + \eta'_2$. From this identity, it follows that if one element in the list $(\eta_1, \eta'_1, \eta_2, \eta'_2)$ occurs three times, then they all have to be equal. In this way, we obtain a contradiction to the assumption that $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$ has cardinality 2.

Case IC { η_1 , η'_1 , η_2 , η'_2 } has cardinality 3.

We recall property (*), and the fact that in Big Case *I*, one has: $\xi_1 - \eta_1 + \eta'_1 = \xi_1 - \eta_2 + \eta'_2$, from where it follows that each element in the list:

$$(\eta_1, \eta_2, \eta'_1, \eta'_2)$$

either occurs at least twice or it equals $\xi_1 - \eta_1 + \eta'_1$. In this way, we get a contradiction to the assumption that $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$ has cardinality 3. Namely, if $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$ had cardinality 3, the two distinct elements of the list $(\eta_1, \eta'_1, \eta_2, \eta'_2)$ which occur only once would both have to equal $\xi_1 - \eta_1 + \eta'_1$.

Case ID { η_1 , η'_1 , η_2 , η'_2 } has cardinality 4. In this case, we obtain a contradiction in the same way as in case IC). *Case IE* { η_1 , η'_1 , η_2 , η'_2 } has cardinality 1.

In this case, we note that:

$$\xi_1 - \eta_1 + \eta'_1 = \xi_1 - \eta_2 + \eta'_2 = \xi_1$$

and

$$\eta_1 = \eta_1' = \eta_2 = \eta_2'.$$

Hence, we need to estimate:

$$\sum_{\substack{\xi_1,\ldots,\xi_k,\xi_1',\ldots,\xi_k',\eta_1\in\mathbb{Z}^3}} \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \left| a_{\xi_1,\xi_2,\ldots,\xi_k,\eta_1;\xi_1',\ldots,\xi_k',\eta_1}^{(k+1)} \right| \right] \\ \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \left| \overline{a}_{\xi_1,\xi_2,\ldots,\xi_k,\eta_1;\xi_1',\ldots,\xi_k',\eta_1}^{(k+1)} \right| \right] \cdot$$

We can use the Cauchy–Schwarz inequality in $\xi_1, \ldots, \xi_k, \xi'_1, \ldots, \xi'_k, \eta_1$ and Plancherel's Theorem to deduce that the above expression is, in absolute value

$$\lesssim \left\| S^{(k+1,\alpha)} \gamma_0^{(k+1)} \right\|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2$$

Here, we only used the fact that $\alpha \ge 0$.

Consequently, we obtain the wanted bound in Big Case I.

In what follows, we may assume that:

$$\xi_1 - \eta_1 + \eta_1' \neq \xi_1 - \eta_2 + \eta_2'. \tag{31}$$

By condition (*), it follows that $\xi_1 - \eta_1 + \eta'_1$ and $\xi_1 - \eta_2 + \eta'_2$ each get paired with distinct elements of $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$. Big Case II $\eta_1 = \eta'_1$. In this Big Case, we obtain:

$$\xi_1 - \eta_1 + \eta'_1 = \xi_1$$

Furthermore, the condition (31) implies that:

$$\eta_2 \neq \eta_2'. \tag{32}$$

By using the condition (*), it follows that:

$$\eta_2, \eta'_2 \in \{\xi_1, \eta_1, \xi_1 - \eta_2 + \eta'_2\}.$$

We need to consider several cases:

Case IIA $\eta_2, \eta'_2 \in \{\xi_1, \eta_1\}$. In case IIA), we need to consider several subcases: $IIA1 \eta_2 = \xi_1, \eta'_2 = \eta_1$. In this subcase, we know: $\xi_1 - \eta_2 + \eta'_2 = \xi_1 - \xi_1 + \eta_1 = \eta_1$. In other words:

$$\begin{cases} \xi_1 - \eta_1 + \eta'_1 = \xi_1 \\ \xi_1 - \eta_2 + \eta'_2 = \eta_1 \\ \eta_1 = \eta'_1 \\ \eta_2 = \xi_1 \\ \eta'_2 = \eta_1 \end{cases}$$

It follows that we have to bound:

$$\sum_{\xi_1,\dots,\xi_k,\xi_1',\dots,\xi_k',\eta_1\in\mathbb{Z}^3} \left[\langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \eta_1 \rangle^0 \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^0 \cdot \left| a_{\xi_1,\xi_2,\dots,\xi_k,\eta_1;\xi_1',\dots,\xi_k',\eta_1}^{(k+1)} \right| \right] \\ \left[\langle \eta_1 \rangle^0 \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^0 \cdot \left| \overline{a}_{\eta_1,\xi_2,\dots,\xi_k,\xi_1;\xi_1',\dots,\xi_k',\eta_1}^{(k+1)} \right| \right].$$

$$(33)$$

We bound the above expression in absolute value by using the Cauchy–Schwarz inequality in ξ_1 , η_1 and in ξ_2 , ..., ξ_k , ξ'_1 , ..., ξ'_k to obtain the bound $\leq \|S^{(k+1,\alpha)} \gamma_0^{(k+1)}\|_{L^2(\Lambda^{k+1} \times \Lambda^{k+1})}^2$.

In what follows, we note that the application of the Cauchy–Schwarz inequality in $\xi_1, \ldots, \xi_k, \xi'_1, \ldots, \xi'_k$ is always straightforward. Let us now develop some shorthand notation:

Definition 3.5. Let C be a list of variables in $\{\xi_1, \eta_1, \eta'_1, \eta_2, \eta'_2\}$. Furthermore, let μ_j ; j = 1, ..., 6 be linear combinations of elements of the list C and let α_j ; j = 1, ..., 6 be elements of $\{0, \alpha\}$. We define the quantity:

$$\mathcal{C}; \begin{bmatrix} \langle \mu_1 \rangle^{\alpha_1} & \langle \mu_2 \rangle^{\alpha_2} & \langle \mu_3 \rangle^{\alpha_3} \\ \langle \mu_4 \rangle^{\alpha_4} & \langle \mu_5 \rangle^{\alpha_5} & \langle \mu_6 \rangle^{\alpha_6} \end{bmatrix}$$

to be the sum over the elements of the list C as well as over $\xi_2, \xi_3, \ldots, \xi_k, \xi'_1, \xi'_2, \ldots, \xi'_k$ of:

$$\begin{bmatrix} \langle \mu_1 \rangle^{\alpha_1} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \mu_2 \rangle^{\alpha_2} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \mu_3 \rangle^{\alpha_3} \\ \cdot |a_{\mu_1,\xi_2,\dots,\xi_k,\mu_2;\xi_1',\dots,\xi_k',\mu_3}^{(k+1)}| \end{bmatrix} \\ \begin{bmatrix} \langle \mu_4 \rangle^{\alpha_4} \cdot \langle \xi_2 \rangle^{\alpha} \cdots \langle \xi_k \rangle^{\alpha} \cdot \langle \mu_5 \rangle^{\alpha_5} \cdot \langle \xi_1' \rangle^{\alpha} \cdots \langle \xi_k' \rangle^{\alpha} \cdot \langle \mu_6 \rangle^{\alpha_6} \\ \cdot |\overline{a}_{\mu_4,\xi_2,\dots,\xi_k,\mu_5;\xi_1',\dots,\xi_k',\mu_6}| \end{bmatrix}.$$

Remark 3.6. In other words, the entries of the first row of the matrix correspond to the derivatives on the Fourier side applied to the first, (k + 1)-st, and (2k + 2)-nd component of the factor of $\widehat{\gamma_0}^{(k+1)}$ which is given by:

$$a_{\mu_1,\xi_2,\ldots,\xi_k,\mu_2;\xi_1',\ldots,\xi_k',\mu_3}^{(k+1)} = \widehat{\gamma_0}^{(k+1)} \left(\mu_1,\xi_2,\ldots,\xi_k,\mu_2;\xi_1',\ldots,\xi_k',\mu_3\right).$$

Hence, we apply derivatives on the Fourier side to the $\xi_1, \xi_{k+1}, \xi'_{k+1}$ components of this factor.

The second row of the matrix corresponds to the same derivatives of the factor of $\overline{\hat{\gamma}_0}^{(k+1)}$, which is given by:

$$\overline{a}_{\mu_4,\xi_2,\ldots,\xi_k,\mu_5;\xi_1',\ldots,\xi_k',\mu_6}^{(k+1)} = \overline{\widetilde{\gamma_0}}^{(k+1)} \left(\mu_4,\xi_2,\ldots,\xi_k,\mu_5;\xi_1',\ldots,\xi_k',\mu_6\right)$$

As before, we apply derivatives on the Fourier side to the ξ_1 , ξ_{k+1} , ξ'_{k+1} components of this factor.

In the definition, the variables which occur before the matrix denote the variables with respect to which we are summing. In this notation, we are excluding the variables $\xi_2, \ldots, \xi_k, \xi'_1, \xi_2, \ldots, \xi'_k$, since we noted above that the application of the Cauchy–Schwarz inequality in these variables is straightforward.

By using the the above definition, we can write the sum in (33) as:

$$\xi_1, \eta_1; \begin{bmatrix} \langle \xi_1 \rangle^{\alpha} & \langle \eta_1 \rangle^0 & \langle \eta_1 \rangle^0 \\ \langle \eta_1 \rangle^0 & \langle \xi_1 \rangle^{\alpha} & \langle \eta_1 \rangle^0 \end{bmatrix}.$$
(34)

We see that, in subcase IIA1), we can apply the Cauchy–Schwarz inequality in ξ_1 , η_1 and obtain the wanted bound since these variables occur in the first and in the second row of the matrix. With the above shorthand notation and with these observations in mind, we continue the analysis of the possible cases.

IIA2 $\eta_2 = \eta_1, \eta'_2 = \xi_1.$

In this subcase, we know that:

$$\begin{split} \xi_1 &- \eta_1 + \eta_1' = \xi_1 \\ \xi_1 &- \eta_2 + \eta_2' = 2\xi_1 - \eta_1 \\ \eta_1' &= \eta_1 \\ \eta_2 &= \eta_1 \\ \eta_2' &= \xi_1. \end{split}$$

Moreover, we use the fractional Leibniz rule in the second row in order to deduce that:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle 2\xi_1 \rangle^{\alpha} \lesssim \langle 2\xi_1 - \eta_1 \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \xi_1 \rangle^0.$$

Consequently, arguing similarly to Case IA, and taking into account the shorthand matrix notation, we need to consider:

$$\xi_1, \eta_1; \begin{bmatrix} \langle \xi_1 \rangle^{\alpha} & \langle \eta_1 \rangle^0 & \langle \eta_1 \rangle^0 \\ \langle 2\xi_1 - \eta_1 \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} & \langle \xi_1 \rangle^0 \end{bmatrix}.$$

We can now apply the Cauchy–Schwarz inequality in ξ_1 and then in η_1 to obtain the wanted bound in this subcase.

Case IIB $\eta_2 = \xi_1 - \eta_2 + \eta'_2$. We recall by (32) that $\eta'_2 \neq \eta_2$, so $\eta'_2 \neq \xi_1 - \eta_2 + \eta'_2$. By the condition (*), it follows that:

$$\eta'_2 \in \{\xi_1, \eta_1\}.$$

We need to consider two possible subcases: *IIB1* $\eta'_2 = \xi_1$. We obtain that: $\eta_2 = \xi_1 - \eta_2 + \xi_1$, which implies: $\eta_2 = \xi_1 = \eta'_2$. This is a contradiction, since we assumed that $\eta_2 \neq \eta'_2$. *IIB2* $\eta'_2 = \eta_1$. We recall that $\eta_2 = \xi_1 - \eta_2 + \eta'_2$, hence it follows that:

$$\eta_2 = \frac{\xi_1 + \eta_2'}{2} = \frac{\xi_1 + \eta_1}{2}$$

In this subcase, we know that:

$$\begin{cases} \xi_1 - \eta_1 + \eta'_1 = \xi_1 \\ \xi_1 - \eta_2 + \eta'_2 = \eta_2 = \frac{\xi_1 + \eta_1}{2} \\ \eta'_1 = \eta_1 \\ \eta_2 = \frac{\xi_1 + \eta_1}{2} \\ \eta'_2 = \eta_1. \end{cases}$$

We use the fractional Leibniz rule in the second row in order to observe that:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \left\langle \frac{\xi_1}{2} \right\rangle^{\alpha} \lesssim \left\langle \frac{\xi_1 + \eta_1}{2} \right\rangle^{\alpha} \cdot \left\langle \frac{\eta_1}{2} \right\rangle^{\alpha} \lesssim \left\langle \frac{\xi_1 + \eta_1}{2} \right\rangle^{\alpha} \cdot \left\langle \frac{\xi_1 + \eta_1}{2} \right\rangle^{0} \cdot \langle \eta_1 \rangle^{\alpha}.$$

Hence, we need to consider:

$$\xi_1, \eta_1; \left[\begin{pmatrix} \langle \xi_1 \rangle^{\alpha} & \langle \eta_1 \rangle^0 & \langle \eta_1 \rangle^0 \\ \left(\frac{\xi_1 + \eta_1}{2} \right)^{\alpha} & \left(\frac{\xi_1 + \eta_1}{2} \right)^0 & \langle \eta_1 \rangle^{\alpha} \end{bmatrix}.$$

We now obtain the wanted bound by first applying the Cauchy–Schwarz inequality, first in ξ_1 , and then in η_1 .

Case IIC $\eta'_2 = \xi_1 - \eta_2 + \eta'_2$. In this case, we obtain that $\xi_1 = \eta_2$. More precisely, we know that:

$$\begin{cases} \xi_1 - \eta_1 + \eta'_1 = \xi_1 \\ \xi_1 - \eta_2 + \eta'_2 = \eta'_2 \\ \eta'_1 = \eta_1 \\ \eta_2 = \xi_1. \end{cases}$$

We need to estimate:

$$\xi_1, \eta_1, \eta'_2; \begin{bmatrix} \langle \xi_1 \rangle^{\alpha} & \langle \eta_1 \rangle^{0} & \langle \eta_1 \rangle^{0} \\ \langle \eta'_2 \rangle^{0} & \langle \xi_1 \rangle^{\alpha} & \langle \eta'_2 \rangle^{0} \end{bmatrix}.$$

In this case, we cannot directly apply the Cauchy–Schwarz inequality in the variables η_1 , η'_2 since the pairs of $\langle \eta_1 \rangle^0$ and $\langle \eta'_2 \rangle^0$ occur in the same rows of the matrix. Instead, we need to argue as in subcase IA3. We multiply and divide the (1, 2) and (1, 3) entry of the matrix with $\langle \eta_1 \rangle^{\alpha}$. Furthermore, we multiply and divide the (2, 1) and (2, 3) entry of the matrix by $\langle \eta'_2 \rangle^{\alpha}$ to deduce that we have to consider:

$$\xi_{1}, \eta_{1}, \eta_{2}^{\prime}; \langle \eta_{1} \rangle^{-2\alpha} \cdot \langle \eta_{2}^{\prime} \rangle^{-2\alpha} \cdot \begin{bmatrix} \langle \xi_{1} \rangle^{\alpha} & \langle \eta_{1} \rangle^{\alpha} & \langle \eta_{1} \rangle^{\alpha} \\ \langle \eta_{2}^{\prime} \rangle^{\alpha} & \langle \xi_{1} \rangle^{\alpha} & \langle \eta_{2}^{\prime} \rangle^{\alpha} \end{bmatrix}.$$
(35)

In this notation, we mean that we multiply the result obtained from the matrix by $\langle \eta_1 \rangle^{-2\alpha} \cdot \langle \eta_2 \rangle^{-2\alpha}$ and we then sum in ξ_1 , η_1 , η'_2 . (Let us emphasize that we do not multiply each row of the matrix by $\langle \eta_1 \rangle^{-2\alpha} \cdot \langle \eta_2 \rangle^{-2\alpha}$.) We now obtain the desired bound in this subcase after we apply the Cauchy–Schwarz inequality in ξ_1 , η_1 , η'_2 . In order to bound the sum in η_1 , η'_2 , we need to use the assumption that $\alpha > \frac{3}{4}$ as in subcase IA3.

The bound in the big case when $\eta_1 = \eta'_1$ now follows. The big case when $\eta_2 = \eta'_2$ is analogous.

Big Case III $\eta_1 = \eta_2$.

By the previous analysis, we may assume:

$$\xi_1 - \eta_1 + \eta'_1 \neq \xi_1 - \eta_2 + \eta'_2, \ \eta_1 \neq \eta'_1, \ \eta_2 \neq \eta'_2.$$

Since $\eta_1 = \eta_2$, we note that $\xi_1 - \eta_1 + \eta'_1 \neq \xi_1 - \eta_1 + \eta'_2$, from where we deduce that $\eta'_1 \neq \eta'_2$. Consequently, condition (*) implies that:

$$\eta'_1 \in \{\xi_1 - \eta_1 + \eta'_1, \xi_1 - \eta_2 + \eta'_2\} = \{\xi_1 - \eta_1 + \eta'_1, \xi_1 - \eta_1 + \eta'_2\}.$$

We consider the following cases:

Case IIIA $\xi_1 - \eta_1 + \eta'_1 = \eta'_1$. In this case, we obtain that $\xi_1 = \eta_1$. Hence:

$$\xi_1 = \eta_1 = \eta_2$$

and

$$\xi_1 - \eta_2 + \eta'_2 = \xi_1 - \xi_1 + \eta'_2 = \eta'_2$$

In the matrix notation, we need to estimate:

$$\xi_1, \eta_1', \eta_2'; \begin{bmatrix} \langle \eta_1' \rangle^0 & \langle \xi_1 \rangle^\alpha & \langle \eta_1' \rangle^0 \\ \langle \eta_2' \rangle^0 & \langle \xi_1 \rangle^\alpha & \langle \eta_2' \rangle^0 \end{bmatrix}.$$

Similar to Case IIC, we cannot estimate this expression directly by using the Cauchy–Schwarz inequality, but we need to write it as:

$$\xi_{1}, \eta_{1}', \eta_{2}'; \langle \eta_{1}' \rangle^{-2\alpha} \cdot \langle \eta_{2}' \rangle^{-2\alpha} \cdot \begin{bmatrix} \langle \eta_{1}' \rangle^{\alpha} & \langle \xi_{1} \rangle^{\alpha} & \langle \eta_{1}' \rangle^{\alpha} \\ \langle \eta_{2}' \rangle^{\alpha} & \langle \xi_{1} \rangle^{\alpha} & \langle \eta_{2}' \rangle^{\alpha} \end{bmatrix}$$
(36)

which we can estimate by using the Cauchy–Schwarz inequality in ξ_1 , η'_1 , η'_2 since $\alpha > \frac{3}{4}$. We note that in (36) we are using the analogous notational convention as was defined in (35).

Case IIIB $\xi_1 - \eta_2 + \eta'_2 = \eta'_1$. In this case, we obtain:

$$\eta_2' = \eta_2 + \eta_1' - \xi_1 = \eta_1 + \eta_1' - \xi_1.$$

We use the fractional Leibniz rule as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \eta_1' \rangle^{\alpha}$$

in the first row and as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_2 + \eta'_2 \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \eta'_2 \rangle^{\alpha} = \langle \eta'_1 \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \eta_1 + \eta'_1 - \xi_1 \rangle^{\alpha}$$

in the second row. Consequently, we need to estimate:

$$\xi_1, \eta_1, \eta_1'; \begin{bmatrix} \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} & \langle \eta_1' \rangle^{\alpha} \\ \langle \eta_1' \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} & \langle \eta_1 + \eta_1' - \xi_1 \rangle^{\alpha} \end{bmatrix}.$$

We estimate this expression by first applying the Cauchy–Schwarz inequality in ξ_1 and then applying the Cauchy–Schwarz inequality in η_1 , η'_1 to obtain the desired bound.

Big Case IV $\eta'_1 = \eta'_2$.

From the previous analysis, we may assume that:

$$\xi_1 - \eta_1 + \eta'_1 \neq \xi_1 - \eta_2 + \eta'_2, \ \eta_1 \neq \eta_2, \ \eta_1 \neq \eta'_1, \ \eta_2 \neq \eta'_2.$$

Hence, from condition (*), we obtain:

$$\{\eta_1, \eta_2\} = \{\xi_1 - \eta_1 + \eta_1', \xi_1 - \eta_2 + \eta_2'\}.$$

We need to consider two cases:

Case IVA $\xi_1 - \eta_1 + \eta'_1 = \eta_1$. In this case, we obtain that:

$$\eta_1 = \frac{\xi_1 + \eta_1'}{2}.$$

We also know that $\eta_2 = \xi_1 - \eta_2 + \eta'_2$ and $\eta'_1 = \eta'_2$ and hence:

$$\eta_2 = \frac{\xi_1 + \eta_2'}{2} = \frac{\xi_1 + \eta_1'}{2}.$$

It follows that:

$$\eta_1 = \eta_2 = \frac{\xi_1 + \eta_1'}{2}.$$

This gives us a contradiction since we assumed that $\eta_1 \neq \eta_2$. *Case IVB* $\xi_1 - \eta_2 + \eta'_2 = \eta_1$. Since $\eta'_1 = \eta'_2$, it follows that $\xi_1 - \eta_2 + \eta'_1 = \xi_1 - \eta_2 + \eta'_2 = \eta_1$. Hence:

$$\eta_2 = \xi_1 - \eta_1 + \eta_1'.$$

We use the fractional Leibniz rule as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \eta_1' \rangle^{\alpha}$$

in the first row and as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_2 + \eta'_2 \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \eta'_2 \rangle^{\alpha} = \langle \eta_1 \rangle^{\alpha} \cdot \langle \xi_1 - \eta_1 + \eta'_1 \rangle^{\alpha} \cdot \langle \eta'_1 \rangle^{\alpha}$$

in the second row. Consequently, we need to estimate:

$$\xi_1, \eta_1, \eta_1'; \begin{bmatrix} \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} & \langle \eta_1' \rangle^{\alpha} \\ \langle \eta_1 \rangle^{\alpha} & \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} & \langle \eta_1' \rangle^{\alpha} \end{bmatrix}.$$

We estimate this expression by first applying the Cauchy–Schwarz inequality in ξ_1 and then in η_1 , η'_1 and we obtain the desired bound.

Big Case V $\eta_1 = \eta'_2$.

By the previous analysis, we may assume that:

$$\xi_1 - \eta_1 + \eta_1' \neq \xi_1 - \eta_2 + \eta_2', \ \eta_1 \neq \eta_2, \ \eta_1 \neq \eta_1', \ \eta_2 \neq \eta_2', \ \eta_1' \neq \eta_2'.$$
(37)

We need to consider several cases:

Case VA $\eta_2 = \eta'_1$.

In this case, we can use the fractional Leibniz rule as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_1 + \eta_1' \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \eta_1' \rangle^{\alpha} = \langle \xi_1 - \eta_1 + \eta_2 \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha}$$

in the first row and as:

$$\langle \xi_1 \rangle^{\alpha} \lesssim \langle \xi_1 - \eta_2 + \eta'_2 \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \eta'_2 \rangle^{\alpha} = \langle \xi_1 - \eta_2 + \eta_1 \rangle^{\alpha} \cdot \langle \eta_2 \rangle^{\alpha} \cdot \langle \eta_1 \rangle^{\alpha}$$

in the second row. Consequently, we need to estimate:

$$\xi_1, \eta_1, \eta_2; \begin{bmatrix} \langle \xi_1 - \eta_1 + \eta_2 \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} & \langle \eta_2 \rangle^{\alpha} \\ \langle \xi_1 - \eta_2 + \eta_1 \rangle^{\alpha} & \langle \eta_2 \rangle^{\alpha} & \langle \eta_1 \rangle^{\alpha} \end{bmatrix}.$$

We can estimate this expression by first applying the Cauchy–Schwarz inequality in ξ_1 and then in η_1 , η_2 to obtain the desired bound.

The next case we need to consider is:

Case VB $\eta_2 \neq \eta'_1$.

From condition (*), the fact that $\eta'_2 = \eta_1$, and from (37), it follows that in this case, one has:

$$\{\eta'_1, \eta_2\} = \{\xi_1 - \eta_1 + \eta'_1, \xi_1 - \eta_2 + \eta'_2\} = \{\xi_1 - \eta_1 + \eta'_1, \xi_1 - \eta_2 + \eta_1\}.$$

We need to consider two subcases:

VB1 $\eta'_1 = \xi_1 - \eta_1 + \eta'_1$, $\eta_2 = \xi_1 - \eta_2 + \eta_1$. From the first equality, it follows that $\xi_1 = \eta_1$. From the second equality, it follows that $\eta_2 = \frac{\xi_1 + \eta_1}{2}$. Consequently:

$$\eta_2 = \xi_1 = \eta_1.$$

This gives us a contradiction since we are assuming that $\eta_1 \neq \eta_2$.

VB2 $\eta'_1 = \xi_1 - \eta_2 + \eta_1$, $\eta_2 = \xi_1 - \eta_1 + \eta'_1$. We combine the two equalities to deduce that:

$$\eta_1' = \xi_1 - (\xi_1 - \eta_1 + \eta_1') + \eta_1 = 2\eta_1 - \eta_1'$$

from where it follows that:

$$\eta'_1 = \eta_1.$$

This gives us a contradiction since we are assuming that $\eta'_1 \neq \eta_1$.

The big case $\eta'_1 = \eta_2$ is analogous. Big Case VI $\xi_1 - \eta_1 + \eta'_1 \neq \xi_1 - \eta_2 + \eta'_2$, $\eta_1 \neq \eta'_1$, $\eta_2 \neq \eta'_2$, $\eta_1 \neq \eta_2$, $\eta'_1 \neq \eta'_2$, $\eta_1 \neq \eta'_2$, $\eta'_1 \neq \eta_2$. This big case cannot satisfy the condition (*). Namely, $\xi_1 - \eta_1 + \eta'_1$ and $\xi_1 - \eta_2 + \eta'_2$ can pair up with at most two distinct elements of $\{\eta_1, \eta'_1, \eta_2, \eta'_2\}$. The claim now follows.

Let us now note:

Remark 3.7. If we consider the domain $\Lambda = \mathbb{T}^d$, the proof of Theorem 3.1 carries over and the condition on α becomes $\alpha > \frac{d}{4}$, since for such α , it is the case that $\frac{1}{\langle n \rangle^{2\alpha}} \in \ell^2(\mathbb{Z}^d)$. We will omit the details.

Remark 3.8. The above proof can be modified to apply in the case when the $(h_{\xi})_{\xi \in \mathbb{Z}^3}$ are independent, identically distributed Gaussian random variables centered at zero. Since then it is no longer the case that $h^2 = 1$, it is better to define the randomized collision operator in a different way. The difference would be not to include the factor of $h_{\xi_1}(\omega)$ in (23) and (24). The proof of the estimate is analogous now and we use the fact the quantity $\int_{\Omega} |h_{\xi}(\omega)|^6 dp(\omega)$ is uniformly bounded in ξ . The latter condition is used when we apply Hölder's inequality in ω . In the Gaussian case, it is necessary to apply Hölder's inequality in ω since it is no longer the estimate to bounding a sum in frequencies satisfying a constraint analogous to (*), as was the case for Bernoulli random variables. We will omit the details.

We have chosen to use the Bernoulli randomization in order relate the randomization to the nonlinear Schrödinger equation. The link with the nonlinear Schrödinger equation is explained in more detail in Section 4.1. In this context, we see that it is essential to take Bernoulli random variables in the randomization. Moreover, the explicit computation of the Duhamel iterates in Section 6.2 is based on the fact that we are taking a Bernoulli randomization.

3.1. An Alternative Form of Randomization

It is possible to randomize in the initial data $\gamma_0^{(k+1)}$ instead of in the collision operator $B_{j,k+1}$ as we did in (23). The former approach was shown to be useful in the study of nonlinear dispersive equations in the references noted in the introduction. In the context of the GP hierarchy, this approach still applies, but it leads to additional difficulties and it does not seem to be well suited to the problem.

More precisely, given $\gamma_0^{(k+1)}$ as in (27), one defines a *randomization map*:

$$\gamma_0^{(k+1)} \mapsto \gamma_{0,\omega}^{(k+1)}.$$
 (38)

This map takes a deterministic object and gives us a random object. $\gamma_0^{(k+1)}$ is defined by:

$$\gamma_{0,\omega}^{(k+1)} = \sum_{\ell_r, j_r} h_{\ell_1, \dots, \ell_{k+1}; j_1, \dots, j_{k+1}}^{(k+1)}(\omega) \cdot a_{\ell_1, \dots, \ell_{k+1}; j_1, \dots, j_{k+1}}^{(k+1)} \cdot b_{\ell_1, \dots, \ell_{k+1}; j_1, \dots, j_{k+1}}^{(k+1)}$$

for $h_{\ell_1,\ldots,\ell_{k+1};j_1,\ldots,j_{k+1}}^{(k+1)}(\omega) := \prod_{r=1}^{k+1} h_{\ell_r}(\omega) \cdot \prod_{r=1}^{k+1} \overline{h_{j_r}(\omega)}$. Here, we assume that (h_j) is a sequence of *independent*, *identically distributed standard real Bernoulli* random variables centered at zero. Since the random variables are all real, we can write $h_{\ell_1,\ldots,\ell_{k+1};j_1,\ldots,j_{k+1}}^{(k+1)}(\omega) := \prod_{r=1}^{k+1} h_{\ell_r}(\omega) \cdot \prod_{r=1}^{k+1} h_{j_r}(\omega)$. Having defined the randomization this way, it is possible to prove:

Theorem 3.9. For the randomization defined in (38), the following bound holds:

$$\left\|S^{(k,\alpha)}B_{j,k+1}\gamma_{0,\omega}^{(k+1)}\right\|_{L^{2}\left(\Omega\times\Lambda^{k}\times\Lambda^{k}\right)} \leq C(\alpha,k)\left\|S^{(k+1,\alpha)}\gamma_{0}^{(k+1)}\right\|_{L^{2}\left(\Lambda^{k+1}\times\Lambda^{k+1}\right)},$$

whenever $\alpha > \frac{3}{4}$. The constant $C = C(\alpha, k) > 0$ depends on both α and k.

The proof of Theorem 3.9 is similar to the proof of Theorem 3.1, and is based on a pairing of frequencies analogous to the condition (*) in the proof of Theorem 3.1. Due to the fact that one is randomizing in all of the frequencies in $\gamma_0^{(k+1)}$, and not just in those occurring in the collision, there are many more pairings to consider. As a result, the analysis is more involved, although it follows the same principle as the proof of Theorem 3.1. There are approximately three times as many cases to check. In the end, the condition that one needs is again $\alpha > \frac{3}{4}$. In the general case of $\Lambda = \mathbb{T}^d$, one needs to again assume that $\alpha > \frac{d}{4}$. We will omit the details. Following the mentioned outline, it is possible to show that the *k*-dependence

in the constant *C* is of the order $\sim_{\alpha} \sqrt{\frac{(2k)!}{2^k k!}}$, which is $\sim \sqrt{k!}$ by Stirling's formula. The reason for this large constant is that the various frequencies can pair up in many different ways, which was not the case in the proof of Theorem 3.1. More precisely, when we take squares and integrate in ω , there are 2k frequencies which pair up in $\sim \frac{(2k)!}{2^k k!}$ different ways. A similar phenomenon is encountered in Section 6.4, where we keep precise track of the implied constant in the estimate.

If, instead of Bernoulli random variables, we take standard Gaussian random variables centered at zero, then a version of Theorem 3.9 still holds. In this case, the constant C has additional dependence on k, in addition to the $\sqrt{k!}$ factor which was noted earlier. This is due to the fact that we need to bound the quantity $\int_{\Omega} |h_{\xi}(\omega)|^{2(k+1)} dp(\omega)$, which is not uniformly bounded in k for (h_{ξ}) a standard Gaussian random variable centered at zero.

For a further discussion of this method of randomization, we refer the reader to Remark 5.1. In particular, we note that the non-commutativity relation given by (51) in the mentioned remark implies that the boardgame argument is not applicable in the model which one obtains by using this method of randomization. The precise details are given in the remark.

4. Properties of the Randomized Gross-Pitaevskii Hierarchy

In this section, we study the properties of the *randomized Gross–Pitaevskii hierarchy*:

$$\begin{cases} i \partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega} (\gamma^{(k+1)}) \\ \gamma^{(k)} \big|_{t=0} = \gamma_0^{(k)}. \end{cases}$$
(39)

We see that, in this hierarchy, there is only one random parameter $\omega \in \Omega$. In other words, we see that all of the collision operators are randomized according to the same parameter, according to which all of the collisions are randomized. As a result, we also sometimes call (39) the *dependently randomized Gross–Pitaevskii hierarchy* in the discussion that follows. We note below that this hierarchy has a lot of properties which carry over from the deterministic GP hierarchy.

4.1. Link with the Nonlinear Schrödinger Equation

In this subsection, we construct *factorized solutions* to (39). As we will see, they will be obtained as tensor products of solutions to a *nonlinear Schrödinger equation* with a *random nonlinearity*. In this way, we obtain a probabilistic analogue of the discussion in Section 2.3. As a result, we can better understand the connection between the deterministic and the randomized GP hierarchy.

In order to avoid notational difficulties of dependence on the randomization parameter ω , we will give a more precise notation for the randomization map given in (1.5). Namely, given $\omega \in \Omega$, we define the map $T^{\omega} : L^2(\Lambda) \to L^2(\Lambda)$ as follows:

$$(T^{\omega}f)\widehat{}(\xi) := h_{\xi}(\omega) \cdot \widehat{f}(\xi), \ \xi \in \mathbb{Z}^3.$$

$$(40)$$

We note that T^{ω} is a linear operator. It corresponds to the map defined in (1.5). Since we are using standard Bernoulli random variables, it follows that $h_{\xi}^2 = 1$. Hence, we obtain that T^{ω} is an involution:

$$T^{\omega} \circ T^{\omega} = Id. \tag{41}$$

Suppose that $\phi = \phi(x, t)$ solves the Cauchy problem associated to the nonlinear Schrödinger equation:

$$\begin{cases} i\partial_t \phi + \Delta \phi = |\phi|^2 \phi \\ \phi|_{t=0} = \phi_0. \end{cases}$$
(42)

Applying the operator T^{ω} to the above equation, it follows that:

$$i\partial_t \left(T^{\omega}\phi\right) + \Delta\left(T^{\omega}\phi\right) = T^{\omega}\left(|\phi|^2\phi\right).$$

By (41), it follows that:

$$i\partial_t \left(T^{\omega}\phi\right) + \Delta\left(T^{\omega}\phi\right) = T^{\omega}\left(\left|T^{\omega}(T^{\omega}\phi)\right|^2 \cdot T^{\omega}(T^{\omega}\phi)\right).$$

Consequently, the function $\psi = \psi(\omega) := T^{\omega}\phi$ solves the equation:

$$\begin{cases} i\partial_t \psi + \Delta \psi = T^{\omega} \Big(|T^{\omega} \psi|^2 \cdot T^{\omega} \psi \Big) \\ \psi|_{t=0} = T^{\omega} \phi_0. \end{cases}$$
(43)

In other words, the function ψ , which is a randomization of the solution of the deterministic NLS (42), solves a randomized NLS-type equation. We can now use the function ψ^{ω} to find a *factorized solution* to the randomized Gross–Pitaevskii hierarchy (39). In particular, if we take initial data $\gamma_0^{(k)} := |T^{\omega}\phi_0\rangle \langle T^{\omega}\phi_0|^{\otimes k}$, we note that a solution to (39) is given by $\gamma^{(k)} := |T^{\omega}\phi_0\rangle \langle T^{\omega}\phi|^{\otimes k}$. This follows from the definition of the operator T^{ω} and from the definition of the Fourier transform of $[B_{j,k+1}]^{\omega}\gamma_0^{(k+1)}$ given in (25).

Remark 4.1. It can be shown that the boardgame argument in general applies in the context of (39), as long as one works in a class of permutation symmetric density matrices. This gives another connection of the randomized GP hierarchy with the nonlinear Schrödinger equation. However, in our further analysis, we will study density matrices in the class of \mathcal{N} given in Definition 6.1 below, in which the objects no longer have permutation symmetry. This symmetry is crucial in applications the boardgame argument, so the combinatorial reduction does not apply in the class in which we will be working. It is important to note that this is not a structural feature of the hierarchy itself. We will henceforth not pursue the issue of applying the boardgame argument in the context of (39).

5. A New Randomized Hierarchy

As was noted in the previous section, the randomized Gross–Pitaevskii hierarchy (39) shares a lot of properties with the deterministic Gross–Pitaevskii hierarchy. However, due to the dependent randomization, it is not possible to directly apply the randomized spacetime estimate from Theorem 3.1. In this section, we circumvent this difficulty by modifying the problem. In particular, we consider the following *independently randomized Gross–Pitaevskii hierarchy*:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega_{k+1}}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_0^{(k)}. \end{cases}$$
(44)

The difference between the randomized GP hierarchy (39) and the hierarchy (44) is the fact that, in the latter case, the randomization parameters $\omega_k \in \Omega$ are not assumed to be equal. We can view this as a way of independently randomizing the full collision operator $B^{(k+1)} = \sum_{j=1}^{k} B_{j,k+1}$ according to (23) and (24).

More precisely, for each k, we perform a sequence of independent Bernoulli trials. In other words, for every frequency $\xi \in \mathbb{Z}^3$, we choose $h_{\xi}(\omega_{k+1})$, which is 1 or -1, each with probability $\frac{1}{2}$. We then define $[B_{j,k+1}]^{\omega_{k+1}}$ according to (23) and (24), where each $h_{\xi}(\omega)$ is replaced by $h_{\xi}(\omega_{k+1})$. We apply the above procedure at each level of the hierarchy. This is in contrast to the (dependently) randomized GP hierarchy (39), where we perform a sequence of independent Bernoulli trials only once to determine $h_{\xi}(\omega)$ for $\xi \in \mathbb{Z}^3$, and we use the obtained random coefficients at each level of the hierarchy.

In the context of (44), it is possible to apply the averaged spacetime estimate from Theorem 3.1 to estimate the Duhamel expansion of the solution evolving from zero initial data. We have to estimate these terms in a norm which involves an average in all of the randomization parameters which occur in the expansion. The precise estimate is given in Theorem 5.2.

5.1. Properties of the Hierarchy (44)

Before we state and give an application of Theorem 3.1 in Section 5.2, let us briefly summarize some of the relevant properties of the hierarchy (44). Unlike the (dependently) randomized GP hierarchy, the independently randomized GP hierarchy has some properties which are different from the deterministic case. We observe that the following holds:

5.1.1. No Obvious Factorized Solutions The hierarchy (44) no longer admits obvious factorized solutions. We cannot in general find solutions $\gamma^{(k)} := |T^{\omega_k}\phi\rangle \langle T^{\omega_k}\phi|^{\otimes k}$, where $i\partial_t\phi + \Delta\phi = |\phi|^2\phi$. The relation:

$$\left(i\partial_t + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \right) | T^{\omega_k} \phi \rangle \langle T^{\omega_k} \phi |^{\otimes k}$$

=
$$\sum_{j=1}^k [B_{j,k+1}]^{\omega_{k+1}} \left(| T^{\omega_{k+1}} \phi \rangle \langle T^{\omega_{k+1}} \phi |^{\otimes (k+1)} \right)$$

only holds when $\omega_k = \omega_{k+1}$.

5.1.2. The Boardgame Argument Does Not Apply We will now show that the *boardgame argument* from [108] does not apply in the context of (44). In order to avoid repetition of the construction of the boargame argument, we refer the reader to [108, Section 3]. The main idea of the current sub-section is the fact that the fact that the ω_k are chosen to be distinct permits us to apply the regrouping of terms as in [108].

Given $k, \ell, m, n \in \mathbb{N}$ mutually distinct such that $1 \le k \le \ell - 1, 1 \le m \le n - 1$, and given $\omega_1, \omega_2 \in \Omega$, the following commutation relation can be shown for the randomized collision operators:

$$[B_{k,\ell}]^{\omega_1} [B_{m,n}]^{\omega_2} = [B_{m,n}]^{\omega_2} [B_{k,\ell}]^{\omega_1}.$$
(45)

Even though the commutation relation (45) holds for (44), the boardgame argument still does not apply. The main reason is the fact that in the derivation of the boardgame argument [108], one can interchange the t_k and x_k variables inside of the integrand. However, in the context of (44), it is not possible to interchange the ω_k parameters. Thus, integrands which were equal in the deterministic case (or in the case that all of the ω_k were equal), now become mutually distinct.

Let us examine this problem on a concrete example. We use the same example given in the explanation of the boardgame argument in [108, equations (26) and (27)]. In the following discussion, we write the operator $\mathcal{U}^{(k)}(t)$ explicitly as $e^{it\Delta_{\pm}^{(k)}}$, with notation as in [108]. Let us start with the time-dependent density matrix $\gamma^{(5)} = \gamma^{(5)}(t, \vec{x}_5; \vec{x}'_5)$ which satisfies the permutation symmetry:

$$\gamma^{(5)}\left(t, x_{\sigma(1)}, \dots, x_{\sigma(5)}; x'_{\sigma(1)}, \dots, x'_{\sigma(5)}\right) = \gamma^{(5)}\left(t, x_1, \dots, x_5; x'_1, \dots, x'_5\right)$$
(46)

for all permutations $\sigma \in S^5$. In the discussion that follows, we write $\gamma^{(5)}(t, \vec{x}_5; \vec{x}'_5)$ just as $\gamma^{(5)}(t)$. We write Δ_{\pm,x_j} for the Laplace operator only in the x_j variable. We also use the following shorthand $\Delta_{\pm}^{(k)} := \Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}$.

With this notation in mind, we consider:

$$I_{1} := \int_{t_{1} \ge t_{2} \ge t_{5} \ge t_{4} \ge t_{3}} e^{i(t_{1}-t_{2})\Delta_{\pm}^{(1)}} [B_{1,2}]^{\omega_{2}} e^{i(t_{2}-t_{3})\Delta_{\pm}^{(2)}} [B_{2,3}]^{\omega_{3}} e^{i(t_{3}-t_{4})\Delta_{\pm}^{(3)}} [B_{1,4}]^{\omega_{4}} e^{i(t_{4}-t_{5})\Delta_{\pm}^{(4)}} [B_{4,5}]^{\omega_{5}} \gamma^{(5)}(t_{5}) dt_{2} dt_{3} dt_{4} dt_{5}$$

$$(47)$$

and

$$I_{2} := \int_{t_{1} \ge t_{2} \ge t_{5} \ge t_{3} \ge t_{4}} e^{i(t_{1}-t_{2})\Delta_{\pm}^{(1)}} [B_{1,2}]^{\omega_{2}} e^{i(t_{2}-t_{3})\Delta_{\pm}^{(2)}} [B_{1,3}]^{\omega_{3}} e^{i(t_{3}-t_{4})\Delta_{\pm}^{(3)}} [B_{3,4}]^{\omega_{4}} e^{i(t_{4}-t_{5})\Delta_{\pm}^{(4)}} [B_{3,5}]^{\omega_{5}} \gamma^{(5)}(t_{5}) dt_{2} dt_{3} dt_{4} dt_{5}.$$
(48)

We note that I_1 and I_2 are functions of t_1, x_1, x'_1 . The collision operators in the integrand act by integrating out the dependence on $x_2, x'_2, x_3, x'_3, x_4, x'_4, x_5, x'_5$.

In the deterministic setting [108], it was shown that the analogues of the expressions I_1 and I_2 are mutually equal. As was noted above, the same is true in the setting of the (dependently) randomized GP hierarchy. We now show that, in the context of the independently randomized GP hierarchy, it is no longer the case that I_1 is equal to I_2 .

As in [108], an identity which follows from the commutation (45) is:

$$e^{i(t_2-t_3)\Delta_{\pm}^{(2)}} [B_{2,3}]^{\omega_3} e^{i(t_3-t_4)\Delta_{\pm}^{(3)}} [B_{1,4}]^{\omega_4} e^{i(t_4-t_5)\Delta_{\pm}^{(4)}} = e^{i(t_2-t_4)\Delta_{\pm}^{(2)}} [B_{1,4}]^{\omega_4} e^{-i(t_3-t_4)(\Delta_{\pm}^{(3)}-\Delta_{\pm,x_3}+\Delta_{\pm,x_4})} [B_{2,3}]^{\omega_3} e^{i(t_3-t_4)\Delta_{\pm}^{(4)}}.$$
(49)

In the following discussion, let $k_{1,2}^{\omega_2}$ formally denote the kernel of $[B_{1,2}]^{\omega_2}$. We use similar notation for the other collision operators. Let:

$$\gamma_{3,4} := \gamma^{(5)} \left(t_5, x_1, x_2, x_3, x_4, x_5; x_1', x_2', x_3', x_4', x_5' \right)$$

and

$$\gamma_{4,3} := \gamma^{(5)} \left(t_5, x_1, x_2, x_4, x_3, x_5; x_1', x_2', x_4', x_3', x_5' \right).$$

The symmetry assumption on $\gamma^{(5)}$ given by (46) guarantees that $\gamma_{3,4} = \gamma_{4,3}$.

If we use (49) and if we write out I_1 in terms of the integral kernels, it follows that:

$$I_{1} = \int_{t_{1} \ge t_{2} \ge t_{5} \ge t_{4} \ge t_{3}} \int_{\Lambda^{4} \times \Lambda^{4}} e^{i(t_{1}-t_{2})\Delta_{\pm}^{(1)}} k_{1,2}^{\omega_{2}} e^{i(t_{2}-t_{4})\Delta_{\pm}^{(2)}} k_{1,4}^{\omega_{4}}$$
$$\times e^{-i(t_{3}-t_{4})\left(\Delta_{\pm}^{(3)} - \Delta_{\pm,x_{3}} + \Delta_{\pm,x_{4}}\right)} k_{2,3}^{\omega_{3}} e^{i(t_{3}-t_{5})\Delta_{\pm}^{(4)}} k_{4,5}^{\omega_{5}} \gamma_{4,3}$$
$$\times dx_{2} \dots dx_{5} dx_{2}' \dots dx_{5}' dt_{2} dt_{3} dt_{4} dt_{5}.$$

We exchange (t_3, x_3, x'_3) and (t_4, x_4, x'_4) to deduce that this expression equals:

$$\int_{t_1 \ge t_2 \ge t_5 \ge t_3 \ge t_4} \int_{\Lambda^4 \times \Lambda^4} e^{i(t_1 - t_2)\Delta_{\pm}^{(1)}} k_{1,2}^{\omega_2} e^{i(t_2 - t_3)\Delta_{\pm}^{(2)}} k_{1,3}^{\omega_4} e^{i(t_3 - t_4)\Delta_{\pm}^{(3)}} k_{2,4}^{\omega_5} e^{i(t_4 - t_5)\Delta_{\pm}^{(4)}} k_{3,5}^{\omega_5} \gamma_{3,4} dx_2 \dots dx_5 dx_2' \dots dx_5' dt_2 dt_3 dt_4 dt_5$$

which since $\gamma_{3,4} = \gamma_{4,3}$ equals:

$$\int_{t_1 \ge t_2 \ge t_5 \ge t_3 \ge t_4} \int_{\Lambda^4 \times \Lambda^4} e^{i(t_1 - t_2)\Delta_{\pm}^{(1)}} k_{1,2}^{\omega_2} e^{i(t_2 - t_3)\Delta_{\pm}^{(2)}} k_{1,3}^{\omega_4} e^{i(t_3 - t_4)\Delta_{\pm}^{(3)}} k_{2,4}^{\omega_3} e^{i(t_4 - t_5)\Delta_{\pm}^{(4)}} k_{3,5}^{\omega_5} \gamma_{4,3} dx_2 \dots dx_5 dx_2' \dots dx_5' dt_2 dt_3 dt_4 dt_5.$$

By definition, this is:

$$\int_{t_1 \ge t_2 \ge t_5 \ge t_3 \ge t_4} e^{i(t_1 - t_2)\Delta_{\pm}^{(1)}} [B_{1,2}]^{\omega_2} e^{i(t_2 - t_3)\Delta_{\pm}^{(2)}} [B_{1,3}]^{\omega_4} e^{i(t_3 - t_4)\Delta_{\pm}^{(3)}} [B_{2,3}]^{\omega_3} e^{i(t_4 - t_5)\Delta_{\pm}^{(4)}} [B_{3,5}]^{\omega_5} \gamma^{(5)}(t_5) dt_2 dt_3 dt_4 dt_5$$

which does not equal

$$I_{2} = \int_{t_{1} \ge t_{2} \ge t_{5} \ge t_{3} \ge t_{4}} e^{i(t_{1}-t_{2})\Delta_{\pm}^{(1)}} [B_{1,2}]^{\omega_{2}} e^{i(t_{2}-t_{3})\Delta_{\pm}^{(2)}} [B_{1,3}]^{\omega_{3}} e^{i(t_{3}-t_{4})\Delta_{\pm}^{(3)}}$$
$$[B_{3,4}]^{\omega_{4}} e^{i(t_{4}-t_{5})\Delta_{\pm}^{(4)}} [B_{3,5}]^{\omega_{5}} \gamma^{(5)}(t_{5}) dt_{2} dt_{3} dt_{4} dt_{5}$$

unless $\omega_3 = \omega_4$. The fact that the deterministic analogue of I_1 is equal to the deterministic analogue of I_2 is an instance of the procedure used in the regrouping of terms in the boardgame argument [108]. As we see, this component of the procedure no longer applies if we add the ω_k dependence as above.

Remark 5.1. If we were to use the randomization mentioned in Section 3.1, we would be led to the study of the following hierarchy:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{x_k} - \Delta_{x'_k}) \gamma^{(k)} = \sum_{j=1}^k B^{\omega}_{j,k+1}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma^{(0)}_k \end{cases}$$
(50)

for $\omega \in \Omega$. Let us explain the notation. The operator $B_{j,k+1}^{\omega}$ is defined as first applying the operator $B_{j,k+1}$ and then applying the randomization given by (38) in Remark 3.7 to the result. In other words, for a fixed density matrix $\sigma_0^{(k+1)}$:

$$\sigma_0^{(k+1)} \mapsto B_{j,k+1}(\sigma_0^{(k+1)}) \mapsto B_{j,k+1}^{\omega}(\sigma_0^{(k+1)}) = \text{randomization of } B_{j,k+1}(\sigma_0^{(k+1)}).$$

It can be shown that, for $k, \ell, m, n \in \mathbb{N}$ mutually distinct such that $1 \le k \le \ell - 1$, $1 \le m \le n - 1$, the analogue of the commutation relation (45) does not hold, that is

$$B_{k,\ell}^{\omega} B_{m,n}^{\omega} \neq B_{m,n}^{\omega} B_{k,\ell}^{\omega}.$$
(51)

Hence, the boardgame argument does not apply in the study of (50).

5.2. An Application of the Randomized Spatial Estimate

In this subsection, we apply the estimate given by Theorem 3.1 to the *inde*pendently randomized Gross–Pitaevskii hierarchy (44). The main result is given in Theorem 5.2 below. Throughout this subsection we assume that $\alpha > \frac{3}{4}$.

Let us fix a *deterministic* sequence of time-dependent density matrices $(\gamma^{(k)}(t))_k = (\gamma^{(k)}(t, \vec{x}_k; \vec{x}'_k))_k$ satisfying the following a priori bound:

$$\left\|S^{(k,\alpha)}\gamma^{(k)}(t)\right\|_{L^2(\Lambda^k \times \Lambda^k)} \le C_1^k \tag{52}$$

uniformly in time for some $C_1 > 0$ independent of k. We note that, for factorized objects $\gamma^{(k)} = |\psi\rangle\langle\psi|^{\otimes k}$, the condition (52) reduces to

$$\|\psi(t)\|_{H^{\alpha}} \le \sqrt{C_1}.\tag{53}$$

In the case $\alpha = 1$, the condition (53) holds globally in time if $\psi = \phi$ for ϕ solving the NLS equation:

$$\begin{cases} i \partial_t \phi + \Delta \phi = |\phi|^2 \phi \\ \phi|_{t=0} = \phi_0 \in H^{\alpha}. \end{cases}$$

Since T^{ω} is an isometry on H^{α} , it follows that the condition (53) also holds for $\psi = T^{\omega}\phi$, whenever $\omega \in \Omega$. If $\alpha > 1$, then we obtain that the condition (53) holds only locally in time for $\psi = \phi$ and $\psi = \phi^{\omega}$. The latter fact can be deduced from the arguments given in [21]. As we will see, this also will be enough for our application.

Since we will not be applying the boardgame argument, we do not need to assume that $\gamma^{(k)}$ satisfies the permutation symmetry property as in (46). In this section, we write $\omega_k \in \Omega_k$ instead of $\omega \in \Omega$ in order to emphasize that the indices are distinct. In other words, the different randomizations will be independent. Given $k, n \in \mathbb{N}$, $t_k > 0$ and $(\omega_{k+1}, \omega_{k+2}, \dots, \omega_{n+k}) \in \Omega_{k+1} \times \Omega_{k+2} \times \dots \times \Omega_{n+k}$,

Given $k, n \in \mathbb{N}, t_k > 0$ and $(\omega_{k+1}, \omega_{k+2}, \dots, \omega_{n+k}) \in \Omega_{k+1} \times \Omega_{k+2} \times \dots \times \Omega_{n+k}$, we define $\sigma_{n; \omega_{k+1}, \dots, \omega_{n+k}}^{(k)}$ by:

$$\sigma_{n;\ \omega_{k+1},\ldots,\omega_{n+k}}^{(k)}(t_k) = (-i)^n \int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} \mathcal{U}^{(k)}(t_k - t_{k+1}) \left[B^{(k+1)}\right]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \left[B^{(k+2)}\right]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \left[B^{(n+k)}\right]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) dt_{n+k} \cdots dt_{k+2} dt_{k+1}.$$
(54)

Here, we recall that by (25):

$$\left[B^{(\ell+1)}\right]^{\omega} = \sum_{j=1}^{\ell} \left[B_{j,\ell+1}^{+}\right]^{\omega} - \sum_{j=1}^{\ell} \left[B_{j,\ell+1}^{-}\right]^{\omega} = \sum_{j=1}^{\ell} \left[B_{j,\ell+1}^{-}\right]^{\omega}.$$

This is the density matrix of order k which we obtain after n Duhamel iterations in the hierarchy (44). In this notation, the superscript k denotes the order of the density matrix and the subscript n denotes the length of the Duhamel expansion. The ω_j are the (fixed) randomization parameters in the probability space Ω . More precisely, let us fix $n \in \mathbb{N}$ and $(\omega_2, \omega_3, \dots, \omega_{n+1}) \in \Omega_2 \times \Omega_3 \times \dots \times \Omega_{n+1}$. Let us look at:

$$\begin{split} \tilde{\gamma}^{(1)} &:= \sigma_{n; \, \omega_2, \omega_3, \omega_4, \omega_5, \dots, \, \omega_{n+1}}^{(1)} \\ \tilde{\gamma}^{(2)} &:= \sigma_{n-1; \, \omega_3, \omega_4, \omega_5, \dots, \, \omega_{n+1}}^{(2)} \\ \tilde{\gamma}^{(3)} &:= \sigma_{n-2; \, \omega_4, \omega_5, \dots, \, \omega_{n+1}}^{(3)} \\ &\vdots \\ \tilde{\gamma}^{(n)} &:= \sigma_{1; \, \omega_{n+1}}^{(n)}. \end{split}$$

Then, by construction, we obtain that:

$$\begin{cases} i \partial_t \tilde{\gamma}^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \tilde{\gamma}^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega_{k+1}} (\tilde{\gamma}^{(k+1)}) \\ \tilde{\gamma}^{(k)} \big|_{t=0} = 0. \end{cases}$$

for all $k \in \{1, 2, ..., n - 1\}$. In other words, we obtain an *arbitrarily long subset* of solutions to the full hierarchy (44) with zero initial data.

Theorem 5.2. Suppose that $\alpha > \frac{3}{4}$ and $k \in \mathbb{N}$. Consider $\sigma_{n; \omega_{k+1}, \omega_{k+2}, \dots, \omega_{n+k}}^{(k)}$, defined as in (54). There exists T > 0 depending only on the constant C_1 in (52) and on α such that:

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\dots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\Omega_{k+1}\times\Omega_{k+2}\times\dots\times\Omega_{n+k};\,L^2(\Lambda^k\times\Lambda^k)\right)} \to 0$$
(55)

as $n \to \infty$. Moreover,

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\prod_{m\geq 2}\Omega_m;L^2(\Lambda^k\times\Lambda^k)\right)} \to 0$$
(56)

as $n \to \infty$.

Proof. We first prove (55). Throughout the proof, let us take T > 0 small which will be precisely determined later and let us consider a fixed $t = t_k \in [0, T]$. We compute:

$$S^{(k,\alpha)}\sigma_{n;\,\omega_{k+1},\ldots,\omega_{n+k}}^{(k)}(t_{k})$$

$$= (-i)^{n} \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} S^{(k,\alpha)} \mathcal{U}^{(k)}(t_{k} - t_{k+1})$$

$$[B^{(k+1)}]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2})$$

$$[B^{(k+2)}]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k})$$

$$[B^{(n+k)}]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) dt_{n+k} \cdots dt_{k+2} dt_{k+1}.$$

Hence, we obtain by Minkowski's inequality:

$$\begin{split} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\dots,\omega_{n+k}}^{(k)} \right\|_{L^{2}\left(\Omega_{k+1}\times\Omega_{k+2}\times\dots\times\Omega_{n+k};\,L^{2}(\Lambda^{k}\times\Lambda^{k})\right)} \\ &\leq \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} \left\| S^{(k,\alpha)} \mathcal{U}^{(k)}(t_{k}-t_{k+1}) \left[B^{(k+1)} \right]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1}-t_{k+2}) \\ &\left[B^{(k+2)} \right]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1}-t_{n+k}) \\ &\left[B^{(n+k)} \right]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) \left\|_{L^{2}\left(\Omega_{k+1}\times\Omega_{k+2}\times\dots\times\Omega_{n+k};\,L^{2}(\Lambda^{k}\times\Lambda^{k})\right)} \, \mathrm{d}t_{n+k} \cdots \, \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1}. \end{split}$$
(57)

We look at the integrand in the *t*-variables, that is at:

$$\left\| S^{(k,\alpha)} \mathcal{U}^{(k)}(t_k - t_{k+1}) \left[B^{(k+1)} \right]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \right. \\ \left. \left[B^{(k+2)} \right]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \right. \\ \left. \left[B^{(n+k)} \right]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) \right\|_{L^2\left(\Omega_{k+1} \times \Omega_{k+2} \times \cdots \times \Omega_{n+k}; L^2(\Lambda^k \times \Lambda^k)\right)}$$

By using the unitarity of $\mathcal{U}^{(k)}(t_k - t_{k+1})$, and the fact that this operator commutes with $S^{(k,\alpha)}$, the above expression equals:

$$\|S^{(k,\alpha)}[B^{(k+1)}]^{\omega_{k+1}}\mathcal{U}^{(k+1)}(t_{k+1}-t_{k+2})[B^{(k+2)}]^{\omega_{k+2}}\cdots\mathcal{U}^{(n+k-1)} (t_{n+k-1}-t_{n+k})\Big[B^{(n+k)}\Big]^{\omega_{n+k}}\gamma^{(n+k)}(t_{n+k})\Big\|_{L^2\big(\Omega_{k+1}\times\Omega_{k+2}\times\cdots\times\Omega_{n+k};L^2(\Lambda^k\times\Lambda^k)\big)}.$$

We use the second bound from Theorem 3.1 to deduce that this expression is:

$$\leq C_0 k \left\| S^{(k+1,\alpha)} [B^{(k+2)}]^{\omega_{k+2}} \mathcal{U}^{(k+2)}(t_{k+2} - t_{k+3}) \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \right\|_{L^2(\Omega_{k+2} \times \Omega_{k+3} \times \cdots \times \Omega_{n+k}; L^2(\Lambda^{k+1} \times \Lambda^{k+1}))}.$$
(58)

Here, C_0 denotes the constant *C* from Corollary 3.2. We note that the last step is justified since we are taking an L^2 norm in all of the variables and hence we can apply the randomized spacetime estimate in a "vector-valued" setting. We can see this by squaring both sides, integrating and using Fubini's theorem. Let us note that,

in order to deduce the bound (58), we also used the unitarity of $\mathcal{U}^{(k+1)}(t_k - t_{k+1})$ and the fact that this operator commutes with $S^{(k+1,\alpha)}$. Iterating this procedure, we obtain:

$$\begin{split} \left\| S^{(k,\alpha)} \mathcal{U}^{(k)}(t_k - t_{k+1}) \left[B^{(k+1)} \right]^{\omega_{k+1}} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \\ \left[B^{(k+2)} \right]^{\omega_{k+2}} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \\ \left[B^{(n+k)} \right]^{\omega_{n+k}} \gamma^{(n+k)}(t_{n+k}) \right\|_{L^2\left(\Omega_{k+1} \times \Omega_{k+2} \times \cdots \times \Omega_{n+k}; L^2(\Lambda^k \times \Lambda^k)\right)} \\ &\leq \left(C_0 k \right) \cdot \left(C_0(k+1) \right) \cdots \left(C_0(n+k) \right) \\ \cdot \left\| S^{(n+k,\alpha)} \gamma^{(n+k)}(t_{n+k}) \right\|_{L^2(\Lambda^{n+k} \times \Lambda^{n+k})}. \end{split}$$

By using the a priori bound on $\gamma^{(n+k)}$ (given by the condition (52)), this is:

$$\leq \frac{(n+k)!}{(k-1)!} \cdot C_0^{n+1} \cdot C_1^{n+k}.$$
(59)

We can assume, without loss of generality that $C_0 \ge 1$. In particular, we can assume that $C_0^{n+1} \le C_0^{n+k}$. Hence, the expression in (59) is:

$$\leq \frac{(n+k)!}{(k-1)!} \cdot M^{n+k} \tag{60}$$

for some $M \sim C_0 \cdot C_1$, which is independent of *n* and *k*. The following identity is deduced by induction on *n*:

$$\int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} \mathrm{d}t_{n+k} \cdots \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1} = \frac{t_k^n}{n!}.$$
 (61)

The gain of n! in the denominator will be useful in the analysis which follows. Namely, we can combine (57), (60) and (61) in order to deduce that:

$$\sup_{t \in [0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\Omega_{k+1} \times \Omega_{k+2} \times \cdots \times \Omega_{n+k};\,L^2(\Lambda^k \times \Lambda^k)\right)}$$

$$\leq \frac{T^n}{n!} \cdot \frac{(n+k)!}{(k-1)!} \cdot M^{n+k} \leq \frac{T^n}{n!} \cdot (n+k)! \cdot M^{n+k}$$

$$= T^n \cdot (n+1) \cdot (n+2) \cdots (n+k) \cdot M^{n+k} \leq T^n \cdot (n+k)^k \cdot M^{n+k}$$

$$\lesssim_k T^n \cdot n^k \cdot M^n \cdot M^k \lesssim_k T^n \cdot (2M)^n \cdot M^k = (2MT)^n \cdot M^k.$$

Here, we used the fact that $n^k = O_k(2^n)$. We now choose T > 0 sufficiently small such that 2MT < 1.

It follows that, for T chosen in this way:

$$\sup_{t \in [0,T]} \|S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\dots,\omega_{n+k}}^{(k)}(t)\|_{L^2(\Omega_{k+1} \times \Omega_{k+2} \times \dots \times \Omega_{n+k};\,L^2(\Lambda^k \times \Lambda^k))} \to 0 \text{ as } n \to \infty.$$

By construction T depends only on M. M in turn depends only on C_1 and α . Hence, T depends only on C_1 and α . This proves the claim given in (55).

In order to prove (56), let us first state a non-trivial extension result from probability theory that allows us to make sense of the norm in which we are studying the convergence.

Theorem 5.3. [106, 111] Suppose that $(\Omega_i, \Sigma_i, p_i)_{i \in \mathcal{I}}$ is an arbitrary, non-empty family of probability spaces. Then, there exists a probability measure p on $\prod_{i \in \mathcal{I}} \Omega_i$ such that, for all finite subsets $F \subseteq I$:

$$p\left(\bigcap_{i\in F}\pi_i^{-1}(A_i)\right) = \prod_{i\in F}p_i(A_i)$$
(62)

whenever $A_i \in \Sigma_i$ for all $i \in F$. Here, for $j \in \mathcal{I}, \pi_j : \prod_{i \in \mathcal{I}} \Omega_i \to \Omega_j$ denotes the projection map onto Ω_j .

The obtained probability space is given by $(\prod_{i \in \mathcal{I}} \Omega_i, \mathcal{B}, p)$, where \mathcal{B} is the Borel field generated by sets of the form:

$$\bigcap_{i\in F} \pi_i^{-1}(A_i)$$

for some finite subset $F \subseteq \mathcal{I}$ and $A_i \in \Sigma_i$.

The first result of this type was proved in the work of KOLMOGOROV [111] in the case in which each probability space is a [0, 1] with Lebesgue measure. Kolmogorov's proof uses the fact that, in this context, the obtained product is topologically compact. The result in full generality was proved in the work of KAKUTANI [106]. This is the version stated in Theorem 5.3. There are also related results by DooB [71]. We use Theorem 5.3 in the case when \mathcal{I} is a countable family.

By (62), it follows that:

$$\begin{split} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t_{k}) \right\|_{L^{2}\left(\Omega_{k+1}\times\Omega_{k+2}\times\cdots\times\Omega_{n+k};\,L^{2}(\Lambda^{k}\times\Lambda^{k})\right)} \\ &= \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)} \right\|_{L^{2}\left(\Omega_{2}\times\Omega_{3}\times\cdots\times\Omega_{k}\times\Omega_{k+1}\times\cdots\times\Omega_{n+k}\times\Omega_{n+k+1}\times\Omega_{n+k+2}\times\cdots;\,L^{2}(\Lambda^{k}\times\Lambda^{k})\right)} \\ &= \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t_{k}) \right\|_{L^{2}\left(\prod_{m>2}\Omega_{m};L^{2}(\Lambda^{k}\times\Lambda^{k})\right)}. \end{split}$$

Here, we used the fact that there is no dependence on $\omega_2, \omega_3, \ldots, \omega_{k-1}$ and on $\omega_{n+2}, \omega_{n+3}, \ldots$ in $S^{(k,\alpha)}\sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t_k)$. In particular, (55) implies that, for $T = T(C_1, \alpha) > 0$,

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega_{k+1},\omega_{k+2},\ldots,\omega_{n+k}}^{(k)}(t) \right\|_{L^2\left(\prod_{m\geq 2}\Omega_m;L^2(\Lambda^k\times\Lambda^k)\right)} \to 0$$

as $n \to \infty$. The second claim of the theorem now follows.

Remark 5.4. The analysis is the same, as well as the bound, if one considers negative times. In the discussion that follows, we consider non-negative times, for simplicity of notation.

Remark 5.5. The above proof gives us that $T \sim_{\alpha} \frac{1}{C_1}$, where C_1 is the constant in (52).

Remark 5.6. The above discussion can be applied on \mathbb{T}^d , whenever $d \ge 1$, provided that $\alpha > \frac{d}{4}$.

6. The Randomized Gross-Pitaevskii Hierarchy Revisited

In this section, we study the randomized Gross–Pitaevskii hierarchy (39) in more detail. In the discussion that follows, we assume that $\alpha \ge 0$. We note that all of the results of this section apply to the spatial domain $\Lambda = \mathbb{T}^d$ for $d \ge 1$.

The main result of this section is Theorem 6.4, which is the analogue of Theorem 5.2 in the dependently randomized setting. In this context, the level of regularity is $\alpha \ge 0$. This should be compared to the assumption that $\alpha > \frac{3}{4}$, which we had to make when we were applying the spacetime estimate from Theorem 3.1 in the study of the independently randomized GP hierarchy (44). Furthermore, since there is only one random parameter involved, the random component is simpler than the infinite product that we had to use in the independently randomized setting, and hence the spaces in which we will work will be simpler than those used in Section 5.

Let us recall that the (depdendently) randomized Gross–Pitaevskii hierarchy is given by:

$$\begin{cases} i\partial_t \gamma^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k})\gamma^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega}(\gamma^{(k+1)}) \\ \gamma^{(k)}|_{t=0} = \gamma_0^{(k)}. \end{cases}$$

As before, we take homogeneous initial data $\gamma_0^{(k)} = 0$ and we study Duhamel terms of the type:

$$(-i)^{n} \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} \mathcal{U}^{(k)}(t_{k} - t_{k+1}) [B^{(k+1)}]^{\omega} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \left[B^{(k+2)}\right]^{\omega} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) [B^{(n+k)}]^{\omega} \gamma^{(n+k)}(t_{n+k}) dt_{n+k} \cdots dt_{k+2} dt_{k+1}.$$

In this case, since the randomizations are no longer mutually independent, we cannot apply the estimate from Theorem 3.1 except in the special case that n = 1. We explain this difficulty in more detail now.

6.1. Difficulties Arising from Higher-Order Duhamel Expansions

If we try to extend the ideas from the proof of Theorem 3.1 to higher Duhamel expansions, we see that the pairings of the frequencies are not enough in order to deduce the needed estimate. In the following section, we will explain this point on an example. Let us note that these arguments can be extended to the case of a Duhamel expansion of length $n \ge 3$. For simplicity, we will construct an example

when the length of the expansion equals 2. This example will help us motivate a special class of density matrices in Section 6.3.

We will show that it is, in general, not possible to use the pairings due to the randomization in order to deduce that, for $\alpha > \frac{3}{4}$:

$$\begin{split} \| S^{(1,\alpha)} \mathcal{U}^{(1)}(t_1 - t_2) \left[B^+_{1,2} \right]^{\omega} \mathcal{U}^{(2)}(t_2 - t_3) \left[B^+_{2,3} \right]^{\omega} \gamma^{(3)} \|_{L^2(\Omega \times \Lambda \times \Lambda)} \\ \lesssim \| S^{(1,\alpha)} \gamma^{(3)} \|_{L^2(\Lambda^3 \times \Lambda^3)}. \end{split}$$

In other words, it is not possible to deduce that:

$$\left\|S^{(1,\alpha)}\left[B_{1,2}^{+}\right]^{\omega}\mathcal{U}^{(2)}(t_{2}-t_{3})\left[B_{2,3}^{+}\right]^{\omega}\gamma^{(3)}\right\|_{L^{2}\left(\Omega\times\Lambda\times\Lambda\right)}\lesssim\left\|S^{(1,\alpha)}\gamma^{(3)}\right\|_{L^{2}(\Lambda^{3}\times\Lambda^{3})}$$

The construction is as follows:

Example 1. Let us consider the case $t_2 - t_3 = 0$. We show that the pairings due to the randomization do not allow us to prove:

$$\left\|S^{(1,\alpha)}\left[B^{+}_{1,2}\right]^{\omega}\left[B^{+}_{2,3}\right]^{\omega}\gamma^{(3)}\right\|_{L^{2}\left(\Omega\times\Lambda\times\Lambda\right)} \lesssim \left\|S^{(1,\alpha)}\gamma^{(3)}\right\|_{L^{2}(\Lambda^{3}\times\Lambda^{3})}.$$
 (63)

We compute:

$$([B_{1,2}^{+}]^{\omega} [B_{2,3}^{+}]^{\omega} \gamma^{(3)}])^{(\xi_{1};\xi_{1}')} = \sum_{\eta_{2},\eta_{2}'} ([B_{2,3}^{+}]^{\omega} \gamma^{(3)}])^{(\xi_{1}-\eta_{2}+\eta_{2}',\eta_{2};\xi_{1}',\eta_{2}')} \cdot h_{\xi_{1}}(\omega) \cdot h_{\xi_{1}-\eta_{2}+\eta_{2}'}(\omega) \cdot h_{\eta_{2}}(\omega) \cdot h_{\eta_{2}'}(\omega) = \sum_{\eta_{2},\eta_{2}',\eta_{3},\eta_{3}'} \widehat{\gamma}^{(3)} (\xi_{1}-\eta_{2}+\eta_{2}',\eta_{2}-\eta_{3}+\eta_{3}',\eta_{3};\xi_{1}',\eta_{2}',\eta_{3}') \cdot h_{\xi_{1}}(\omega) \cdot h_{\xi_{1}-\eta_{2}+\eta_{2}'}(\omega) \cdot h_{\eta_{2}'}(\omega) \cdot h_{\eta_{2}-\eta_{3}+\eta_{3}'}(\omega) \cdot h_{\eta_{3}}(\omega) \cdot h_{\eta_{3}'}(\omega).$$
(64)

In the above sum, we used the fact that $h_{\eta_2}^2(\omega) = 1$.

Hence, we need to consider:

$$\sum_{\substack{\xi_1,\xi_1'\\\tilde{\eta}_2,\eta_2',\eta_3,\eta_3'\\\tilde{\eta}_2,\tilde{\eta}_2',\tilde{\eta}_3,\tilde{\eta}_3'}} \sum_{\substack{\tilde{\eta}_2,\tilde{\eta}_2',\tilde{\eta}_3,\tilde{\eta}_3'\\\tilde{\eta}_2,\tilde{\eta}_2',\tilde{\eta}_3,\tilde{\eta}_3'}} \langle \xi_1 \rangle^{\alpha} \cdot \langle \xi_1' \rangle^{\alpha} \cdot \hat{\gamma}^{(3)} \left(\xi_1 - \tilde{\eta}_2 + \tilde{\eta}_2', \tilde{\eta}_2 - \tilde{\eta}_3 + \tilde{\eta}_3', \tilde{\eta}_3; \xi_1', \tilde{\eta}_2', \tilde{\eta}_3' \right) \cdot \\ h_{\xi_1 - \eta_2 + \eta_2'}(\omega) \cdot h_{\eta_2 - \eta_3 + \eta_3'}(\omega) \cdot h_{\eta_3}(\omega) \cdot h_{\eta_2'}(\omega) \cdot h_{\eta_3'}(\omega) \cdot \\ h_{\xi_1 - \tilde{\eta}_2 + \tilde{\eta}_2'}(\omega) \cdot h_{\tilde{\eta}_2 - \tilde{\eta}_3 + \tilde{\eta}_3'}(\omega) \cdot h_{\tilde{\eta}_3}(\omega) \cdot h_{\tilde{\eta}_3'}(\omega) \cdot h_{\tilde{\eta}_3'}(\omega).$$

In the above formula, we used the fact that $h_{\xi_1}^2(\omega) = 1$.

It is simple to apply the Cauchy–Schwarz inequality in ξ'_1 . We will henceforth consider the sum in the other variables. In particular, with the notation as in Definition 3.5, we need to consider:

$$\begin{split} &\xi_{1}, \eta_{2}, \eta_{2}', \eta_{3}, \eta_{3}', \tilde{\eta}_{2}, \tilde{\eta}_{2}', \tilde{\eta}_{3}, \tilde{\eta}_{3}'; \langle \xi_{1} \rangle^{2\alpha} \\ & \cdot \begin{bmatrix} \langle \xi_{1} - \eta_{2} + \eta_{2}' \rangle^{0} \langle \eta_{2} - \eta_{3} + \eta_{3}' \rangle^{0} \langle \eta_{3} \rangle^{0} \langle \eta_{2}' \rangle^{0} \langle \eta_{3}' \rangle^{0} \\ \langle \xi_{1} - \tilde{\eta}_{2} + \tilde{\eta}_{2}' \rangle^{0} \langle \eta_{2} - \tilde{\eta}_{3} + \tilde{\eta}_{3}' \rangle^{0} \langle \tilde{\eta}_{3} \rangle^{0} \langle \tilde{\eta}_{2}' \rangle^{0} \langle \tilde{\eta}_{3}' \rangle^{0} \end{bmatrix}$$

with the additional condition that we have a pairing of frequencies that comes from the randomization. By the randomization, we can obtain the analogue of property (*) from the proof of Theorem 3.1, that is we can deduce that each element in the set:

$$\left\{\xi_{1}-\eta_{2}+\eta_{2}',\eta_{2}-\eta_{3}+\eta_{3}',\eta_{3},\eta_{2}',\eta_{3}',\xi_{1}-\tilde{\eta}_{2}+\tilde{\eta}_{2}',\tilde{\eta}_{2}-\tilde{\eta}_{3}+\tilde{\eta}_{3}',\tilde{\eta}_{3},\tilde{\eta}_{2}',\tilde{\eta}_{3}'\right\}$$

occurs at least twice in the list:

$$\left(\xi_1 - \eta_2 + \eta'_2, \eta_2 - \eta_3 + \eta'_3, \eta_3, \eta'_2, \eta'_3, \xi_1 - \tilde{\eta}_2 + \tilde{\eta}'_2, \tilde{\eta}_2 - \tilde{\eta}_3 + \tilde{\eta}'_3, \tilde{\eta}_3, \tilde{\eta}'_2, \tilde{\eta}'_3\right).$$

The goal would be to distribute a factor of $\langle \xi_1 \rangle^{\alpha}$ over the first and second row of the matrix. Let us consider one pairing in which we see that this is not possible:

$$\begin{split} \xi_1 &- \eta_2 + \eta_2' = \eta_2 - \eta_3 + \eta_3' \\ \xi_1 &- \tilde{\eta}_2 + \tilde{\eta}_2' = \tilde{\eta}_2 - \tilde{\eta}_3 + \tilde{\eta}_3' \\ \eta_3 &= \tilde{\eta}_3 \\ \eta_2' &= \eta_3' \\ \tilde{\eta}_2' &= \tilde{\eta}_3'. \end{split}$$

In this case, we note that:

$$\xi_1 - \eta_2 + \eta'_2 = \eta_2 - \eta_3 + \eta'_3 = \eta_2 - \eta_3 + \eta'_2.$$

Hence, it follows that:

$$\xi_1 = 2\eta_2 - \eta_3.$$

We analogously obtain that:

$$\xi_1 = 2\tilde{\eta}_2 - \tilde{\eta}_3 = 2\tilde{\eta}_2 - \eta_3.$$

Consequently:

$$\eta_2 = \tilde{\eta}_2.$$

Moreover, we can eliminate the sum in ξ_1 since $\xi_1 = 2\eta_2 - \eta_3$. Hence, the expression we need to bound is:

$$\begin{split} &\eta_{2}, \eta_{2}', \tilde{\eta}_{2}', \eta_{3}; \left(2\eta_{2} - \eta_{3}\right)^{2\alpha} \\ &\cdot \begin{bmatrix} \langle \eta_{2} - \eta_{3} + \eta_{2}' \rangle^{0} & \langle \eta_{2} - \eta_{3} + \eta_{2}' \rangle^{0} & \langle \eta_{3} \rangle^{0} & \langle \eta_{2}' \rangle^{0} & \langle \eta_{2}' \rangle^{0} \\ & \langle \eta_{2} - \eta_{3} + \tilde{\eta}_{2}' \rangle^{0} & \langle \eta_{2} - \eta_{3} + \tilde{\eta}_{2}' \rangle^{0} & \langle \eta_{3} \rangle^{0} & \langle \tilde{\eta}_{2}' \rangle^{0} & \langle \tilde{\eta}_{2}' \rangle^{0} \end{bmatrix} . \end{split}$$

We change variables from $\eta_2 - \eta_3$ to η_2 and the above sum becomes:

$$\eta_{2}, \eta_{2}', \tilde{\eta}_{2}', \eta_{3}; \langle 2\eta_{2} + \eta_{3} \rangle^{2\alpha} \cdot \begin{bmatrix} \langle \eta_{2} + \eta_{2}' \rangle^{0} & \langle \eta_{2} + \eta_{2}' \rangle^{0} & \langle \eta_{3} \rangle^{0} & \langle \eta_{2}' \rangle^{0} \\ \langle \eta_{2} + \tilde{\eta}_{2}' \rangle^{0} & \langle \eta_{2} + \tilde{\eta}_{2}' \rangle^{0} & \langle \eta_{3} \rangle^{0} & \langle \tilde{\eta}_{2}' \rangle^{0} & \langle \tilde{\eta}_{2}' \rangle^{0} \end{bmatrix}.$$

At this point, we are supposed to distribute the two factors of $(2\eta_2 + \eta_3)^{\alpha}$. We can do this by using the fractional Leibniz rule in the first row as:

$$\langle 2\eta_2 + \eta_3 \rangle^{\alpha} \lesssim \langle \eta_2 + \eta_2' \rangle^{\alpha} \cdot \langle \eta_2 + \eta_2' \rangle^{\alpha} \cdot \langle \eta_3 \rangle^{\alpha} \cdot \langle \eta_2' \rangle^{\alpha} \cdot \langle \eta_2' \rangle^{\alpha},$$

which follows from the fact that:

$$2\eta_2 + \eta_3 = (\eta_2 + \eta'_2) + (\eta_2 + \eta'_2) + \eta_3 - \eta'_2 - \eta'_2.$$

Similarly, we use the fractional Leibniz rule in the second row as:

$$(2\eta_2+\eta_3)^{\alpha} \lesssim (\eta_2+\tilde{\eta}_2')^{\alpha} \cdot (\eta_2+\tilde{\eta}_2')^{\alpha} \cdot (\eta_3)^{\alpha} \cdot (\tilde{\eta}_2')^{\alpha} \cdot (\tilde{\eta}_2')^{\alpha}.$$

In particular, it suffices to estimate:

$$\eta_2, \eta'_2, \tilde{\eta}'_2, \eta_3; \begin{bmatrix} \langle \eta_2 + \eta'_2 \rangle^{\alpha} & \langle \eta_2 + \eta'_2 \rangle^{\alpha} & \langle \eta_3 \rangle^{\alpha} & \langle \eta'_2 \rangle^{\alpha} \\ \langle \eta_2 + \tilde{\eta}'_2 \rangle^{\alpha} & \langle \eta_2 + \tilde{\eta}'_2 \rangle^{\alpha} & \langle \eta_3 \rangle^{\alpha} & \langle \tilde{\eta}'_2 \rangle^{\alpha} & \langle \tilde{\eta}'_2 \rangle^{\alpha} \end{bmatrix}.$$

It is possible to estimate the sum in η_2 and in η_3 first by using the Cauchy–Schwarz inequality in these variables. If we want to estimate the sum in η'_2 or in $\tilde{\eta}'_2$, we need to lose derivatives in these variables and it is not possible to sum otherwise. This is due to the fact that the variables η'_2 and $\tilde{\eta}'_2$ occur in the same row of the matrix. In particular, we can use the previous method to estimate the sum from above by:

$$\lesssim \|S^{(3,\alpha+\frac{3}{4}+\epsilon)}\gamma^{(3)}\|_{L^2(\Lambda^3\times\Lambda^3)}^2.$$

Hence, in the case of the second Duhamel iteration, we cannot argue as in the Case IA3 in the proof of Theorem 3.1. The reason why this is the case is that the frequency η_2 does not appear in the sum (64). We are thus forced to use the fractional Leibniz rule in all of the variables.

Consequently, it is not possible to prove an estimate of the form (63) for the second Duhamel iteration in the class of general density matrices by using the pairings which come from the randomization. In what follows, we will explicitly write out the Duhamel expansion of arbitrary length. We will then use this formula to prove a randomized spacetime bound for a specific class of density matrices.

6.2. The Precise Form of the Duhamel Expansion Term

In this subsection, we explicitly write out the full form of the Duhamel expansion which occurs in the study of the randomized Gross–Pitaevskii hierarchy (39). In

this expansion, we will again use the fact that the randomization is obtained by using standard Bernoulli random variables. We will use this explicit formula in the proof of the randomized spacetime estimates in non-resonant classes given in Propositions 6.2 and 6.2 in the following two subsections.

The explicit expansion is obtained as follows. Let us fix $n, \ell \in \mathbb{N}$ and j_1, \ldots, j_ℓ , $k_1, \ldots, k_\ell \in \mathbb{N}$, with $j_1 < k_1 \le n + \ell, \ldots, j_\ell < k_\ell \le n + \ell$ and $t_1, t_2, \ldots, t_{\ell+1} \in \mathbb{R}$. We show that, for all sequences $(\gamma^{(m)})_m$ of density matrices, the following identity holds:

$$\left(\mathcal{U}^{(n)}(t_{1}-t_{2}) \left[B_{j_{1},k_{1}}^{\pm} \right]^{\omega} \mathcal{U}^{(n+1)}(t_{2}-t_{3}) \left[B_{j_{2},k_{2}}^{\pm} \right]^{\omega} \cdots \\ \cdots \mathcal{U}^{(n+\ell-1)}(t_{\ell}-t_{\ell+1}) \left[B_{j_{\ell},k_{\ell}}^{\pm} \right]^{\omega} \gamma^{(n+\ell)} \right)^{\widehat{}}(\xi_{1},\ldots,\xi_{n};\xi_{1}',\ldots,\xi_{n}') = \\ \sum_{\substack{\eta_{1},\ldots,\eta_{n+\ell}\\\eta_{1}',\ldots,\eta_{n+\ell}'}}^{*} e^{i(t_{1}-t_{2})(\cdots)} \cdot e^{i(t_{2}-t_{3})(\cdots)} \cdots e^{i(t_{\ell}-t_{\ell+1})(\cdots)} \\ \cdot \int_{\eta_{1}',\ldots,\eta_{n+\ell}'} \eta_{1}',\ldots,\eta_{n+\ell}';\eta_{1}',\ldots,\eta_{n+\ell}') \cdot \\ \cdot \prod_{\substack{1\leq j\leq n\\\xi_{j}\in\mathcal{A}}} \left\{ h_{\xi_{j}}(\omega) \cdot h_{\eta_{1}^{j,1}}(\omega) \cdot h_{\eta_{2}^{j,1}}(\omega) \cdots h_{\eta_{N_{j}}^{j,1}}(\omega) \right\} \\ \cdot \prod_{\substack{1\leq j\leq n\\\xi_{j}'\in\mathcal{B}}} \left\{ h_{\xi_{j}'}(\omega) \cdot h_{\eta_{1}^{j,2}}(\omega) \cdot h_{\eta_{2}^{j,2}}(\omega) \cdots h_{\eta_{M_{j}}^{j,2}}(\omega) \right\}.$$
(65)

Here, we need to explain the notation:

We say that $\xi_j \in \mathcal{A}$ for some $1 \leq j \leq n$ if a collision operator $[B_{j,k}^{\pm}]^{\omega}$ was applied to this frequency. In other words, $\xi_j \in \mathcal{A}$ if the frequency ξ_j does not appear in the list $(\eta_1, \ldots, \eta_{n+\ell}; \eta'_1, \ldots, \eta'_{n+\ell})$. Analogously, we say that $\xi'_j \in \mathcal{B}$ if a collision operator $[B_{j,k}^{\pm}]^{\omega}$ was applied to this frequency. The sets \mathcal{A} and \mathcal{B} are uniquely determined by the collision operators that we are applying.

Moreover,

$$\sum_{\substack{\eta_1,\ldots,\eta_{n+\ell}\\\eta_1',\ldots,\eta_{n+\ell}'}}^*(\cdots)$$

denotes the sum over $\eta_1, \ldots, \eta_{n+\ell}, \eta'_1, \ldots, \eta'_{n+\ell}$ which satisfy the following constraints:

(i) The set $\{\eta_1, \ldots, \eta_{n+\ell}, \eta'_1, \ldots, \eta'_{n+\ell}\}$ can be written as the union of:

$$\left(\bigcup_{\substack{1\leq j\leq n\\\xi_j\in\mathcal{A}}} \{\eta_1^{j,1},\ldots,\eta_{N_j}^{j,1}\}\right) \cup \left(\bigcup_{\substack{1\leq j\leq n\\\xi'_j\in\mathcal{B}}} \{\eta_1^{j,2},\ldots,\eta_{M_j}^{j,2}\}\right)$$

(these are the new frequencies which we obtain after applying the collision operators) and:

$$\{\xi_1,\ldots,\xi_n,\xi_1',\ldots,\xi_n'\}\setminus (\mathcal{A}\cup\mathcal{B}).$$

(These are the frequencies on which we do not apply the collision operators and which remain the same.)

(ii) Each $\xi_i \in \mathcal{A}$ can be written as:

$$\xi_j = \pm \eta_1^{j,1} \pm \eta_2^{j,1} \pm \dots \pm \eta_{N_j}^{j,1}$$

(iii) Each $\xi'_i \in \mathcal{B}$ can be written as:

$$\xi'_{j} = \pm \eta_{1}^{j,2} \pm \eta_{2}^{j,2} \pm \dots \pm \eta_{M_{j}}^{j,2}$$

The choice of \pm above is uniquely determined by the collision operators that we are applying. The choice of collision operators also uniquely determines the quantities N_j and M_j for j = 1, ..., n.

To summarize, in $\sum_{k=1}^{\infty}$, we are summing over all the frequency configurations which satisfy the above conditions. Let us note that the (\cdots) in $e^{i(t_j - t_{j+1})(\cdots)}$ denotes a real-valued function in the frequencies. Consequently, the exponential has modulus one and hence does not affect the subsequent estimates. Finally, in the above calculations and all of its subsequent applications, we will always fix a choice of signs for each $[B_{i,k}^{\pm}]^{\omega}$.

Before we proceed, let us give an explicit example of the expansion (65):

Example 2. We suppose that $n = 2, \ell = 3$. For simplicity of notation, let us analyze the case when $t_1 = t_2 = t_3 = t_4 = 0$. As was noted earlier, in the general case, we would multiply all the terms by factors of modulus 1. Let us consider the collision operators $[B_{1,2}^+]^{\omega}$, $[B_{2,3}^-]^{\omega}$, $[B_{4,5}^-]^{\omega}$.

In particular, we consider:

$$\begin{split} & \left([B_{1,2}^+]^{\omega} [B_{2,3}^-]^{\omega} [B_{4,5}^-]^{\omega} \gamma^{(5)} \right)^{\widehat{}} (\xi_1, \xi_2; \xi_1', \xi_2') \\ &= \sum_{\xi_3, \xi_3'} \left([B_{2,3}^-]^{\omega} [B_{4,5}^-]^{\omega} \gamma^{(5)} \right)^{\widehat{}} \left(\xi_1 - \xi_3 + \xi_3', \xi_3, \xi_2; \xi_1', \xi_3', \xi_2' \right) \cdot h_{\xi_1}(\omega) \\ & \cdot h_{\xi_1 - \xi_3 + \xi_3'}(\omega) \cdot h_{\xi_3}(\omega) \cdot h_{\xi_3'}(\omega) \\ &= \sum_{\xi_3, \xi_3', \xi_4, \xi_4'} \left([B_{4,5}^-]^{\omega} \gamma^{(5)} \right)^{\widehat{}} \left(\xi_1 - \xi_3 + \xi_3', \xi_3, \xi_4, \xi_2; \xi_1', \xi_3' - \xi_4' + \xi_4, \xi_4', \xi_2' \right) \cdot \\ & h_{\xi_1}(\omega) \cdot h_{\xi_1 - \xi_3 + \xi_3'}(\omega) \cdot h_{\xi_3}(\omega) \cdot h_{\xi_3'}(\omega) \cdot h_{\xi_3'}(\omega) \cdot h_{\xi_3'}(\omega) \cdot h_{\xi_4'}(\omega) \cdot h_{\xi_4}(\omega) \cdot h_{\xi_4'}(\omega). \end{split}$$

Since $h_{\xi'_3}^2(\omega) = 1$, this expression equals:

$$\begin{split} &\sum_{\xi_3,\xi_3',\xi_4,\xi_4'} \left([B_{4,5}^-]^\omega \gamma^{(5)} \right) \widehat{} \left(\xi_1 - \xi_3 + \xi_3', \xi_3, \xi_4, \xi_2; \xi_1', \xi_3' - \xi_4' + \xi_4, \xi_4', \xi_2' \right) \cdot \\ &h_{\xi_1}(\omega) \cdot h_{\xi_1 - \xi_3 + \xi_3'}(\omega) \cdot h_{\xi_3}(\omega) \cdot h_{\xi_3' - \xi_4' + \xi_4}(\omega) \cdot h_{\xi_4}(\omega) \cdot h_{\xi_4'}(\omega) \\ &= \sum_{\xi_3,\xi_3',\xi_4,\xi_4',\xi_5,\xi_5'} \widehat{\gamma}^{(5)} \left(\xi_1 - \xi_3 + \xi_3', \xi_3, \xi_4, \xi_2, \xi_5; \xi_1', \xi_3' \right) \\ &- \xi_4' + \xi_4, \xi_4', \xi_2' - \xi_5' + \xi_5, \xi_5' \right) \cdot \\ &h_{\xi_1}(\omega) \cdot h_{\xi_1 - \xi_3 + \xi_3'}(\omega) \cdot h_{\xi_3}(\omega) \cdot h_{\xi_3' - \xi_4' + \xi_4}(\omega) \cdot h_{\xi_4}(\omega) \cdot h_{\xi_4'}(\omega) \cdot \\ &h_{\xi_2'}(\omega) \cdot h_{\xi_2' - \xi_5' + \xi_5}(\omega) \cdot h_{\xi_5}(\omega) \cdot h_{\xi_5'}(\omega). \end{split}$$

Hence, we take $\mathcal{A} := \{\xi_1\}$ and $\mathcal{B} := \{\xi'_2\}$ and we note that:

$$\begin{aligned} \xi_1 &= \left(\xi_1 - \xi_3 + \xi_3'\right) + \xi_3 + \xi_4 - \left(\xi_3' - \xi_4' + \xi_4\right) - \xi_4' \\ \xi_2' &= -\xi_5 + \left(\xi_2' - \xi_5' + \xi_5\right) + \xi_5'. \end{aligned}$$

In the notation from (65), we hence take:

$$\begin{split} \eta_1^{1,1} &:= \xi_1 - \xi_3 + \xi_3', \eta_2^{1,1} := \xi_3, \eta_3^{1,1} := \xi_4, \eta_4^{1,1} := \xi_3' - \xi_4' + \xi_4, \eta_5^{1,1} := \xi_4' \\ \eta_1^{2,2} &:= \xi_5, \eta_2^{2,2} := \xi_2' - \xi_5' + \xi_5, \eta_3^{2,2} := \xi_5'. \end{split}$$

Consequently, we can write:

$$\begin{split} & \left([B_{1,2}^+]^{\omega} [B_{2,3}^-]^{\omega} [B_{4,5}^-]^{\omega} \gamma^{(5)} \right)^{\widehat{}}(\xi_1,\xi_2;\xi_1',\xi_2') \\ &= \sum_{\substack{\eta_1^{1,1} + \eta_2^{1,1} + \eta_3^{1,1} - \eta_4^{1,1} - \eta_5^{1,1} = \xi_1 \\ - \eta_1^{2,2} + \eta_2^{2,2} + \eta_3^{2,2} = \xi_2'} \widehat{\gamma}^{(5)} \left(\eta_2^{1,1}, \eta_2^{1,1}, \eta_3^{1,1}, \xi_2, \eta_1^{2,2}; \xi_1' \right) \\ & h_{\xi_1}(\omega) \cdot h_{\eta_1^{1,1}}(\eta_2^{1,2}, \eta_3^{2,2}) \cdot \\ h_{\xi_1}(\omega) \cdot h_{\eta_1^{1,1}}(\omega) \cdot h_{\eta_2^{1,1}}(\omega) \cdot h_{\eta_3^{1,1}}(\omega) \cdot h_{\eta_4^{1,1}}(\omega) \cdot h_{\eta_5^{1,1}}(\omega) \cdot \\ & h_{\xi_2'}(\omega) \cdot h_{\eta_1^{2,2}}(\omega) \cdot h_{\eta_2^{2,2}}(\omega) \cdot h_{\eta_3^{2,2}}(\omega). \end{split}$$

This is the expansion given in (65).

We now prove the identity (65) in the general case by induction on ℓ . Both the base case $\ell = 1$ and the inductive step follow from the form of the randomized collision operator. In particular, we note the following two identities, for $r \in \mathbb{N}$ and for $j < k \le r + 1$:

(1) Identity for $[B_{i,k}^+]^{\omega}$.

$$\begin{pmatrix} \mathcal{U}^{(r)}(t)[B_{j,k}^{+}]^{\omega}\gamma_{\omega}^{(r+1)} \end{pmatrix} \widehat{}(\vec{\mu}_{r};\vec{\mu}_{r}') = \sum_{\lambda,\lambda'} e^{it(-|\mu_{j}-\lambda+\lambda'|^{2}-|\lambda|^{2}-|\vec{\mu}_{r}|^{2}+|\mu_{j}|^{2}+|\vec{\mu}_{r}'|^{2}+|\lambda'|^{2})} \cdot \\ \widehat{\gamma}_{\omega}^{(r+1)}(\mu_{1},\ldots,\mu_{j-1},\mu_{j}-\lambda+\lambda',\mu_{j+1},\ldots,\mu_{k-1},\lambda,\mu_{k+1},\ldots,\mu_{r}; \\ \mu_{1}',\ldots,\mu_{k-1}',\lambda',\mu_{k+1}',\ldots,\mu_{r}') \cdot \\ h_{\mu_{j}}(\omega) \cdot h_{\mu_{j}-\lambda+\lambda'}(\omega) \cdot h_{\lambda}(\omega) \cdot h_{\lambda'}(\omega).$$
(66)

Here, the subscript ω in $\gamma_{\omega}^{(r+1)}$ denotes ω dependence in the density matrix (through factors of $h_{\xi}(\omega)$). We note in the above formula, that μ_j is no longer taken to be a frequency in $\widehat{\gamma_{\omega}}^{(r+1)}$, but the random factor $h_{\mu_j}(\omega)$ corresponding to μ_j still occurs. It is important to analyze this factor. We observe that the following two cases can occur:

- (a) $h_{\mu_j}(\omega)$ appears in $\widehat{\gamma_{\omega}}^{(r+1)}$; In this case we use the identity $h_{\mu_j}^2(\omega) = 1$ and we cancel the factor of $h_{\mu_j}(\omega)$ in $\widehat{\gamma_{\omega}}^{(r+1)}$ with the factor of $h_{\mu_j}(\omega)$ in (66). Consequently, the term $h_{\mu_j}(\omega)$ corresponding to μ_j does not appear in the random part at this stage.
- (b) $h_{\mu_j}(\omega)$ does not appear in $\widehat{\gamma_{\omega}}^{(r+1)}$; In this case $\mu_j = \xi_\ell$ for some $1 \le \ell \le r$ and the factor $h_{\xi_\ell}(\omega)$ corresponding to ξ_ℓ appears in the random part at this stage. We say that $\xi_\ell \in \mathcal{A}$.

In either case, we have the following relation between the frequencies of the left and right-hand side of (66)

$$\mu_{j} = (\mu_{j} - \lambda + \lambda') + \lambda - \lambda'.$$
(67)

(2) Identity for $[B_{j,k}^-]^{\omega}$.

For $\gamma_{\omega}^{(r+1)}$ as before, we note that:

$$\begin{pmatrix} \mathcal{U}^{(r)}(t)[B_{j,k}^{-}]^{\omega}\gamma_{\omega}^{(r+1)} \end{pmatrix} \widehat{} (\vec{\mu}_{r};\vec{\mu}_{r}') = \sum_{\lambda,\lambda'} e^{it(-|\vec{\mu}_{r}|^{2}-|\lambda|^{2}+|\mu_{j}'+\lambda-\lambda'|^{2}+|\lambda'|^{2}+|\vec{\mu}_{r}'|^{2}-|\mu_{j}|^{2})} \cdot \\ \widehat{\gamma}_{\omega}^{(r+1)}(\mu_{1},\ldots,\mu_{k-1},\lambda,\mu_{k+1},\ldots,\mu_{n};\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{j+1}',\ldots,\mu_{k-1}',\lambda,\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j-1}',\mu_{j}'+\lambda-\lambda',\mu_{k+1}',\ldots,\mu_{n}';\mu_{1}',\ldots,\mu_{j}',\mu_{j}$$

As before, two cases can occur:

(a) $h_{\mu'_j}(\omega)$ appears in $\widehat{\gamma}^{(r+1)}_{\omega}$; In this case, we use the fact that $h^2_{\mu'_j}(\omega) = 1$ and we deduce as before that the term $h_{\mu'_j}(\omega)$ corresponding to μ'_j does not appear in the random part at this stage.

469

(b) h_{μ'_j}(ω) does not appear ŷ_ω^(r+1); In this case, we note that μ'_j = ξ'_ℓ for some 1 ≤ ℓ ≤ n. Consequently, the factor h_{ξ'_ℓ}(ω) corresponding to ξ'_ℓ appears in the random part at this stage. We say that ξ'_ℓ ∈ B.

We furthermore note the following relation between the frequencies occuring on the left and right-hand side of (68):

$$\mu'_{j} = \left(\mu'_{j} - \lambda + \lambda'\right) - \lambda + \lambda'. \tag{69}$$

The identity (65) now follows by induction and by the use of (66) and (68). The construction of the $\eta_{\ell}^{j,1}$ and $\eta_{\ell}^{j,2}$ is based on the relations (67) and (69).

Similarly as before, we denote by:

$$\sum_{\substack{\tilde{\eta}_1,\ldots,\tilde{\eta}_{n+\ell}\\\tilde{\eta}_1',\ldots,\tilde{\eta}_{n+\ell}'}^*}$$

the sum over $\tilde{\eta}_1, \ldots, \tilde{\eta}_{n+\ell}, \tilde{\eta}'_1, \ldots, \tilde{\eta}'_{n+\ell}$ with the analogous constraints satisfied when the η and η' variables are replaced by the $\tilde{\eta}$ and $\tilde{\eta}'$ variables. In other words, for the same choice of collision operators and for the same definition of sets \mathcal{A} and \mathcal{B} and numbers N_j and M_j as before, we are summing over $\tilde{\eta}_1, \ldots, \tilde{\eta}_{n+\ell}, \tilde{\eta}'_1, \ldots, \tilde{\eta}'_{n+\ell}$ such that:

(i) The set $\{\tilde{\eta}_1, \ldots, \tilde{\eta}_{n+\ell}, \tilde{\eta}'_1, \ldots, \tilde{\eta}'_{n+\ell}\}$ can be written as the union of:

$$\left(\bigcup_{\substack{1\leq j\leq n\\\xi_j\in\mathcal{A}}} \{\tilde{\eta}_1^{j,1},\ldots,\tilde{\eta}_{N_j}^{j,1}\}\right) \cup \left(\bigcup_{\substack{1\leq j\leq n\\\xi_i'\in\mathcal{B}}} \{\tilde{\eta}_1^{j,2},\ldots,\tilde{\eta}_{M_j}^{j,2}\}\right)$$

(these are the new frequencies which we obtain after applying the collision operators) and:

 $\{\xi_1,\ldots,\xi_n,\xi_1',\ldots,\xi_n'\}\setminus (\mathcal{A}\cup\mathcal{B}).$

(These are the frequencies on which we do not apply the collision operators and which remain the same.)

(ii) Each $\xi_i \in \mathcal{A}$ can be written as:

$$\xi_j = \pm \tilde{\eta}_1^{j,1} \pm \tilde{\eta}_2^{j,1} \pm \cdots \pm \tilde{\eta}_{N_j}^{j,1}.$$

(iii) Each $\xi'_i \in \mathcal{B}$ can be written as:

$$\xi'_j = \pm \tilde{\eta}_1^{j,2} \pm \tilde{\eta}_2^{j,2} \pm \cdots \pm \tilde{\eta}_{M_j}^{j,2}.$$

From (65), we deduce that:

$$\left| \left(S^{(n,\alpha)} \mathcal{U}^{(n)}(t_1 - t_2) \left[B^{\pm}_{j_1,k_1} \right]^{\omega} \mathcal{U}^{(n+1)}(t_2 - t_3) \left[B^{\pm}_{j_2,k_2} \right]^{\omega} \cdots \right. \\ \cdots \mathcal{U}^{(n+\ell-1)}(t_{\ell} - t_{\ell+1}) \left[B^{\pm}_{j_{\ell},k_{\ell}} \right]^{\omega} \gamma^{(n+\ell)} \right) \widehat{} (\xi_1, \dots, \xi_n; \xi'_1, \dots, \xi'_n) \right|^2$$

$$= \langle \xi_{1} \rangle^{2\alpha} \cdots \langle \xi_{n} \rangle^{2\alpha} \cdot \langle \xi_{1}^{i} \rangle^{2\alpha} \cdots \langle \xi_{n}^{i} \rangle^{2\alpha} \cdot \sum_{\substack{\eta_{1}, \dots, \eta_{n+\ell} \\ \eta_{1}^{i}, \dots \\ \eta_{1}^{i}, \dots, \eta_{n+\ell} \\ \eta_{1}^{i}, \dots \\$$

Here, we used the fact that $h_{\xi_j}^2(\omega) = h_{\xi'_j}^2(\omega) = 1$. Let us note that the factors $e^{i(t_j - t_{j+1})(\cdots)}$ in (65) and in (70) are not necessarily the same. In what follows, we will only use the fact that these factors have modulus equal to 1.

From (70) and Plancherel's theorem, it follows that:

$$S^{(n,\alpha)}\mathcal{U}^{(n)}(t_{1}-t_{2})\left[B^{\pm}_{j_{1},k_{1}}\right]^{\omega}\mathcal{U}^{(n+1)}(t_{2}-t_{3})\left[B^{\pm}_{j_{2},k_{2}}\right]^{\omega}$$

$$\cdots\mathcal{U}^{(n+\ell-1)}(t_{\ell}-t_{\ell+1})\left[B^{\pm}_{j_{\ell},k_{\ell}}\right]^{\omega}\gamma^{(n+\ell)}\Big\|_{L^{2}(\Omega\times\Lambda^{n}\times\Lambda^{n})}^{2}$$

$$= \int_{\Omega}\left\{\sum_{\substack{\xi_{1},...,\xi_{n}\\\xi_{1}',...,\xi_{n}'}} \langle\xi_{1}\rangle^{2\alpha}\cdots\langle\xi_{n}\rangle^{2\alpha}\cdot\langle\xi_{1}'\rangle^{2\alpha}\cdots\langle\xi_{n}'\rangle^{2\alpha}$$

$$\cdot\sum_{\substack{\eta_{1},...,\eta_{n+\ell}\\\xi_{1}',...,\xi_{n}'}}^{*}\sum_{\substack{i=1\\ \gamma'_{1},...,\eta_{n+\ell}}}^{*}e^{i(t_{1}-t_{2})(\cdots)}\cdot e^{i(t_{2}-t_{3})(\cdots)}\cdots e^{i(t_{\ell}-t_{\ell+1})(\cdots)}$$

$$\tilde{\gamma}^{(n+\ell)}(\eta_{1},\ldots,\eta_{n+\ell};\eta_{1}',\ldots,\eta_{n+\ell}')\cdot\overline{\gamma^{(n+\ell)}}(\tilde{\eta}_{1},\ldots,\tilde{\eta}_{n+\ell};\tilde{\eta}_{1}',\ldots,\tilde{\eta}_{n+\ell}')$$

$$\cdot\prod_{\substack{1\leq j\leq n\\\xi_{j}\in\mathcal{A}}}\left\{h_{\eta_{1}^{j,1}}(\omega)\cdot h_{\eta_{2}^{j,1}}(\omega)\cdots h_{\eta_{N_{j}}^{j,1}}(\omega)\cdot h_{\tilde{\eta}_{1}^{j,1}}(\omega)\cdot h_{\tilde{\eta}_{2}^{j,1}}(\omega)\cdots h_{\tilde{\eta}_{N_{j}}^{j,2}}(\omega)\right\}dp(\omega).$$

$$\prod_{\substack{1\leq j\leq n\\\xi_{j}'\in\mathcal{B}}}\left\{h_{\eta_{1}^{j,2}}(\omega)\cdot h_{\eta_{2}^{j,2}}(\omega)\cdots h_{\eta_{M_{j}}^{j,2}}(\omega)\cdot h_{\tilde{\eta}_{1}^{j,2}}(\omega)\cdots h_{\tilde{\eta}_{M_{j}}^{j,2}}(\omega)$$

$$(71)$$

This is the quantity that we would like to estimate.

6.3. A Special Class of Density Matrices

In this subsection, we study a special class of density matrices $(\gamma^{(m)})_m$ in which we can prove an upper bound for the expression in (71). We note that by the results in Section 6.1, we cannot estimate this quantity using pairing of frequencies due to randomization in the general class of density matrices, even in the case when $\ell = 2$. As we will see below, we will be able to prove an upper bound on (71) if we impose additional *non-resonance* conditions. This is reminiscent of the ideas in [23,61,128]. The precise bound is given in Proposition 6.2 below.

More precisely, we consider:

Definition 6.1. Let \mathcal{N} denote the class of all density matrices $(\gamma^{(m)})_m$ such that:

(i) For all $m \in \mathbb{N}$ and for all $(\xi_1, \ldots, \xi_m, \xi'_1, \ldots, \xi'_m) \in (\mathbb{Z}^3)^m \times (\mathbb{Z}^3)^m$,

$$\widehat{\gamma}^{(m)}(\xi_1,\ldots,\xi_m;\xi_1',\ldots,\xi_m')=0$$

unless

$$|\xi_1| > |\xi_2| > \cdots > |\xi_m| > |\xi'_1| > |\xi'_2| > \cdots |\xi'_m|.$$

(ii) There exists $C_1 > 0$ independent of *m* such that for all $m \in \mathbb{N}$:

$$\left\|S^{(m,\alpha)}\gamma^{(m)}\right\|_{L^2(\Lambda^m\times\Lambda^m)}\leq C_1^m.$$

We call the class \mathcal{N} the class of *non-resonant density matrices*. Let us note that the elements in \mathcal{N} are *not symmetric* in the sense that it is not in general true that for all $m \in \mathbb{N}$, $(x_1, \ldots, x_m, x'_1, \ldots, x'_m) \in \Lambda^m \times \Lambda^m$, $\sigma \in S_m$:

$$\gamma^{(m)}\left(x_{\sigma(1)},\ldots,x_{\sigma(m)};x'_{\sigma(1)},\ldots,x'_{\sigma(m)}\right)=\gamma^{(m)}\left(x_1,\ldots,x_m;x'_1,\ldots,x_m\right).$$

Furthermore, we observe that the class \mathcal{N} is non-empty. We can construct an element of \mathcal{N} by taking a density matrix satisfying the condition (ii) and by then setting all of the Fourier coefficients not satisfying the constraint from (i) to equal zero. Finally, let us note that the class \mathcal{N} does not contain the factorized objects studied earlier. In the analysis that follows, we see that the ordering in condition (i) on the sizes of the frequencies $\xi_1, \ldots, \xi_m, \xi'_1, \ldots, \xi'_m$ could be replaced by any other (fixed) ordering of the elements of $\{1, 2, \ldots, 2m\}$.

The rest of this section is devoted to the study of the randomized Gross–Pitaevskii hierarchy in the non-resonant class \mathcal{N} .

6.4. The Randomized Estimate in the Class \mathcal{N}

We use the randomization in (71) in order to deduce that each element of:

$$\mathcal{S} := \left(\bigcup_{\substack{1 \le j \le n \\ \xi_j \in \mathcal{A}}} \{\eta_1^{j,1}, \dots, \eta_{N_j}^{j,1}\} \right) \cup \left(\bigcup_{\substack{1 \le j \le n \\ \xi'_j \in \mathcal{B}}} \{\eta_1^{j,2}, \dots, \eta_{M_j}^{j,2}\} \right)$$

gets paired with at least one element of:

$$\tilde{\mathcal{S}} := \left(\bigcup_{\substack{1 \le j \le n \\ \xi_j \in \mathcal{A}}} \{ \tilde{\eta}_1^{j,1}, \dots, \tilde{\eta}_{N_j}^{j,1} \} \right) \cup \left(\bigcup_{\substack{1 \le j \le n \\ \xi'_j \in \mathcal{B}}} \{ \tilde{\eta}_1^{j,2}, \dots, \tilde{\eta}_{M_j}^{j,2} \} \right).$$

We recall that, by definition of the class \mathcal{N} , the elements of each of the sets \mathcal{S} and $\tilde{\mathcal{S}}$ are mutually distinct. Since $\#\mathcal{S} = \#\tilde{\mathcal{S}}$, it follows that each element of \mathcal{S} gets paired with exactly one element of $\tilde{\mathcal{S}}$ and vice versa. Let us recall the *fractional Leibniz rule*, which states that for all $\alpha \ge 0$, there exists $C_3 > 0$ such that, for all $x, y \in \mathbb{R}^3$:

$$\langle x + y \rangle^{\alpha} \le C_3 \langle x \rangle^{\alpha} \cdot \langle y \rangle^{\alpha}.$$
 (72)

The constant C_3 depends on α . We iteratively use the fractional Leibniz rule and the fact that there is a bijection between S and \tilde{S} to deduce that, for $\alpha \ge 0$:

$$\begin{split} \left\| S^{(n,\alpha)} \mathcal{U}^{(n)}(t_{1}-t_{2}) \left[B_{j_{1},k_{1}}^{\pm} \right]^{\omega} \mathcal{U}^{(n+1)}(t_{2}-t_{3}) \left[B_{j_{2},k_{2}}^{\pm} \right]^{\omega} \\ \cdots \mathcal{U}^{(n+\ell-1)}(t_{\ell}-t_{\ell+1}) \left[B_{j_{\ell},k_{\ell}}^{\pm} \right]^{\omega} \gamma^{(n+\ell)} \right\|_{L^{2}(\Omega \times \Lambda^{n} \times \Lambda^{n})}^{2} \\ &\leq C_{3}^{2(n+\ell)} \sum_{\substack{\eta_{1},\ldots,\eta_{n+\ell},\eta_{1}',\ldots,\eta_{n+\ell}'\\\tilde{\eta}_{1},\ldots,\tilde{\eta}_{n+\ell},\tilde{\eta}_{1}',\ldots,\tilde{\eta}_{n+\ell}'}^{**}} \langle \eta_{1} \rangle^{\alpha} \cdots \langle \eta_{n+\ell} \rangle^{\alpha} \cdot \langle \eta_{1}' \rangle^{\alpha} \cdots \langle \eta_{n+\ell}' \rangle^{\alpha} \\ \cdot \langle \tilde{\eta}_{1} \rangle^{\alpha} \cdots \langle \tilde{\eta}_{n+\ell} \rangle^{\alpha} \cdot \langle \tilde{\eta}_{1}' \rangle^{\alpha} \cdots \langle \tilde{\eta}_{n+\ell}' \rangle^{\alpha} \\ \cdot \left| \hat{\gamma}^{(n+\ell)}(\eta_{1},\ldots,\eta_{n+\ell};\eta_{1}',\ldots,\eta_{n+\ell}') \right| \cdot \left| \hat{\gamma}^{(n+\ell)}(\tilde{\eta}_{1},\ldots,\tilde{\eta}_{n+\ell};\tilde{\eta}_{1}',\ldots,\tilde{\eta}_{n+\ell}') \right|. \end{split}$$
(73)

More precisely, we use conditions (ii) and (iii) from the definitions of the sets $\{\eta_1, \ldots, \eta_{n+\ell}, \eta'_1, \ldots, \eta'_{n+\ell}\}$ and $\{\tilde{\eta}_1, \ldots, \tilde{\eta}_{n+\ell}, \tilde{\eta}'_1, \ldots, \tilde{\eta}'_{n+\ell}\}$ in the sums:

$$\sum_{\substack{\eta_1,\ldots,\eta_{n+\ell}\\\eta'_1,\ldots,\eta'_{n+\ell}}}^* (\cdots)$$

and

$$\sum_{\substack{\tilde{\eta}_1,\ldots,\tilde{\eta}_{n+\ell}\\\tilde{\eta}_1',\ldots,\tilde{\eta}_{n+\ell}'}^*}^*(\cdots)$$

and we apply the estimate (72) at most $2(n+\ell)$ times to obtain the factor of $C_3^{2(n+\ell)}$ in the bound. Here,

$$\sum_{\substack{\eta_1,\ldots,\eta_{n+\ell},\eta_1',\ldots,\eta_{n+\ell}'\\\tilde{\eta}_1,\ldots,\tilde{\eta}_{n+\ell},\tilde{\eta}_1',\ldots,\tilde{\eta}_{n+\ell}'}^{**}}$$

denotes the sum



with the additional constraint that there exists a bijective pairing between the sets S and \tilde{S} , which was noted earlier. We also use the fact that $|e^{i(t_j - t_{j+1})(\cdots)}| = 1$ for all $j = 1, \ldots, \ell$. Finally, for frequencies $\xi_j \notin A$ and $\xi'_k \notin B$, we use the fact that $\langle \xi_j \rangle^{\alpha} \ge 1$ and $\langle \xi'_k \rangle^{\alpha} \ge 1$.

If $(\gamma^{(m)})_m \in \mathcal{N}$, we can use condition (i) in the definition \mathcal{N} in order to deduce that the only possible pairing is given by:

$$\eta_1 = \tilde{\eta}_1, \eta_2 = \tilde{\eta}_2, \dots, \eta_{n+\ell} = \tilde{\eta}_{n+\ell} \eta'_1 = \tilde{\eta}'_1, \eta'_2 = \tilde{\eta}'_2, \dots, \eta'_{n+\ell} = \tilde{\eta}'_{n+\ell}.$$

Consequently, the right-hand side of the expression in (73) equals:

$$C_{3}^{2(n+\ell)} \sum_{\substack{\eta_{1},\dots,\eta_{n+\ell}\\\eta_{1}',\dots,\eta_{n+\ell}'}} \langle \eta_{1} \rangle^{2\alpha} \cdots \langle \eta_{n+\ell} \rangle^{2\alpha} \cdot \langle \eta_{1}' \rangle^{2\alpha} \cdots \langle \eta_{n+\ell}' \rangle^{2\alpha} \\ \left| \widehat{\gamma}^{(n+\ell)}(\eta_{1},\dots,\eta_{n+\ell};\eta_{1}',\dots,\eta_{n+\ell}') \right|^{2} = C_{3}^{2(n+\ell)} \| S^{(n+\ell,\alpha)} \gamma^{(n+\ell)} \|_{L^{2}(\Lambda^{n+\ell} \times \Lambda^{n+\ell})}^{2}.$$

We hence deduce the following result:

Proposition 6.2. Suppose that $\alpha \geq 0$. Then, there exists $C_2 > 0$, depending only on α such that for all $(\gamma^{(m)})_m \in \mathcal{N}$, for all $n, \ell \in \mathbb{N}$; $j_1, \ldots, j_\ell, k_1, \ldots, k_\ell \in \mathbb{N}$, with $j_1 < k_1 \leq n + \ell, \ldots, j_\ell < k_\ell \leq n + \ell$ and $t_1, t_2, \ldots, t_{\ell+1} \in \mathbb{R}$, the following bound holds:

$$\begin{split} & \| S^{(n,\alpha)} \mathcal{U}^{(n)}(t_1 - t_2) [B^{\pm}_{j_1,k_1}]^{\omega} \mathcal{U}^{(n+1)}(t_2 - t_3) [B^{\pm}_{j_2,k_2}]^{\omega} \\ & \cdots \mathcal{U}^{(n+\ell-1)}(t_\ell - t_{\ell+1}) [B^{\pm}_{j_\ell,k_\ell}]^{\omega} \gamma^{(n+\ell)} \|_{L^2(\Omega \times \Lambda^n \times \Lambda^n)} \\ & \leq C_2^{n+\ell} \| S^{(n+\ell,\alpha)} \gamma^{(n+\ell)} \|_{L^2(\Lambda^{n+\ell} \times \Lambda^{n+\ell})}. \end{split}$$

Proof. The result immediately follows from the above discussion if we take C_2 to equal the constant C_3 from the estimate (72).

Remark 6.3. We observe that we are only using condition (i) in the definition of the class \mathcal{N} .

6.5. An Application of the Randomized Estimate to the Study of the Dependently Randomized GP Hierarchy

We now apply the randomized estimate from Proposition 6.2 to the study of the randomized Gross–Pitaevskii hierarchy (39). We argue similarly as before and we fix $(\gamma^{(m)}(t))_m$ to be a time-dependent sequence in \mathcal{N} . Here, we assume that the constant C_1 from part (ii) of the definition of \mathcal{N} is uniform in *t*. In this case, we say that $(\gamma^{(m)}(t))_m$ belongs to \mathcal{N} uniformly in time.

Given $k, n \in \mathbb{N}$, $t_k > 0$ and $\omega \in \Omega$, we define $\sigma_{n;\omega}^{(k)}$ by:

$$\sigma_{n;\,\omega}^{(k)}(t_{k})$$

$$:= (-i)^{n} \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} \mathcal{U}^{(k)}(t_{k} - t_{k+1}) [B^{(k+1)}]^{\omega} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2})$$

$$\begin{bmatrix} B^{(k+2)} \end{bmatrix}^{\omega} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k})$$

$$[B^{(n+k)}]^{\omega} \gamma^{(n+k)}(t_{n+k}) dt_{n+k} \cdots dt_{k+2} dt_{k+1}.$$
(74)

This is the *k*-density matrix which we obtain after *n* Duhamel iterations in the hierarchy (39). In this notation, the superscript *k* denotes the number of particles and the subscript *n* denotes the length of the Duhamel expansion. Furthermore, $\omega \in \Omega$ is a fixed randomization parameter.

Given $(\sigma_{n;\omega}^{(k)})_{n,k}$, we define:

$$\begin{split} \tilde{\gamma}^{(1)} &:= \sigma_{n;\,\omega}^{(1)} \\ \tilde{\gamma}^{(2)} &:= \sigma_{n-1;\,\omega}^{(2)} \\ \tilde{\gamma}^{(3)} &:= \sigma_{n-2;\,\omega}^{(3)} \\ &\vdots \\ \tilde{\gamma}^{(n)} &:= \sigma_{1:\,\omega}^{(n)}. \end{split}$$

By construction, it follows that:

$$\begin{cases} i \partial_t \tilde{\gamma}^{(k)} + (\Delta_{\vec{x}_k} - \Delta_{\vec{x}'_k}) \tilde{\gamma}^{(k)} = \sum_{j=1}^k [B_{j,k+1}]^{\omega} (\tilde{\gamma}^{(k+1)}) \\ \tilde{\gamma}^{(k)} \Big|_{t=0} = 0 \end{cases}$$

for all $k \in \{1, 2, ..., n-1\}$. In other words, we obtain an *arbitrarily long subset of solutions* to the full randomized Gross–Pitaevskii hierarchy (39) with homogeneous initial data. Let us note that the property of belonging to the class \mathcal{N} is not necessarily preserved under the evolution of (39). However, we do would not need to use this fact since we are studying the behavior of the fixed Duhamel expansions defined in (74).

We will now state the main result of this section:

Theorem 6.4. Suppose that $\alpha \ge 0$ and $k \in \mathbb{N}$. Consider $\sigma_{n;\omega}^{(k)}$, defined as in (74) for $(\gamma^{(m)}(t))_m \in \mathcal{N}$, uniformly in time. There exists T > 0, depending only on C_1 and α such that:

$$\sup_{t \in [0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega}^{(k)}(t) \right\|_{L^2\left(\Omega \times \Lambda^k \times \Lambda^k\right)} \to 0 \text{ as } n \to \infty$$

Proof. Let us take $t = t_k \in [0, T]$ and we compute:

$$S^{(k,\alpha)}\sigma_{n;\omega}^{(k)}(t_{k}) = (-i)^{n} \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} S^{(k,\alpha)} \mathcal{U}^{(k)}(t_{k} - t_{k+1}) [B^{(k+1)}]^{\omega} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) [B^{(k+2)}]^{\omega} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) [B^{(n+k)}]^{\omega} \gamma^{(n+k)}(t_{n+k}) dt_{n+k} \cdots dt_{k+2} dt_{k+1}$$

Hence, by Minkowski's inequality:

$$\begin{split} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega}^{(k)}(t_{k}) \right\|_{L^{2}\left(\Omega \times \Lambda^{k} \times \Lambda^{k}\right)} \\ &\leq \int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} \left\| S^{(k,\alpha)} \mathcal{U}^{(k)}(t_{k} - t_{k+1}) \left[B^{(k+1)} \right]^{\omega} \mathcal{U}^{(k+1)}(t_{k+1} - t_{k+2}) \\ & \left[B^{(k+2)} \right]^{\omega} \cdots \mathcal{U}^{(n+k-1)}(t_{n+k-1} - t_{n+k}) \left[B^{(n+k)} \right]^{\omega} \gamma^{(n+k)}(t_{n+k}) \right\|_{L^{2}\left(\Omega \times \Lambda^{k} \times \Lambda^{k}\right)} \\ & dt_{n+k} \cdots dt_{k+2} d_{k+1}. \end{split}$$

Since we are working in the class \mathcal{N} of asymmetric non-resonant density matrices, we cannot apply the boardgame argument. Hence, we write out the full collision operators in terms of the individual collision operators. More precisely, we note that the integrand can be bounded from above by a sum of $\frac{(n+k)!}{(k-1)!}$ terms of the type estimated by Proposition 6.2. We then use Proposition 6.2 to estimate the whole expression from above by:

$$\int_{0}^{t_{k}} \int_{0}^{t_{k+1}} \cdots \int_{0}^{t_{n+k-1}} C_{2}^{n+k} \cdot \frac{(n+k)!}{(k-1)!} \\ \cdot \|S^{(n+k,\alpha)}\gamma^{(n+k)}(t_{n+k})\|_{L^{2}(\Lambda^{n+k} \times \Lambda^{n+k})} \, \mathrm{d}t_{n+k} \cdots \, \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1}.$$

By using the a priori assumption (ii) on $\gamma^{(n+k)}(t_{n+k})$, that is the fact that $(\gamma^{(m)}(t))_m \in \mathcal{N}$, uniformly in time, it follows that this quantity is:

$$\leq \int_0^{t_k} \int_0^{t_{k+1}} \cdots \int_0^{t_{n+k-1}} C_2^{n+k} \cdot \frac{(n+k)!}{(k-1)!} \cdot C_1^{n+k} \, \mathrm{d}t_{n+k} \cdots \, \mathrm{d}t_{k+2} \, \mathrm{d}t_{k+1},$$

which by (61) is:

$$\leq \frac{T^n}{n!} \cdot C_2^{n+k} \cdot \frac{(n+k)!}{(k-1)!} \cdot C_1^{n+k}.$$

From the previous analysis, it follows that there exists $M \sim C_1 \cdot C_2$, independent of *n* and *k*, such that:

$$\sup_{t\in[0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega}^{(k)}(t) \right\|_{L^2\left(\Omega \times \Lambda^k \times \Lambda^k\right)} \leq \frac{T^n}{n!} \cdot M^{n+k} \cdot \frac{(n+k)!}{(k-1)!}$$

Arguing analogously as in the proof of Theorem 5.2, this quantity is:

$$\lesssim_k (2MT)^n \cdot M^k.$$

We now choose T sufficiently small such that 2MT < 1 and we deduce that:

$$\sup_{t \in [0,T]} \left\| S^{(k,\alpha)} \sigma_{n;\,\omega}^{(k)}(t) \right\|_{L^2\left(\Omega \times \Lambda^k \times \Lambda^k\right)} \to 0 \text{ as } n \to \infty$$

By construction T depends only on C_1 and on C_2 . Since C_2 depends only on α , it follows that T depends only on C_1 and α . The claim now follows.

Remark 6.5. As in the proof of Theorem 5.2, we can deduce that $T \sim_{\alpha} \frac{1}{C_1}$, for the constant C_1 in the definition of the class \mathcal{N} .

Acknowledgments The authors would like to thank PHILIP GRESSMAN and ANTTI KNOWLES for valuable discussions. They would also like to thank HRVOJE ŠIKIĆ for several useful comments. They are grateful to MARTIN ZWIERLEIN for directing their attention to the results [112,160] in the experimental physics literature. The authors would like to thank the referees for their constructive suggestions. VEDRAN SOHINGER was partially supported by a Simons Postdoctoral Fellowship. GIGLIOLA STAFFILANI was partially supported by NSF Grant DMS-1068815.

References

- ADAMI, R., BARDOS, C., GOLSE, F., TETA, A.: Towards a rigorous derivation of the cubic nonlinear Schrödinger equation in dimension one. *Asymptot. Anal.* 40(2), 93– 108 (2004)
- ADAMI, R., GOLSE, F., TETA, A.: Rigorous derivation of the cubic NLS in dimension one. J. Stat. Phys. 127(6), 1193–1220 (2007)
- ADAMS, S., COLLEVECCHIO, A., KÖNIG, W.: A variational formula for the free energy of an interacting many-particle system. *Ann. Prob.* 39(2), 683–728 (2011)
- AIZENMAN, M., LIEB, E.H., SEIRINGER, R., SOLOVEJ, J.P., YNGVASON, J.: Bose– Einstein quantum phase transition in an optical lattice model. *Phys. Rev. A* 70, 023612 (2004)
- AIZENMAN, M., LIEB, E.H., SEIRINGER, R., SOLOVEJ, J.P., YNGVASON, J.: Bose– Einstein condensation as a quantum phase transition in an optical lattice. *Math. Phys. Q. Mech. Lect. Notes Phys.*, 690, 199–215 (2006)
- 6. ANAPOLITANOS, I.: Rate of convergence towards the Hartree-von Neumann limit in the mean-field regime. *Lett. Math. Phys.* **98**(1), 1–31 (2011)
- 7. ANDERSON, M.H., ENSHER, J.R., MATTHEWS, M.R., WIEMAN, C.E., CORNELL, E.A.: Observations of Bose–Einstein condensation in a dilute atomic vapor. *Science* **269**, 198–201 (1995)
- AYACHE, A., TZVETKOV, N.: L^p Properties for Gaussian random series. Trans. Am. Math. Soc. 360(8), 4425–4439 (2008)
- BARDEEN, J., COOPER, L.N., SCHRIEFFER, J.R.: Microscopic theory of superconductivity. *Phys. Rev.* 102, 162–164 (1957)
- BARDEEN, J., COOPER, L.N., SCHRIEFFER, J.R.: Theory of superconductivity. *Phys. Rev.* 108(5), 1175–1204 (1957)
- 11. BARDOS, C., GOLSE, F., MAUSER, N.: Weak coupling limit of the *N*-particle Schrödinger equation. *Methods Appl. Anal.* **7**, 275–293 (2000)
- 12. BECKNER, W.: Multilinear embedding estimates for the fractional Laplacian. *Math. Res. Lett.* **19**, 175–189 (2012)
- 13. BECKNER, W.: Convolution estimates and the Gross–Pitaevskii hierarchy (preprint) (2011). arXiv:1111.3857
- BEN AROUS, G., KIRKPATRICK, K., SCHLEIN, B.: A central limit theorem in many-body quantum dynamics. *Commun. Math. Phys.* 321(2), 371–417 (2013)
- BENEDIKTER, N., PORTA, M., SCHLEIN, B.: Mean-field evolution of fermionic systems. Commun. Math. Phys. 331(3), 1087–1131 (2014)
- 16. BENEDIKTER, N., PORTA, M., SCHLEIN, B.: Mean-field dynamics of fermions with relativistic dispersion. J. Math. Phys. 55(2), Art. ID 021901, 10 pp (2014)
- BENEDIKTER, N., DE OLIVEIRA, G., SCHLEIN, B.: Quantitative derivation of the Gross– Pitaevskii equation (preprint) (2012). arXiv:1208.0373
- BENYI, A., OH, T., POCOVNICU, O.: Wiener randomization on unbounded domains and an application to almost sure well-posedness of NLS (preprint) (2014). arXiv:1405.7326 (to appear in excursions in harmonic analysis)
- 19. BENYI, A., OH, T., POCOVNICU, O.: On the probabilistic Cauchy theory of the cubic nonlinear Schrödinger equation on \mathbb{R}^d , $d \ge 3$ (preprint) (2014). arXiv:1405.7327
- BOSE, S.N.: Plancks Gesetz und Lichtquantenhypothese. Zeitschrift f
 ür Physik 26, 178 (1924)
- BOURGAIN, J.: Fourier transform restriction phenomena for certain lattice subsets and applications to nonlinear evolution equations, I: Schrödinger equations. *Geom. Funct. Anal.* 3, 107–156 (1993)
- BOURGAIN, J.: Periodic nonlinear Schrödinger equation and invariant measures. Commun. Math. Phys. 166(1), 1–26 (1994)
- BOURGAIN, J.: On the Cauchy problem and invariant measure problem for the periodic Zakharov system. *Duke Math. J.* 76, 175–202 (1994)

- BOURGAIN, J.: Invariant measures for the 2D-defocusing nonlinear Schrödinger equation. *Commun. Math. Phys.* 176(2), 421–445 (1996)
- 25. BOURGAIN, J.: Invariant measures for the Gross–Pitaevskii equation. J. Math. Pures et Appl. **T76**(F. 8), 649–702 (1997)
- 26. BOURGAIN, J.: Refinements of Strichartz's Inequality and applications to 2D-NLS with critical nonlinearity. *Int. Math. Res. Not.* **5**, 253–283 (1998)
- 27. BOURGAIN, J.: Global well-posedness of the defocusing critical nonlinear Schrödinger equation in the radial case. J. Am. Math. Soc. **12**(1), 145–171 (1999)
- BOURGAIN, J., BULUT, A.: Gibbs measure evolution in radial nonlinear wave and Schrödinger equations on the ball. C. R. Math. Acad. Sci. Paris 350(11–12), 571–575 (2012)
- BOURGAIN, J., BULUT, A.: Almost sure global well posedness for the radial nonlinear Schrödinger equation on the unit ball II: the 3D case. J. Eur. Math. Soc. (JEMS) 16(6), 1289–1325 (2014)
- 30. BOURGAIN, J., BULUT, A.: Invariant Gibbs measure evolution for the radial nonlinear wave equation on the 3D ball. *J. Funct. Anal.* **266**(4), 2319–2340 (2014)
- BURQ, N., THOMANN, L., TZVETKOV, N.: Global infinite energy solutions for the cubic wave equation (preprint) (2012). arXiv:1210.2086
- 32. BURQ, N., TZVETKOV, N.: Invariant measure for a three dimensional nonlinear wave equation. *Int. Math. Res. Not. IMRN* **2007**(22), Art. ID rnm108, 26 pp (2007)
- BURQ, N., TZVETKOV, N.: Random data Cauchy theory for supercritical wave equations. I. Local theory. *Invent. Math.* 173(3), 449–475 (2008)
- BURQ, N., TZVETKOV, N.: Random data Cauchy theory for supercritical wave equations. II. Local theory. *Invent. Math.* 173(3), 477–496 (2008)
- BURQ, N., TZVETKOV, N.: Probabilistic well-posedness for the cubic wave equation. J. Eur. Math. Soc. 16(1), 1–30 (2014)
- 36. CACCIAFESTA, F., DE SUZZONI, A.-S.: Invariant measure for the Schrödinger equation on the real line (preprint) (2014). arXiv:1405.5107
- 37. CHATTERJEE, S., DIACONIS, P.: Fluctuations of the Bose–Einstein condensate. J. Phys. A 47(8), 085201 (2014) (23 pp)
- CHEN, L., LEE, J.O., SCHLEIN, B.: Rate of convergence towards Hartree dynamics. J. Stat. Phys. 144(4), 872–903 (2011)
- CHEN, T., HAINZL, C., PAVLOVIĆ, N., SEIRINGER, R.: Unconditional uniqueness for the cubic Gross–Pitaevskii hierarchy via quantum de Finetti (preprint) (2013). arXiv:1307.3168
- CHEN, T., HAINZL, C., PAVLOVIĆ, N., SEIRINGER, R.: On the well-posedness and scattering for the Gross–Pitaevskii hierarchy via quantum de Finetti. *Lett. Math. Phys.* 104(7), 871–891 (2014)
- CHEN, T., PAVLOVIĆ, N.: On the Cauchy problem for focusing and defocusing Gross– Pitaevskii hiearchies. *Discr. Contin. Dyn. Syst.* 27(2), 715–739 (2010)
- 42. CHEN, T., PAVLOVIĆ, N.: The quintic NLS as the mean field limit of a Boson gas with three-body interactions. *J. Funct. Anal.* **260**(4), 959–997 (2011)
- CHEN, T., PAVLOVIĆ, N.: Higher order energy conservation and global well-posedness for Gross–Pitaevskii hierarchies. *Commun. Partial Differ. Equ.* 39(9), 1597–1634 (2014)
- CHEN, T., PAVLOVIĆ, N.: A new proof of existence of solutions for focusing and defocusing Gross–Pitaevskii hierarchies. Proc. Am. Math. Soc. 141(1), 279–293 (2013)
- CHEN, T., PAVLOVIĆ, N.: Derivation of the cubic NLS and Gross–Pitaevskii hierarchy from many-body dynamics in *d* = 2, 3 based on spacetime norms. *Ann. Henri Poincaré* 15(3), 543–588 (2014)
- CHEN, T., PAVLOVIĆ, N., TZIRAKIS, N.: Energy conservation and blowup of solutions for focusing and defocusing Gross–Pitaevskii hierarchies. *Ann. H. Poincaré (C), Anal. Non-Lin.* 27(5), 1271-1290 (2010)
- CHEN, T., PAVLOVIĆ, N., TZIRAKIS, N.: Multilinear Morawetz Identities for the Gross– Pitaevskii Hierarchy. Recent Advances in Harmonic Analysis and Partial Differential

Equations, 39–62, Contemporary Mathematics, vol. 581. American Mathematical Society, Providence (2012)

- CHEN, T., TALIAFERRO, K.: Derivation in strong topology and global well-posedness of solutions to the Gross-Pitaevskii hierarchy. *Commun. Partial Differ. Equ.* 39(9), 1658–1693 (2014)
- CHEN, X.: The Grillakis–Machedon–Margetis second order corrections to mean field evolution for weakly interacting Bosons in the case of 3-body interactions (preprint) (2009). arXiv:0911.4153
- CHEN, X.: Second order corrections to mean field evolution for weakly interacting bosons in the case of three-body interactions. *Arch. Ration. Mech. Anal.* 203, 455–497 (2012)
- CHEN, X.: Collapsing estimates and the rigorous derivation of the 2d cubic nonlinear Schrödinger equation with anisotropic switchable quadratic traps. J. Math. Pures Appl. (9) 98(4), 450–478 (2012)
- 52. CHEN, X.: On the rigorous derivation of the 3D cubic nonlinear Schrödinger Equation with a quadratic trap. *Arch. Ration. Mech. Anal.* **210**(2), 365–408 (2013)
- CHEN, X., HOLMER, J.: On the rigorous derivation of the 2D cubic nonlinear Schrödinger equation from 3D quantum many-body dynamics. *Arch. Ration. Mech. Anal.* 210(3), 909–954 (2013)
- 54. CHEN, X., HOLMER, J.: On the Klainerman–Machedon conjecture of the quantum BBGKY hierarchy with self-interaction (preprint) (2013). arXiv:1303.5385
- CHEN, X., HOLMER, J.: Focusing quantum many-body dynamics: the rigorous derivation of the 1D focusing cubic nonlinear Schrödinger equation (preprint) (2013). arXiv:1308.3895
- CHEN, X., HOLMER, J.: Focusing quantum many-body dynamics II: the rigorous derivation of the 1D focusing cubic nonlinear Schrödinger equation from 3D (preprint) (2014). arXiv:1407.8457
- 57. CHEN, X., HOLMER, J.: Correlation structures, many-body scattering processes and the derivation of the Gross–Pitaevskii hierarchy (preprint) (2014). arXiv:1409.1425
- CHEN, X., SMITH, P.: On the unconditional uniqueness of solutions to the infinite radial Chern–Simons–Schrödinger hierarchy. *Anal. PDE* 7(7), 1683–1712 (2014)
- 59. CHEN, Z., LIU, C.: On the Cauchy problem for Gross–Pitaevskii hierarchies. J. Math. Phys. **52**(3), 032103 (2011)
- COLLIANDER, J., KEEL, M., STAFFILANI, G., TAKAOKA, H., TAO, T.: Global wellposedness for Schrödinger equations with derivative. *SIAM J. Math. Anal.* 33(3), 649– 669 (2001)
- 61. COLLIANDER, J., OH, T.: Almost sure well-posedness of the cubic nonlinear Schrödinger equation below $L^2(\mathbb{T})$. Duke Math. J. 161(3), 367–414 (2012)
- 62. COOPER, L.N.: Bound electron pairs in a degenerate fermi gas. *Phys. Rev.* **104**, 1189–1190 (1956)
- 63. CRAMER, M., EISERT, J.: A quantum central limit theorem for non-equilibrium systems: exact relaxation of correlated states. *New. J. Phys.* **12**, 055020 (2009)
- 64. CUSHEN, C.D., HUDSON, R.L.: A quantum-mechanical central limit theorem. J. Appl. Prob. 8, 454 (1971)
- DAVIS, K.B., MEWES, M.O., ANDREWS, M.R., VAN DRUTEN, N.J., DURFEE, D.S., KURN, D.M., KETTERLE, W.: Bose–Einstein condensation in a gas of sodium atoms. *Phys. Rev. Let.* **75**(22), 3969–3973 (1995)
- 66. DENG, C., CUI, S.: Random-data Cauchy problem for the Navier–Stokes equations on \mathbb{T}^3 . J. Differ. Equ. **251**(4–5), 902–917 (2011)
- 67. DENG, C., CUI, S.: Random-data Cauchy problem for the periodic Navier–Stokes equations with initial data in negative-order Sobolev spaces. J. Differ. Equ. 251(4–5), 902–917 (2011)
- DENG, Y.: Two dimensional NLS equation with random initial data. Anal. PDE 5(5), 913–960 (2012)

- 69. DENG, Y.: Invariance of the Gibbs measure for the Benjamin–Ono equation (preprint) (2012). arXiv:1210.1542
- 70. DENG, Y., TZVETKOV, N., VISCIGLIA, N.: Invariant measures and long time behaviour for the Benjamin–Ono equation III (preprint) (2014). arXiv:1405.4954
- DOOB, J.L.: Stochastic processes with an integral-valued parameter. *Trans. Am. Math. Soc.* 44(1), 87–150 (1938)
- 72. EINSTEIN, A.: Quantentheorie des einatomigen idealen Gases. Sitzungsberichte der Preussischen Akademie der Wissenschaften 1, 3 (1925)
- ELGART, A., ERDOS, L., SCHLEIN, B., YAU, H.-T.: Gross–Pitaevskii equation as the mean field limit of weakly coupled bosons. *Arch. Ration. Mech. Anal.* 179, 265–283 (2006)
- ELGART, A., SCHLEIN, B.: Mean field dynamics of boson stars. Commun. Pure Appl. Math. 60(4), 500–545 (2007)
- ERDOS, L., SCHLEIN, B.: Quantum dynamics with mean field interactions: a new approach. J. Stat. Phys. 134(5), 859–870 (2009)
- ERDOS, L., SCHLEIN, B., YAU, H.-T.: Derivation of the Gross–Pitaevskii hierarchy for the dynamics of Bose–Einstein condensate. *Commun. Pure Appl. Math.* 59(12), 1659–1741 (2006)
- ERDOS, L., SCHLEIN, B., YAU, H.-T.: Derivation of the cubic non-linear Schrödinger equation from quantum dynamics of many-body systems. *Invent. Math.* 167(3), 515– 614 (2007)
- ERDOS, L., SCHLEIN, B., YAU, H.-T.: Rigorous derivation of the Gross–Pitaevskii equation. *Phys. Rev. Lett.* 98(4), 040404 (2007)
- ERDOS, L., SCHLEIN, B., YAU, H.-T.: Rigorous derivation of the Gross-Pitaevskii equation with a large interaction potential. J. Am. Math. Soc. 22(4), 1099–1156 (2009)
- ERDOS, L., SCHLEIN, B., YAU, H.-T.: Derivation of the Gross–Pitaevskii equation for the dynamics of Bose–Einstein condensate. *Ann. Math.* (2) 172(1), 291–370 (2010)
- ERDOS, L., YAU, H.-T.: Derivation of the nonlinear Schrödinger equation from a many body Coulomb system. *Adv. Theor. Math. Phys.* 5(6), 1169–1205 (2001)
- FEDERBUSH, P.: A partially alternate derivation of a result of Nelson. J. Math. Phys. 10, 50–52 (1969)
- FICHTNER, K.-H.: On the position distribution of the ideal Bose gas. *Math. Nachr.* 151, 59–67
- FRÖHLICH, J., GRAFFI, S., SCHWARZ, S.: Mean-field and classical limit of many body Schrödinger dynamics for bosons. *Commun. Math. Phys.* 271(3), 681–697 (2007)
- FRÖHLICH, J., KNOWLES, A., PIZZO, A.: Atomism and quantization. J. Phys. A 40(12), 3033–3045 (2007)
- FRÖHLICH, J., KNOWLES, A., SCHWARZ, S.: On the mean-field limit of bosons with Coulomb two-body interaction. *Commun. Math. Phys.* 288(3), 1023–1059 (2009)
- FRÖHLICH, J., LENZMANN, E.: Mean-Field Limit of Quantum Bose Gases and Nonlinear Hartree Equation. Sem. É.D.P. 2003–2004, Exp. No. XIX, p. 26. Sémin. Équ. Dériv. Partielles, École Polytech., Palaiseau (2004)
- GINIBRE, J., VELO, G.: The classical field limit of scattering theory for nonrelativistic many-boson systems I. *Commun. Math. Phys.* 66(1), 37–76 (1979)
- GINIBRE, J., VELO, G.: The classical field limit of scattering theory for nonrelativistic many-boson systems II. *Commun. Math. Phys.* 68(1), 45–68 (1979)
- GLIMM, J.: Boson fields with non-linear self-interaction in two dimensions. *Commun. Math. Phys.* 8, 12–25 (1968)
- GODERIS, D., VERBEURE, A., VETS, P.: About the mathematical theory of quantum fluctuations. In: *Mathematical Methods in Statistical Mechanics, Leuven Notes in Mathematical and Theoretical Physics*, vol. 1. Leuven University Press, Leuven (1989)
- 92. GRESSMAN, P., SOHINGER, V., STAFFILANI, G.: On the uniqueness of solutions to the 3D periodic Gross–Pitaevskii hierarchy. J. Funct. Anal. **266**(7), 4705–4764 (2014)
- GRILLAKIS, M., MACHEDON, M.: Pair excitations and the mean field approximation of interacting bosons, I. *Comunn. Math. Phys.* 324(2), 601–636 (2013)

- GRILLAKIS, M., MACHEDON, M., MARGETIS, D.: Second order corrections to mean field evolution of weakly interacting bosons, I. *Commun. Math. Phys.* 294, 273–301 (2010)
- GRILLAKIS, M., MACHEDON, M., MARGETIS, D.: Second order corrections to mean field evolution of weakly interacting bosons, II. Adv. Math. 228(3), 1788–1815 (2011)
- 96. GROSS, E.: Structure of a quantized vortex in boson systems. *Nuovo Cimento* **20** (1961), 454–466.
- 97. GROSS, L.: Logarithmic Sobolev inequalities. Am. J. Math. 97(4), 1061–1083 (1975)
- GROSS, L.: Hypercontractivity and logarithmic Sobolev inequalities for the Clifford Dirichlet form. Duke Math. J. 42(3), 383–396 (1975)
- HAYASHI, M.: Quantum estimation and the quantum central limit theorem. *Sci. Technol.* 227, 95 (2006)
- HEPP, K.: The classical limit for quantum mechanical correlation functions. *Commun. Math. Phys.* 35, 265–277 (1974)
- HEPP, K., LIEB, E.H.: Phase transitions in reservoir-driven open systems with applications to lasers and superconductors. *Helv. Phys. Acta* 46, 573–603 (1973)
- 102. HERR, S., SOHINGER, V.: The Gross–Pitaevskii hierarchy on general rectangular tori (preprint) (2014). arXiv:1410.5338
- 103. HONG, Y., TALIAFERRO, K., XIE, Z.: Unconditional uniqueness of the cubic Gross-Pitaevskii hierarchy with low regularity (preprint) (2014). arXiv:1402.5347
- HONG, Y., TALIAFERRO, K., XIE, Z.: Uniqueness of solutions to the 3D quintic Gross– Pitaevskii hierarchy (preprint) (2014). arXiv:1410.6961
- 105. JAKSIĆ, V., PAUTRAT, Y., PILLET, C.-A.: A quantum central limit theorem for sums of iid random variables. J. Math. Phys. **51**, 015208 (2010)
- 106. KAKUTANI, S.: Notes on infinite product measure spaces. *Proc. Imp. Acad.* **19**(3), 148–151 (1943)
- 107. KIRKPATRICK, K., SCHLEIN, B., STAFFILANI, G.: Derivation of the cubic nonlinear Schrödinger equation from quantum dynamics of many-body systems: the periodic case. *Am. J. Math.* **133**(1), 91–130 (2011)
- 108. KLAINERMAN, S., MACHEDON, M.: On the uniqueness of solutions to the Gross-Pitaevskii hierarchy. *Commun. Math. Phys.* **279**(1), 169–185 (2008)
- 109. KNOWLES, A.: Limiting dynamics in large quantum systems. Ph.D. thesis, ETH (2009)
- KNOWLES, A., PICKL, P.: Mean-field dynamics: singular potentials and rate of convergence. *Commun. Math. Phys.* 298(1), 101–139 (2010)
- 111. KOLMOGOROV, A.N.: Grundbegriffe der Wahrscheinlichkeitsrechnung. Springer, Berlin, 1933
- KU, M.J.H., SOMMER, A.T., CHEUK, L.W., ZWIERLEIN, M.W.: Revealing the superfluid lambda transition in the universal thermodynamics of a unitary fermi gas. *Science* 335, 563–567 (2012)
- 113. KUPERBERG, G.: A tracial quantum central limit theorem. *Trans. Am. Math. Soc.* **357**(2), 459–471 (2005)
- 114. LEBOWITZ, J., ROSE, H., SPEER, E.: Statistical mechanics of the nonlinear Schrödinger equation. *J. Stat. Phys.* **50**(3–4), 657–687 (1988)
- 115. LEE, J.O.: Rate of convergence towards semi-relativistic Hartree dynamics. Ann. H. Poincaré (C) 14(2), 313–346 (2013)
- 116. LEWIN, M., NAM, P.T., ROUGERIE, N.: Derivation of nonlinear Gibbs measures from many-body quantum dynamics (preprint) (2014). arXiv:1410.0335
- 117. LIEB, E.H., SEIRINGER, R.: Proof of Bose–Einstein condensation for dilute trapped gases. *Phys. Rev. Lett.* 88, 170409-1-4 (2002)
- 118. LIEB, E.H., SEIRINGER, R., SOLOVEJ, J.P., YNGVASON, J.: *The mathematics of the Bose gas and its condensation. Oberwolfach Seminars*, vol. 34. Birkhäuser Verlag, Basel, 2005
- 119. LIEB, E.H., SEIRINGER, R., YNGVASON, J.P.: Bosons in a trap: a rigorous derivation of the Gross–Pitaevskii energy functional. *Phys. Rev. A* **61**, 043602 (2000)

- 120. LIEB, E., SEIRINGER, R., YNGVASON, J.: A rigorous derivation of the Gross-Pitaevskii energy functional for a two-dimensional Bose gas. Dedicated to Joel L. Lebowitz. *Comm. Math. Phys.* **224**(1), 17–31 (2001)
- LÜHRMANN, J.: Mean-field quantum dynamics with magnetic fields. J. Math. Phys. 53(2), 022105 (2012) (19 pp)
- 122. LÜHRMANN, J., MENDELSON, D.: Random data Cauchy theory for nonlinear wave equations of power-type on \mathbb{R}^3 . *Commun. Partial Differ. Equ.* **39**(12), 2262–2283 (2014)
- 123. MARCINKIEWICZ, J., ZYGMUND, A.: Sur les foncions indépendantes. *Fund. Math.* 28, 60–90 (1937); Reprinted in Józef Marcinkiewicz, Collected papers, edited by Antoni Zygmund, Panstwowe Wydawnictwo Naukowe, Warsaw, pp. 233–259 (1964)
- 124. MICHELANGELI, A., SCHLEIN, B.: Dynamical collapse of boson stars. *Commun. Math. Phys.* **311**(3), 645–687 (2012)
- NAHMOD, A., OH, T., REY-BELLET, L., STAFFILANI, G.: Invariant Wiener measures and almost sure global well-posedness for the periodic derivative NLS. J. Eur. Math. Soc. (JEMS) 14(4), 1275–1330 (2012)
- NAHMOD, A., REY-BELLET, L., SHEFFIELD, S., STAFFILANI, G.: Absolute continuity of Gaussian measures under certain gauge transformations. *Math. Res. Lett.* 18(5), 875–887 (2011)
- NAHMOD, A., PAVLOVIĆ, N., STAFFILANI, G.: Almost sure existence of global weak solutions for super-critical Navier–Stokes equations. *SIAM J. Math. Anal.* 45(6), 3431– 3452 (2013)
- 128. NAHMOD, A., STAFFILANI, G.: Randomization in nonlinear pde and the supercritical periodic quintic NLS in 3*D* (preprint) (2013). arXiv:1308.1169
- 129. NELSON, E.: The free Markoff field. J. Funct. Anal. 12, 211–227 (1973)
- OH, T.: Gibbs measures and almost sure global well-posedness for coupled KdV systems. *Differ. Integr. Equ.* 22(7-8), 637–668 (2009)
- 131. OH, T.: Invariance of the white noise for KdV. *Commun. Math. Phys.* **292**(1), 217–236 (2009); Erratum: "Invariance of the white noise for KdV" (in preparation)
- 132. OH, T.: Invariance of the Gibbs measure for the Schrödinger–Benjamin–Ono System. *SIAM J. Math. Anal.* **41**(1), 2207–2225 (2009)
- 133. OH, T., SULEM, C.: On the one-dimensional cubic nonlinear Schrodinger equation below L^2 . *Kyoto J. Math.* **52**(1), 99–115 (2012)
- 134. PALEY, R.E.A.C., ZYGMUND, A.: On some series of functions 1. *Proc. Camb. Philos. Soc.* **26**, 337–357 (1930)
- 135. PALEY, R.E.A.C., ZYGMUND, A.: On some series of functions 2. *Proc. Camb. Philos. Soc.* **26**, 458–474 (1930)
- 136. PALEY, R.E.A.C., ZYGMUND, A.: On some series of functions 3. Proc. Camb. Philos. Soc. 28, 190–205 (1932)
- 137. PICKL, P.: Derivation of the time dependent Gross–Pitaevskii equation with external fields. *J. Stat. Phys.* **140**(1), 76–89 (2010)
- 138. PICKL, P.: A simple derivation of mean field limits for quantum systems. *Lett. Math. Phys.* **97**(2), 151–164 (2011)
- 139. PITAEVSKII, L.: Vortex lines in an imperfect Bose gas. Sov. Phys. JETP 13, 451–454 (1961)
- 140. PROKOF'EV, N., SVISTUNOV, B.: Bold diagrammatic Monte Carlo technique: When the sign problem is welcome. *Phys. Rev. Lett.* **99**, 250201 (2007)
- 141. PROKOF'EV, N., SVISTUNOV, B.: Bold diagrammatic Monte Carlo: A generic signproblem tolerant technique for polaron models and possibly interacting many-body problems. *Phys. Rev. B* **77**, 125101 (2008)
- RADEMACHER, H.: Einige Sätze über Reihen von allgemeinen Orthogonalfunktionen. Math Ann. 87, 112–138 (1922)
- 143. RAFLER, M.: Gaussian loop- and Pólya processes: a point process approach. Ph.D. thesis, University of Potsdam (2009)

- 144. RICHARDS, G.: Invariance of the Gibbs measure for the periodic quartic gKdV (preprint) (2012). arXiv:1209.4337
- 145. RODNIANSKI, I., SCHLEIN, B.: Quantum fluctuations and rate of convergence towards mean field dynamics. *Commun. Math. Phys.* **291**(1), 31–61 (2009)
- 146. SCHLEIN, B.: Derivation of effective evolution equations from microscopic quantum dynamics, chapter. In: D. ELLWOOD, I. RODNIANSKI, G. STAFFILANI, J. WUNSCH (eds.) *Clay Mathematics Proceedings of Evolution Equations*, vol. 17, p. 572, 2013
- 147. SOHINGER, V.: Local existence of solutions to Randomized Gross-Pitaevskii hierarchies. *Trans. Am. Math. Soc* (preprint) (2014). arXiv:1401.0326 (to appear)
- 148. SOHINGER, V.: A rigorous derivation of the defocusing cubic nonlinear Schrödinger equation from the dynamics of many-body quantum systems. *Annales Institut Henri Poincaré C, Analyse Non-Linéaire* (preprint) (2014). arXiv:1405.3003 (to appear)
- 149. SPOHN, H.: Kinetic equations from Hamiltonian Dynamics. *Rev. Mod. Phys.* **52**(3), 569–615 (1980)
- 150. DE SUZZONI, A.-S.: Invariant measure for the cubic wave equation on the unit ball of \mathbb{R}^3 . Dyn. Partial Differ. Equ. 8(2), 127–147 (2011)
- 151. DE SUZZONI, A.-S.: On the use of normal forms in the propagation of random waves (preprint) (2013). arXiv:1307.0619
- 152. DE SUZZONI, A.-S.: Invariant measure for the Klein–Gordon equation in a non periodic setting (preprint) (2014). arXiv:1403.2274
- 153. DE SUZZONI, A.-S., TZVETKOV, N.: On the propagation of weakly nonlinear random dispersive waves. *Arch. Ration. Mech. Anal.* **212**(3), 849–874 (2014)
- 154. THOMANN, L.: Random data Cauchy problem for supercritical Schrödinger equations. Ann. Inst. H. Poincaré **26**(6), 2385–2402 (2009)
- 155. THOMANN, L., TZVETKOV, N.: Gibbs measure for the periodic derivative nonlinear Schrödinger equation. *Nonlinearity* **23**(11), 2771–2791 (2010)
- 156. TZVETKOV, N.: Invariant measures for the nonlinear Schrödinger equation on the disc. *Dyn. Partial Differ. Equ.* **3**(2), 111–160 (2006)
- 157. TZVETKOV, N.: Invariant measures for the defocusing nonlinear Schrödinger equation. *Ann. Inst. Fourier (Grenoble)* **58**(7), 2543–2604 (2008)
- TZVETKOV, N.: Construction of a Gibbs measure associated to the periodic Benjamin– Ono equation. *Probab. Theory Rel. Fields* 146, 481–514 (2010)
- 159. VAN HOUCKE, K., KOZIK, E., PROKOF'EV, N., SVISTUNOV, B.: In: LANDAU, D.P., LEWIS, S.P., SCHUTTLER H.B. (eds.) Computer Simulation Studies in Condensed Matter Physics XXI. Springer, 2008
- 160. VAN HOUCKE, K., WERNER, F., KOZIK, E., PROKOF'EV, N., SVISTUNOV, B., KU, M.J.H., SUMMER, A.T., CHEUK, L.W., SCHIROTZEK, A., ZWIERLEIN, M.W.: Feynman diagrams versus Fermi-gas Feynman emulator. *Nat. Phys.*, 8, May (2012)
- 161. WOLFF, T.H.: Lectures on harmonic analysis. University Lecture Series, vol. 29. American Mathematical Society, Providence, RI (2003)
- XIE, Z.: Derivation of a nonlinear Schrödinger equation with a general power-type nonlinearity (preprint) (2013). arXiv:1305.7240
- 163. XU, S.: Invariant Gibbs measure for 3D NLW in infinite volume (preprint) (2014). arXiv:1405.3856
- ZHANG, T., FANG, D.: Random data Cauchy theory for the incompressible three dimensional Navier–Stokes equations. *Proc. AMS* 139(8), 2827–2837 (2011)
- ZHIDKOV, P.E.: An invariant measure for the nonlinear Schrödinger equation. (Russian) Dokl. Akad. Nauk SSSR 317(3), 543–546 (1991); translation in Soviet Math. Dokl. 43(2), 431–434
- 166. ZHIDKOV, P.E.: Korteweg-de Vries and Nonlinear Schrödinger Equations: Qualitative Theory. Lecture Notes in Mathematics, vol. 1756. Springer, Berlin, 2001

Departement Mathematik, Eidgenössische Technische Hochschule Zürich, Office G49.1, Rämistrasse 101, 8092 Zürich, Switzerland. e-mail: vedran.sohinger@math.ethz.ch URL: http://www.math.ethz.ch/~vedrans

and

Department of Mathematics, Massachusetts Institute of Technology, Building E17, Office 330, 77 Massachusetts Avenue, Cambridge, MA 01239-4301, USA. e-mail: gigliola@math.mit.edu URL: http://math.mit.edu/~gigliola/

(Received October 1, 2013 / Accepted March 17, 2015) Published online April 4, 2015 – © Springer-Verlag Berlin Heidelberg (2015)